PSYCHOLOGICAL DETERMINANTS OF FINANCIAL PLANNING FOR RETIREMENT AMONG UNIVERSITY EMPLOYEES

CHOO SIEW MING

INSTITUTE FOR ADVANCED STUDIES UNIVERSITY OF MALAYA KUALA LUMPUR

2020

PSYCHOLOGICAL DETERMINANTS OF FINANCIAL PLANNING FOR RETIREMENT AMONG UNIVERSITY EMPLOYEES

CHOO SIEW MING

THESIS SUBMITTED IN FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

INSTITUTE FOR ADVANCED STUDIES UNIVERSITY OF MALAYA KUALA LUMPUR

2020

UNIVERSITY OF MALAYA ORIGINAL LITERARY WORK DECLARATION

Name of Candidate: CHOO SIEW MING

Matric No: HHB110016

Name of Degree: **DOCTOR OF PHILOSOPHY**

Title of Project Paper/Research Report/Dissertation/Thesis ("this Work"):

PSYCHOLOGICAL DETERMINANTS OF FINANCIAL PLANNING FOR RETIREMENT AMONG UNIVERSITY EMPLOYEES

Field of Study: ECONOMICS AND PSYCHOLOGY

I do solemnly and sincerely declare that:

- (1) I am the sole author/writer of this Work;
- (2) This Work is original;
- (3) Any use of any work in which copyright exists was done by way of fair dealing and for permitted purposes and any excerpt or extract from, or reference to or reproduction of any copyright work has been disclosed expressly and sufficiently and the title of the Work and its authorship have been acknowledged in this Work;
- (4) I do not have any actual knowledge nor do I ought reasonably to know that the making of this work constitutes an infringement of any copyright work;
- (5) I hereby assign all and every rights in the copyright to this Work to the University of Malaya ("UM"), who henceforth shall be owner of the copyright in this Work and that any reproduction or use in any form or by any means whatsoever is prohibited without the written consent of UM having been first had and obtained;
- (6) I am fully aware that if in the course of making this Work I have infringed any copyright whether intentionally or otherwise, I may be subject to legal action or any other action as may be determined by UM.

Candidate's Signature

Date:

Subscribed and solemnly declared before,

Witness's Signature

Date:

Name:

Designation:

PSYCHOLOGICAL DETERMINANTS OF FINANCIAL PLANNING FOR RETIREMENT AMONG UNIVERSITY EMPLOYEES

ABSTRACT

Retirement refers to the post-work period of a person's life. During this period, individuals cease working, thus affecting their regular source of income. Hence, it is important that individuals plan to cater for the financial requirements during retirement. However, planning is not an easy task as individuals are besieged by a variety of psychological, environmental, financial and life challenges. Current research in this field had been inconclusive and fragmented on the determinants influencing planning behaviours. This research aims to draw on the constructs of the Theory of Planned Behaviour (TPB) to investigate this behaviour, in particular planning financially for retirement. TPB assumes that behaviour can be predicted and explained by using a set of determinants; that is attitudes, subjective norms and perceived behavioural control. Research using TPB's integrative framework has been positive in many studies but its use in financial and retirement behaviour studies is limited. To investigate the suitability of TPB, a survey is conducted among a sample of university employees in Malaysia. The findings reveal that subjective norms and attitudes influence intentions to plan. Conversely, perceived behavioural control has no influence on intentions. These findings provide empirical support for the use of TPB as a general model of planning for financial retirement in Malaysia. It contributes to the body of knowledge on financial behaviours, by stressing the importance of understanding the motivations of individuals in their choices on financial matters, planning and retirement issues.

Keywords: attitudes, subjective norms, perceived behavioural control, beliefs, Theory of Planned Behaviour

PENENTU PSIKOLOGI PERANCANGAN KEWANGAN UNTUK PERSARAAN DIKALANGAN PEKERJA UNIVERSITY

ABSTRAK

Persaraan merujuk kepada fasa kehidupan dimana seseorang tidak lagi bekerja. Tanpa pekerjaan, sumber pendapatan tetap terjejas. Oleh itu, adalah penting individu membuat persediaan dengan merancang keperluan kewangan mereka semasa persaraan. Walau bagaimanapun, merancang bukanlah tugas yang mudah kerana individu menghadapi pelbagai cabaran psikologi, persekitaran, kewangan dan kehidupan. Kajian semasa dalam bidang ini juga tidak teratur dan menyakinkan. Olehitu, tujuan kajian ini adalah untuk menggunakan konstruk Teori Perilaku Terancang (TPB) untuk mendalami perlakuan merancang, khususnya dalam perancangan kewangan untuk persaraan. TPB mengandai yang tingkah laku dapat diramalkan dan dijelaskan dengan menggunakan satu set penentu; iaitu sikap, norma subjektif dan persepsi kawalan perlakuan. Dari tinjauan, kajian yang menggunakan struktur integrative TPB membuahkan hasil yang positif. Namun demikian, penggunaannya dalam tingkahlaku kewangan dan persaraan agak terhad. Untuk mengkaji kesesuaian TPB, satu tinjauan dijalankan dikalangan pekerja universiti di Malaysia. Hasil kajian menunjukkan bahawa norma subjektif dan sikap mempengaruhi niat untuk merancang. Sebaliknya, persepsi kawalan perlakuan tidak mempengaruhi niat. Penemuan ini secara empirikalnya menyokong penggunaan TPB sebagai model umum dalam perancangan kewangan untuk persaraan di Malaysia. Ini mempertingkatkan pemahaman dalam tingkah laku kewangan, dan menekankan kepentingan motivasi seseorang dalam isu-isu kewangan, perancangan dan persaraan.

Kata kunci: sikap, norma subjektif, persepsi kawalan perlakuan, kepercayaan, Teori Perilaku Terancang

ACKNOWLEDGEMENTS

I would like to take this opportunity to document my heartfelt gratitude to all the kind and helpful people who have lend me a hand in my PhD. journey. It is almost impossible to cover everyone in just a one-page acknowledgement but I would attempt to encompass everyone.

Firstly, I would like to thank my supervisors, Emeritus Professor Datuk Norma Mansor and Associate Professor Dr. Diana-Lea Baranovich, who was always there for me in my time of need and had given me valuable advice and guidance in understanding the various nuances of research. Next, I would like to document my appreciation to Employees Provident Fund (EPF) for providing me with the funding required for the research. I would also like to thank the staff, friends and employees at the Social Wellbeing Research Centre (SWRC), University of Malaya for their advice, help and support.

In every step of my PhD journey, I had the privilege to meet many esteem individuals, professors, academics, friends and colleagues which have given me new insights and revelations. They are people who shared ideas and extended help generously, both in University of Malaya and in the many other universities in Malaysia that I have visited. To these people, I would like to thank them and acknowledge their contributions in my study.

On a personal basis, this journey has its ups and downs. Periods of joy, laced with emotional turmoil and challenges. Family members have been my beacons in this arduous journey. I would like to acknowledge them here, as they are my source of inspiration and hope when things are bleak.

Thank you all for your kindness, help and support during this PhD journey!

TABLE OF CONTENTS

Abst	ract	iii
Abst	rak	iv
Ackn	nowledgements	V
Table	e of Contents	vi
List (Of Figures	xi
List o	of Tables	xii
List o	of Symbols and Abbreviations	xiv
List o	of Appendices	xvi
СНА	APTER 1: INTRODUCTION	1
1.1	Retirement Income in Malaysia	2
1.2	Statement of Problem	4
1.3	Research Model	12
	1.3.1 Development of Research Objectives and Research Questions	14
1.4	Research Methodology	19
1.5	Population of the Study	20
1.6	Contributions of the Study	20
1.7	Limitations of the Research	23
1.8	Future Research	24
1.9	Definitions of Key Terms	25
1.10	Overview	28

CHA	CHAPTER 2: LITERATURE REVIEW			
2.1	Financ	ial Planning for Retirement	29	
	2.1.1	Underpinning Theories	32	

2.2	Attituc	les and Beliefs	
2.3	Contex	xtual Environment	42
2.4	Percep	otions of control	47
	2.4.1	Cognitive Requirements	47
	2.4.2	Time elements	53
	2.4.3	Resources	55
2.5	Theore	etical Framework	60
	2.5.1	Attitudes and Behavioural Beliefs	
	2.5.2	Subjective Norms and Normative Beliefs	65
	2.5.3	Perceived Behavioural Control and Control Beliefs	68
	2.5.4	Intentions	71
2.6	Hypot	heses Development	72
2.7	Summ	ary	76
CHA	APTER	3: RESEARCH METHODOLOGY	78
3.1	Develo	opment of Instrument	79
	3.1.1	Adapted items	
	3.1.2	Qualitative Interviews	85
	3.1.3	Additional Information	97
3.2	Goodn	ness of Measures	
	3.2.1	Expert Validation	
	3.2.2	Validation for Behavioural Beliefs	
	3.2.3	Validation for Normative Beliefs	104
	3.2.4		104
		Validation for Control Beliefs	104
	3.2.5	Validation for Control Beliefs Validation for Items Adapted	104
3.3	3.2.5 Pilot S	Validation for Control Beliefs Validation for Items Adapted	104
3.3	3.2.5 Pilot S 3.3.1	Validation for Control Beliefs Validation for Items Adapted Studies Results of the Pilot Study	104

3.4	Ethics .	Approval	.115
3.5	Populat	tion and Sampling	.115
3.6	Data co	ollection	.117
3.7	Prelimi	nary Data Analysis	.118
	3.7.1	Data Screening	.118
	3.7.2	Exploratory Factor Analysis	.121
	3.7.3	Common Method Variance	.127
3.8	Descrip	otive Analysis	. 128
3.9	Structu	re Equation Modelling	.129
	3.9.1	Path models	. 129
	3.9.2	Covariance based and Variance based Techniques	.132
	3.9.3	Justification for Using PLS-SEM	.136
	3.9.4	Evaluation Using PLS-SEM	.138
	3.9.5	Evaluating the Measurement Model	.138
	3.9.6	Evaluating the Structural Model	.142
3.10	Summa	ıry	.147
СНА	PTER 4	4: RESULTS	.148
4.1	Results	of Preliminary Analysis	.148
	4.1.1	Data Screening	. 148
	4.1.2	Exploratory Factor Analysis	.151
	4.1.3	Common Method Variance	.156
4.2	Respon	dents' Profile	.157
4.3	Descrip	otive Analysis	.167
4.4	Partial	Least Square (PLS) SEM Assessment	. 192
4.5	Measur	rement Model Assessment	. 192
	4.5.1	Convergent Validity	. 193

	4.5.2	Internal Consistency Reliability		
	4.5.3	Discriminant Validity	195	
4.6	Structu	ıral Model Assessment		
	4.6.1	Path Coefficients		
	4.6.2	Effect Size (f^2)		
	4.6.3	Coefficient of Determination (R^2)	201	
	4.6.4	Predictive Relevance (Q^2)	202	
4.7	Hypoth	neses Results		
4.8	Summa	ary		
CHA	APTER	5: DISCUSSION		
5.1	Demog	graphic Results		
5.2	Data A	nalysis	210	
	5.2.1	Influence on Intentions	210	
	5.2.2	Influence of beliefs	217	
5.3	Summa	ary	224	
CHA	APTER	6: CONCLUSION		
6.1	Contril	butions		
	6.1.1	Empirical Contributions	227	
	6.1.2	Theoretical Contributions		
	6.1.3	Methodological Contributions		
6.2	Practic	al Implications	231	
6.3	Limita	tions	232	
6.4	Future	Research	233	
6.5	Concluding Comments			
REF	ERENC	ES		

List Of Publications And Papers Presented	
1	
Appendix	

university

LIST OF FIGURES

Figure 1.1:	Theory of Planned Behaviour	14
Figure 2.1:	Processual depiction of planning	31
Figure 2.2:	Research Hypotheses	76
Figure 3.1:	Measurement and Structural Model	.130
Figure 3.2:	Reflective and Formative Constructs	.132
Figure 3.3:	Structural Model evaluation procedures	.142
Figure 4.1:	Respondents' age profile	.158
Figure 4.2:	Respondents' education background profile	.159
Figure 4.3:	Respondents' gender profile	.160
Figure 4.4:	Respondents' ethnicity profile	.161
Figure 4.5:	Respondents' religion profile	.162
Figure 4.6:	Respondents' marital status profile	.163
Figure 4.7:	Respondents' occupation type profile	.164
Figure 4.8:	Respondents' annual personal income profile	.165
Figure 4.9:	Respondents' monthly household debt repayment profile	.166
Figure 4.10	: Respondents' satisfaction with current financial situation	.167
Figure 4.11	: PLS Results of the Path Analysis	.205

LIST OF TABLES

Table 3.1: Adapted Scale items for attitudes	81
Table 3.2: Adapted Scale items for subjective norms	82
Table 3.3: Adapted Scale items for Perceived Behavioural Control	83
Table 3.4: Adapted Scale items for intentions	84
Table 3.5: Summary of Adapted Scale Items	85
Table 3.6: Normative beliefs identified from preliminary interviews	93
Table 3.7: Summary of Content Validation by experts	107
Table 3.8: Results of reliability analysis (pilot study)	114
Table 3.9: Summary of survey forms received	118
Table 3.10: Summary of Extraction Procedures	124
Table 3.11: Summary of Rotational Techniques	125
Table 3.12: Comparison between PLS-SEM and CB-SEM	136
Table 3.13: Evaluation of reflective models using PLS-SEM	138
Table 3.14: Guidelines for Assessing Reflective Measurement Model	141
Table 4.1: Missing data	149
Table 4.2: Non applicable responses	150
Table 4.3: Collinearity Statistics	151
Table 4.4: Initial theoretical framework and the 8-factor model	153
Table 4.5: Results of Harman's single-factor test	157
Table 4.6: Descriptive statistics for behavioural beliefs	172
Table 4.7: Descriptive statistics for Normative beliefs	178
Table 4.8: Descriptive statistics for control beliefs	183
Table 4.9: Descriptive statistics for attitudes	185

Table 4.10: Descriptive statistics for subjective norms	188
Table 4.11: Descriptive statistics for Perceived behavioural control	190
Table 4.12: Descriptive statistics for intentions	192
Table 4.13: Convergent validity analysis	194
Table 4.14: Internal consistency reliability	195
Table 4.15: Cross loadings output	196
Table 4.16: Fornell-Larker's criteria	197
Table 4.17: HTMT values	198
Table 4.18: Path Coefficients, t-Statistics and p-value hypothesized path	200
Table 4.19: Effect size	200
Table 4.20: Coefficient of Determination (R^2)	201
Table 4.21: Predictive relevance	202
Table 4.22: Results of Hypothesis Testing	204

LIST OF SYMBOLS AND ABBREVIATIONS

ATT Attitude : AVE Average Variance Extracted : BB Behavioural Beliefs : BI Behavioural Intent : **Control Beliefs** CB : **CBSEM** Covariance Based Structural Equation Modelling : CMV : Common Method Variance CR Composite Reliability : Content Validity Index CVI : DN Descriptive Norms : DNB **Descriptive Normative Beliefs** : EPF : Employee Provident Fund Expectancy Value Theory EVT : f : Effect Size FOMCA : Federation of Malaysian Consumers Association HTMT Heterotrait-Monotrait ratio : IN Injunctive Norms INB Injunctive Normative Beliefs INT Intentions : KMO Kaiser-Meyer-Olkin : Life-Cycle Hypothesis LCH : MOE Ministry of Education : MQA Malaysian Qualifications Agency : NB Normative Beliefs :

- OLS : Ordinary Least Squares
- PAF : Principal Axis Factoring
- PBC : Perceived Behavioural Control
- PLS : Partial Least Squares
- Q^2 : Predictive Relevance
- R^2 : Coefficient of Determination
- RM : Ringgit Malaysia
- SEM : Structural Equation Modelling
- SN : Subjective Norms
- SPM : Sijil Pelajaran Malaysia
- SPSS : Statistical Package for Social Sciences
- TACT : Target, Action, Context, Time
- TPB : Theory of Planned Behaviour
- TRA : Theory of Reasoned Action
- VIF : Variance Inflation Factor

LIST OF APPENDICES

Appendix A:	Elicitation Interviews	266
Appendix B:	Definition, Coding And Categories	269
Appendix C:	Content Validation For Behavioural Beliefs	275
Appendix D:	Content Validation For Normative Beliefs	277
Appendix E:	Content Validation For Control Beliefs	278
Appendix F:	Content Validation For Attitudes	280
Appendix G:	Content Validation For Subjective Norms	281
Appendix H:	Content Validation For Perceived Behavioural Control	282
Appendix I:	Content Validation For Intentions	283
Appendix J:	Sample Survey Instrument	284
Appendix K:	Results of Pilot study	292
Appendix L:	Results of Preliminary Qualitative Interviews	304
Appendix M:	Profile of Survey Respondents	307
Appendix N:	Frequency Tables	310
Appendix O:	Indicator loadings	317
Appendix P:	List of Public and Private Universities in Malaysia	318
Appendix Q:	Results of Exploratory Factor Analysis	321

CHAPTER 1: INTRODUCTION

Retirement is a period of a person's life when he or she is no longer working. Denton and Spencer (2009, p. 72) emphasise this period as a period where "... an older person ... has a much reduced labour force attachment". Bowlby (2007, p. 17) summarises it as "...a person who is aged 55 and older, is not in the labour force, and receives 50 % or more of his or her total income from retirement-like sources". A person may voluntarily choose to retire or is forced to retire due to debilitating physical conditions or as a result of legislation concerning their position. To many, being able to work is important, as it generates income to sustain their lives. Hence, not working is worrisome, as the income stream for sustenance ceased and alternatives means of support has to be available. To understand the various means of income support during retirement, World Bank (1994) has depicted it in a threepillar system. It reflects a three-legged stool, with each pillar outlining initiatives to protect the retiring individual.

The first pillar emphasises the importance of government initiatives to provide a pension to sustain basic needs and reduce poverty. In some countries, it is a form of social security which provides basic but universal coverage. It reflects efforts by the government to provide social protection, from the tax revenues collected, for its citizens. However, many of these countries also witness a demographic phenomenon, where the proportion of the aged to total population has grown tremendously in recent years (United Nations, 2002). Growth in the aged population has stressed the resources of countries, as there is a smaller working population to fund a growing group of retirees. Countries that provide pension and social security worry if they will have enough resources to continue to provide these financial supports (Morgan & Eckert, 2004). As a result, some governments are increasingly shifting responsibilities to individuals. For example, many pension funds

require workers to manage their retirement income actively by investing to generate future income (Smeeding, 2007).

Pillar two represents employers who provide pension plans for their employees. These plans were initially designed as a reward for loyal older workers (McDonald & Donahue, 2011). Both employers and employees will be required to make contributions, and with it, the monies will be used to generate funds for future retirement income (Morgan & Eckert, 2004). However, weak economic conditions have lead many employers to discontinue such practices. Some older workers are forced into early retirement as organisations begin cost-cutting initiatives. Employers who have previously provided pensions via defined benefits systems have also shifted towards defined contributions systems, which require individuals to manage the funds themselves (Smeeding, 2007; Hershey, Jacobs-Lawson & Austin, 2013).

Pillar 3 reflects individual initiatives to supplement their retirement income. It is mainly voluntary in nature, and involve efforts by the individual to manage their wealth. It encompasses savings, investments and insurance purchases from public or private providers (Morgan & Eckert, 2004). In view of global developments affecting Pillar 1 and 2, these voluntary initiatives are becoming increasingly important.

1.1 Retirement Income in Malaysia

In this study, the target population are Malaysians currently working in universities. Universities provide education, which is a priority in Malaysia's roadmap to development. It received the highest allocation of resources (Ministry of Finance, 2020) and is the focus of the government national agenda. In addition, universities educate future generations that will lead the country. Hence, the education industry has an important role in the Malaysia. Universities are under the jurisdiction of the Ministry of Education (MOE) of Malaysia. The MOE oversees all the higher education institutions and government agencies involved in higher education activities such as the Malaysian Qualifications Agency (MQA), the National Higher Education Fund Corporation (Perbadanan Tabung Pendidikan Tinggi Nasional – PTPTN) and the Tunku Abdul Rahman Foundation (Yayasan Tunku Abdul Rahman).

The providers of higher education in Malaysia can be grouped into two major categories. The first consists of public universities, largely funded by the government. The Malaysian government offers a pension scheme for its core employees. This scheme mirrors a form of defined benefit plan, where upon retirement, an employee is entitled to a lifelong monthly payment calculated based on the number of years of service and the last drawn basic pay. It is wholly borne by the federal government through annual allocation from the federal budget. The scheme is quite generous, but in view of the rising share of government employees in the labour force, there are risks of fiscal sustainability (Sim & Hamid, 2010). With this in view, public universities employees are also offered alternative routes to retirement, such as savings through the Employees Provident Fund (EPF).

EPF is a trust fund established to provide a long-term retirement savings route for all employees in the private sector and non-pensionable employees in the public sector. Its retirement savings accounts are similar to defined contribution plans, where the contributions are based on a prescribed rate for employees and employers. These contributions are invested and returns earned are credited to the individual's account. The accumulated contributions can be fully withdrawn upon retirement. In 2009, an estimated 20 public universities were located in various states in Malaysia (Malaysian Qualifying Agency, 2009). However, these public universities could not provide for the growing number of students in Malaysia. Public universities resources are also limited, as they depend on funds provided by the government.

To cater for the increasing number of students to ensure that opportunities are created for students who were unsuccessful in gaining entry into public universities, the MOE has encouraged the growth of the second category of provider of higher education, the private higher educational institutions. They are funded by private means, such as investments from corporations or alumni. These institutions are mandated to contribute to EPF.

In summary, university employees are covered by the formal social protection schemes in Malaysia which consist of the pension scheme and EPF. In addition, individuals may also supplement their retirement income by saving or investing. These initiatives are voluntary. They include savings or fixed deposits in banks and investments in unit trust or capital markets. There are also private insurance schemes (especially life and medical insurance) and savings that Malaysians, regardless of age, could subscribe to. Such efforts represent voluntary initiatives by the individual to financially plan and prepare for their retirement.

1.2 Statement of Problem

Voluntary initiatives to financially plan and prepare for retirement encompass a series of steps used by an individual to progressively attain retirement goals. This would involve various activities to organize a person's finances such as evaluating the suitability of banking products to a person needs (checking, savings accounts, credit cards and consumer loans), investments, (stock market, bonds, mutual funds) insurance (life insurance, health insurance, disability insurance), participation and monitoring of private retirement plans and income tax management.

Researchers have noted that initiatives towards financial planning for retirement is moving at a slow pace (KRC Research, 2016; Employee Benefit Research Institute, 2018). Should this continue, there would be inadequate financial protection for a person in retirement. Traces of such inadequacy can be seen in the low saving rates in many countries (Brady, 2010; Kennedy & Matwijiw, 2010; MacDonald, Jones, Morrison, Brown & Hardy, 2011; Croy, Gerrans & Speelman, 2012; Ellen, Wiener & Fitzgerald, 2012; Pfau & Kariastanto, 2012). For example, Britain have reported a sharp decline since 1990s. Countries like United States, Japan and Australia also reported falling rates (Organisation for Economic Co-operation and Development, 2016).For Malaysia, EPF has reported that individuals will need a basic savings of RM196,800 by the age of 55, to sustain their retirement. However, in 2013, 69% of members have savings of RM50,000 and below while 54% have savings of RM20,000 and below, and only 31% are still working at the age of 54 years. At these rate, the amount saved will not be able to support the retiree during their retirement period (Employees Provident Fund, 2015).

This situation is made worse with social changes to the traditional family unit. Family provides various form of social support of emotional (nurture), informational (advice) and tangible (financial assistance) in nature. For many developing countries, it is not only an important form of old age support, but the only form of support (Holzmann, Mac Arthur & Sin, 2000; Holzmann & Hinz, 2005; World Bank, 2008; Antolín & Stewart, 2009).

However, changing social conditions, such as the increase in single-parent families, the dramatic increase in female employment and the geographic dispersion of families, has resulted in many family members not being able to provide the care required to an ageing person. Sim and Hamid (2010) reported a drop in the number of older Malaysians live in extended family households and an increase in the number of nuclear family households. Similarly, Tengku Aizan and Chai (2013) recognises intergenerational assistance as an important source of old age support, but note its decreasing trend in recent years.

With such challenges, it is unsurprising that achieving financial security in retirement may be an illusion to some. Countries have reported that their citizens seems paralysed to inaction at the insurmountable amount of work that they have to do to plan for their retirement future (Harrison, Waite & White, 2006). Some seems uninterested and unmotivated in managing their retirement affairs. It is unclear whether such attitudes are a distraction to their current unfortunate financial state or merely because retirement planning is not a priority in their lives. Whatever the reasons are, the reality is that there will be adverse outcomes to such inactions. In fact, many countries have reported that their retirees have to continue to work, more often because they cannot afford to retire fully (McDonald & Donahue, 2011; Hellevik & Settersten, 2013).

Research into this area are attempted in a fragmented way. There is lack of data on why and how people plan, not only in Malaysia but worldwide (Sim & Hamid, 2010; Hershey et al., 2013; Börsch-Supan, Bucher-Koenen, Coppola & Lamla, 2015). One popular focus of these research is on the demographics of the individual. Age frequently received attention in many studies. This is because as one grows older, the level of preparation increases (Hershey, Jacobs-Lawson, McArdle & Hamagami, 2007; Hershey, Henkens & Van Dalen, 2007; Padawer, Jacobs-Lawson, Hershey & Thomas, 2007). The fact that they are drawing nearer to retirement makes it more immediate. Gender also has its influences, with men reported to be more likely to engage in financial planning. The income level is another determinant (Ekerdt, De Viney & Kosloski, 1996; Ekerdt, 2010; Ekerdt & Baker, 2014). Those more highly educated with a permanent position can plan more effectively. In summary, older educated males earning a higher income are reported to be more actively engaged in financial planning. Frequently, individuals with this demographic profile are afforded with more opportunities in life (Ekerdt et al., 1996; Petkoska & Earl, 2009; Ekerdt & Baker, 2014).

However, the use of demographic indicators have been critique. Hershey et al. (2013) states that these indicators are not explanatory variables and have limited use in theory development. Mechanism that underlies the association of planning with age, gender, income and education is lacking. Nonetheless, they are useful in developing interventions that target those at risk of poverty in old age and developing public policy initiatives aimed at encouraging individuals to plan and save. On the other hand, Fishbein and Ajzen (2010); Ajzen and Klobas (2013) opines that these variables should be considered as background factors, which influence intentions and behaviours indirectly via beliefs.

However, to assume that individuals do not plan at all may be too premature, as there are also studies that have shown that individuals do not remain idle in terms preparing for their retirement needs. In a survey conducted by (Princeton Survey Research Associates International, 2013), nine out of ten households in America are reported to be involved in some form of planning. Overall, almost 90% of its respondents indicate that they have done some form of planning. Those that take the initiatives to plan, experience positive outcomes, especial during times of economic uncertainty.

Unfortunately, the adequacy of their plans are questionable, in particular, its sufficiency in meeting the long-term needs of a retiree (Morgan & Eckert, 2004). Further analysis reveal that the degree of engagement in planning differs. Findings reveal that 19% are comprehensive planners, 38% are basic planners, 33% are limited planners, and 10% are non-planners. Comprehensive planners have a complete financial plan, have worked with professionals from the financial industry, and are somewhat confident in their financial choices. Basic planners may or may not have a complete plan. Only about 31% reported that they are more likely to plan in the following year (Princeton Survey Research Associates International, 2013).

In addition, the goals set are unclear. Only 80% have some specific goal in their savings plan. About 66% mentioned that they have a budget. For limited planners, only 11% are likely to make a plan in the following year, while 1% have a comprehensive plan. 44% have a budget, with plans to have at least one savings goal. 61% report that they do not have any credit card or credit card debt. As for non-planners, none of the respondents have a comprehensive plan or a budget. 38% have high credit card debt. Unfortunately, only 47% have a clear plan to reduce their debt.

Similarly, Brucker and Leppel (2013) reported that individuals that have planned had not executed their retirement plans well enough. Their study reveals that only 44% of the respondents have indicated that they had a financial plan with specific goals for retirement. For those within five years of expected retirement, about 50% had a plan. At the same time, based on a study of individuals over the age of 50, Lusardi and Mitchell (2011a) found that 31% of respondents had attempted to determine what they needed to save for retirement. However, of those who tried to do so, only 67% created some plan. For those who developed a plan, 88% followed through with it. The study also reveals an average of 19% success rate in the planning initiatives. The authors highlighted that the low success rate is due to the difficulties in planning for retirement. For those who attempted it, many are not very clear on what they intend to do to achieve their goals.

The research conducted also documented a range of feelings ranging from positive feelings of optimism, confident, secure, content, happy to negative feelings such as uncertain, concern, worried, anxious and afraid of financial planning. KRC Research (2016) discovered that respondents in their survey can be categorised to four roughly equal segments based on their feelings towards various aspects of financial planning. These segments are labelled as Confident Savers (22%), Concerned Strivers (27%), Tentative Savers (24%) and Stretched Worriers (26%). Overall, 36% were optimistic, while 28% were uncertain and another 28% were concerned. Most of the optimism is from the Confident Savers and Concerned Strivers segments. The Stretched Worriers segments are struggling to stay current with bills and paying off debt and have not made much savings. At the same time, Confident Saver, Concerned Strivers and Tentative Savers have voice concern about the challenges they face in managing their credit card debt and mortgages.

Those who plan are more confident about their financial abilities. They save more and are progressively meeting their saving goals. Such sentiments are reflected in a survey by Employee Benefit Research Institute (2018). 64% of workers are reported to be confident (17%) or somewhat confident (47%) of their ability to live comfortably during retirement. However, confidence is inadequate. It must be accompanied by action. Equally of importance, findings have also revealed that respondents to the survey made assumptions which may give rise to a false sense of confidence. Some of these workers expect to retire later than retirees do. 68% expect to work for pay to provide them income in retirement. Also, those without a retirement savings plan are not confident of having enough resources for their old age.

Similarly, many individuals believe that regular contributions made towards their retirement accounts will provide a regular income stream during retirement. However, in a study by Cohen (2014) on superannuation retirement balances in Australia, it is discovered that these assumptions are questionable. Many retirees, despite making regular contributions, find their retirement balances below expectations (Brady, 2010; Kennedy & Matwijiw, 2010; MacDonald et al., 2011; Croy et al., 2012; Ellen et al., 2012; Pfau & Kariastanto, 2012). Findings reveal that while regular contributions have its benefits, what is more important is for the individual to actively manage and monitoring their contributions. Such action will enable early interventions if the amount accumulated fall below the desired target. In summary, it is no longer possible to rely on prudent behaviour alone to achieve retirement objectives. Instead, the individual is required to be familiar with the factors in the environment they live in that will affect their welfare and that of others during retirement.

Economic conditions are not the only influences on a person's retirement decisions. Croy, Gerrans and Speelman (2010a) and Croy et al. (2012) also identified several important social referents in a person's contextual environment, such as spouse, employers and financial advisers, that influence retirement savings decisions. Koposko and Hershey (2014); Van Campenhout (2015); Koposko and Hershey (2016); Koposko, Kiso, Hershey and Gerrans (2016) also emphasised the importance of parents, peers and early influences on a person's savings and retirement decisions. However, different populations are influence by different groups of social referents.

Briefly, financial planning for retirement involved budgeting, savings and money management, whilst taking into consideration various financial risks and future life events. The complexities in planning is demanding, both cognitively and motivationally. A survey investigating the financial capability and literacy among Malaysians reported that a

majority display inadequate knowledge on financial matters, particularly among vulnerable groups (Central Bank Of Malaysia, 2018). Most of them have short term tendencies, ignoring the possible long-term benefits of financial planning. Only about 40% are financially ready for retirement. Most of the respondents tend to rely on children or debt for emergency or expensive items. 75% have difficulty in raising money to meet emergency needs. Also, only 25% have made some form of investment. Many also reveal that they would face tremendous financial pressure if they lose their current source of income. The survey also reveals that most Malaysians have not been taking proactive financial decisions that are important for their well-being. Although most of them are worried about retirement, they adopt a passive strategy on retirement planning. They prefer to rely on statutory bodies such as EPF or possible welfare and social protection during old age.

Another survey by the Federation of Malaysian Consumers Associations (FOMCA) (2016) reveals similar results. It stressed the importance of thinking about retirement. Findings reveal that 37% of respondents have never thought about retirement. For those who have thought about retirement, 60% were either not satisfied or only somewhat satisfied.

In summary, studies in this area show mixed results and appear fragmented. Some individuals are thoroughly prepared for retirement while others are either at a loss as to what to do or have no future plans (Selnow, 2004; Adams & Rau, 2011). Still many are also unable to plan due to economic or social challenges (Holzmann et al., 2000; Holzmann & Hinz, 2005; World Bank, 2008; Antolín & Stewart, 2009). Even for those that do plan, there are inconsistencies and weaknesses in their efforts (Lusardi & Mitchell, 2011a; Brucker & Leppel, 2013; Princeton Survey Research Associates International, 2013; Central Bank Of Malaysia, 2018). Some hold false assumptions or confidence about

retirement (Cohen, 2014; Employee Benefit Research Institute, 2018). What influence their actions (or inactions) on financial planning for retirement are unclear.

As a result, this study aims to investigates the determinants of intentions to plan financially for retirement and provide more significant insights into the planning done for a person's later life in retirement, within a selected sample in Malaysia, namely university employees. The key objective is to examine the influence of the beliefs, attitudes, subjective norms and perceptions of control on intention to plan financially for retirement within this population.

1.3 Research Model

To understand the various determinants and processes involved, this study bases its conceptual framework in the Theory of Planned Behaviour (TPB). TPB assumes that financial planning behaviours are a result of a reasoned decision. This means that behaviour is formed in a reasonable, consistent and often automatic fashion from the beliefs people possess. Beliefs guide a person's intentions to perform (or not perform) the behaviour in question (Fishbein & Ajzen, 1975; Fishbein & Ajzen, 2010)

According to TPB, three kinds of beliefs are distinguished. The first are behavioural beliefs. These beliefs refer to the perceived positive or negative consequences of a behaviour (such as planning financially for retirement) and its subjective values or evaluations. In aggregate, behavioural beliefs lead to the formation of a positive or negative attitude toward a behaviour. Generally, if the performance of the behaviour is perceived to result in positive outcomes, the attitude toward the behaviour will be favourable (Ajzen & Fishbein, 2005; Ajzen, 2015).

The second is normative beliefs. These beliefs reflect the perceived expectations and behaviours of important referent individuals or groups. Combined with motivation to comply with the referent in question, they produce perceived social pressure or subjective norm with respect to a behaviour. it is assumed that if more important others are believed to approve, and are performing the behaviour, people are likely to perceive social pressure to engage in the behaviour in question. The opposite would then happen for beliefs that are not favoured by the significant people in a person's life (Fishbein & Ajzen, 1975)

Third is the control beliefs. These beliefs refer to the perceived presence of factors that influence the ability of a person. Together with the perceived power of these factors, control beliefs produce perceived behavioural control. If control beliefs identify more facilitating than inhibiting factors, perceived behavioural control should be high and vice versa.

The attitudes, subjective norms and perceived behavioural control form behavioural intentions, or a readiness to perform the behaviour. Generally, the more favourable the attitude and subjective norm, and the greater the perceived behavioural control, the stronger the person's intentions. However, the relative importance of these determinants is expected to vary from one behaviour to another and from one population to another. A schematic representation of TPB is shown below.



Figure 1.1: Theory of Planned Behaviour

1.3.1 Development of Research Objectives and Research Questions

In TPB, attitudes, subjective norms and perceived behavioural control are the direct antecedents to intentions. Attitudes reflect the degree to which the behaviour is positively or negatively valued. However, the influence of attitudes on intentions to plan financially for retirement are mixed. Research has identified a myriad of attitudes from positive feelings such as optimism, confident and happy to negative feelings such as uncertainty, worried, anxious and afraid in financial behaviour research (Princeton Survey Research Associates International, 2013; KRC Research, 2016). Some individuals view planning for retirement as highly necessary while others could not plan due to lack of resources or are just uninterested (Petkoska & Earl, 2009; Brucker & Leppel, 2013; Segel-Karpas & Werner, 2014; Van Deventer, De Klerk & Bevan-Dye, 2014). Hence, it is important to examine the influence of attitude on intentions to plan for retirement among University employees. Thus, the following research objective and question is developed:

<u>Research Objective 1</u> - To examine the influence of attitudes on the intention to plan financially for retirement among University employees

<u>Research Question 1</u> - What is the influence of attitudes on the intention to plan financially for retirement among University employees?

Subjective norms reflect the perceived social pressure to engage or not to engage in a behaviour. Social pressure comes from the social environment, particularly from the various important persons in an individual's life. These are strong influences not limited to friends or family. At times, society members, organisations and cultural groups also play important roles.

This determinant has an injunctive and descriptive component. Injunctive norms are concerned with gaining approval (or suffering from disapproval) from significant others in relations to the behaviour of study (Cialdini, Demaine, Sagarin, Barrett, Rhoads & Winter, 2006; Cialdini, 2007). It deals with one's perception of what others believe to be appropriate conduct and is different from one's personal opinion or reflections. Descriptive norms are related to one's perception of what significant others do in relations to the behaviour of study (Cialdini, 2007; Fishbein & Ajzen, 2010; Ajzen & Klobas, 2013; Klobas & Ajzen, 2015).

How these norms influence a person in financial and retirement planning is unclear. Earlier studies have revealed that subjective norms has a weak effect of intentions (Trafimow & Findlay, 1996). However, later studies reveal that differences are expected. Different population and context of study should exhibit different influences (Fishbein & Ajzen, 2010; Klobas & Ajzen, 2015). Also, initial studies considered only the injunctive component in subjective norms (Fishbein & Ajzen, 1975; Ajzen & Fishbein, 1980; Fishbein & Ajzen, 2010). However, these earlier interpretations are deemed inadequate, as later studies have proven that both norms are important (Cialdini, 2007; Fishbein & Ajzen, 2010; Ajzen & Klobas, 2013; Klobas & Ajzen, 2015). On the onset, the norms display similarities, but upon more indepth analysis, they differ conceptually and motivationally. Injunctive norms motivate a person into action via social evaluation, but descriptive norms motivate by providing social information. Most people perceive a behaviour socially adopted by the majority as a wise and effective form of conduct.

With the objective to explain the influence of subjective norms on the intentions to plan financially for retirement among university employees, the following research question is developed:

<u>Research Objective 2</u> - To explain the influence of subjective norms on the intention to plan financially for retirement among University employees <u>Research Question 2</u> - What is the influence of subjective norms on the intention to plan financially for retirement among University employees?

Early developments of this model have focus on attitudes and subjective norms. At that time, the authors had referred this theory as the Theory of Reasoned Action (TRA) (Fishbein, 1963; Fishbein & Ajzen, 1975; Ajzen & Fishbein, 1977; Ajzen & Fishbein, 1980). However, TRA could not be applied to all situations and behaviour (Beck & Ajzen, 1991). In many studies, these determinants could only explain approximately 10% of the variance in behaviour (Ajzen & Fishbein, 2000). Ajzen and Fishbein (1980); Ajzen (1985) account this as due to the fact that many goals and behaviours are not under complete volitional control of a person. In recognition of this fact, Ajzen and Fishbein (1980); Ajzen

(1991) introduced an additional determinant, perceived behavioural control. This extension to TRA was subsequently called the Theory Of Planned Behaviour (TPB).

Perceived behavioural control refers to the extent to which an individual believes that they are capable of performing the target behaviour (Fishbein & Ajzen, 1975; Fishbein & Ajzen, 2010). This involves considering the presence or absence of factors that may facilitate or impede the performance of the behaviour, and whether the individual has perceived control over these factors.

A majority of studies have shown that intentions can be predicted with considerable accuracy using perceived behavioural control (Armitage & Conner, 1999; Armitage & Conner, 2001; Ajzen & Albarracin, 2007). However, the influence of perceived behavioural control differs depending on the population and behaviour. There are studies where perceived behavioural control have no direct correlation with intentions. Similarly, there are reports that it has exhibited moderating effects on attitudes and subjective norms (Fishbein & Ajzen, 2010). Hence, to determine the influence of perceived behavioural control on the intention to plan financially for retirement among University employees, the following research question is developed:

<u>Research Objective 3</u> - To determine the influence of perceived behavioural control on the intention to plan financially for retirement among University employees <u>Research Question 3</u> - What is the influence of perceived behavioural control on the intention to plan financially for retirement among University employees?

Attitudes, subjective norms and perceived behavioural control can be analysed further to its antecedents. A person holds many beliefs but only those that are accessible explain intentions and behaviours. The accessible beliefs depend heavily on the population of study. For example, behavioural beliefs are formed by associating a behaviour, such as financial planning for retirement, with various characteristics, qualities and attributes experience in lifespan of a person. Similarly, normative beliefs are beliefs regarding the prescriptions and behaviours of referents deemed important to the individual. Control beliefs, in turn, are beliefs about the resources (or opportunities) and obstacles (or impediments) individuals think they possess. All these beliefs may be formed directly from observation; indirectly from various sources or may be self-generated through inference processes. Beliefs are dynamic; they may persist over time, be forgotten and new beliefs can be formed.

Hence, in order to draw insights into the attitudes, subjective norms and perceived behavioural control of retirement planning behaviours, it is important to obtain information on the accessible beliefs of the population of study. These beliefs are identified in formative research where accessible outcomes are elicited in a free-response format from a representative sample of the population. The beliefs most frequently mentioned are then included in the survey of the main study.

Hence, to identify and to determine the influence of these accessible beliefs among university employees, the following research objective and question is developed:

<u>Research Objective 4</u> - To determine the influence of beliefs on attitudes, subjective norms and perceived behavioural control among University employees <u>Research Question 4</u> - What is the influence of beliefs on attitudes, subjective norms and perceived behavioural control among University employees?

1.4 Research Methodology

This study employs quantitative research methods to answer the research questions. The data is collected using a survey method. In developing the items in the survey instrument, two separate methods are used (Morgan, 1998; Tashakkori & Teddlie, 1998; Fishbein & Ajzen, 2010). Items to answer research questions 1, 2 and 3 (attitudes, subject norms and perceived behavioural control) are adapted from existing validated scales from the literature whilst items addressing research question 4 (behavioural beliefs, normative beliefs and control beliefs) is developed via formative research which consist of qualitative interviews. From these interviews, the accessible beliefs are elicit (Ajzen, 1991; Ajzen, 2006). The frequently cited beliefs are then included in the survey. The goodness of the measures developed are subjected to validity and reliability checks. To ensure the validity of the survey, a team of experts is recruited. This team assesses each item in the survey instrument based on its consistency, representativeness, relevance and clarity (Lynn, 1986). As for reliability, a pilot study is conducted. The objective is to test the research methodology, survey instruments, data collection techniques and other research techniques in preparation for a more comprehensive study (Hassan, Schattner & Mazza, 2006; In, 2017).

Upon completion of data collection, all completed survey forms were checked, coded and entered into SPSS (Meyers, Gamst & Guarino, 2006). The data is then subjected to a series of preliminary analyses involving checking for missing values, normality, identifying outliers and collinearity issues. Descriptive analysis is conducted on demographic information such as age, education level, gender, ethnicity, religion, marital status and details of occupation. Subsequently, analysis using Structural Equation Modelling (SEM) via the Partial Least Squares (PLS) method is also used to determine the measurement properties and to testing the key theoretical relationships (Hair, Black, Babin & Anderson, 2010).

19
1.5 Population of the Study

The population selected for this study are universities employees in Malaysia. Universities in Malaysia can be grouped into public and private universities. These employees are covered by the formal social protection schemes in Malaysia, that is the pension scheme and EPF. Public universities offer two separate schemes to retirement for its employees, a pension scheme and savings through EPF. However, with the rising share of government employees in the labour force, eligibility for pension benefits have become more stringent in recent years. This has led to an increase in employees which have to opt for savings through EPF contributions (Sim & Hamid, 2010). As for private universities, they are mandated by law to contribute a prescribed rate to EPF. In addition to the formal social protection schemes, individuals may, voluntarily, supplement their retirement income by saving or investing. However, such efforts depend on income availability, resources and the initiative of the individual.

From the population, a simple random sampling process is used. SPSS will be used to generate the random numbers in the selection process. This method has the least bias and the most generalisability.

1.6 Contributions of the Study

Studies have demonstrated that the constructs in TPB have different effects in different contexts (Fishbein & Ajzen, 2010; Ajzen & Klobas, 2013; Klobas & Ajzen, 2015). Likewise, the constructs in this study have different effects on planning intentions. Subjective norms and attitudes have positive effects, while perceptions of controllability have no influence on planning intentions.

Subjective norms, which reflect the perceived social pressure from important referents, play a stronger role in this study. The important referents for university employees are mainly family members (spouse, parents, siblings and relatives). For many Eastern countries, family members are a leading source of formal and informal retirement support. Despite changing social conditions (single-parent families, increase in female employment, geographic dispersion of families), the findings from this study reaffirms the role of family in providing retirement support.

In comparison, though attitudes towards financial planning are positive, it plays a limited role in motivating behaviour among university employees. Perceptions of controllability, in turn, have no influence on planning intentions. Such findings indicate that though university employees have positive attitudes, they are more reliant on the existing formal mechanism (pension, EPF) as a mean of planning and savings for retirement. This expands the body of knowledge on financial behaviours, providing empirical contribution to the study of retirement planning in Malaysia.

This study used TPB as its research model. It is a robust and well accepted model in social psychology, but its use in financial behaviour studies is limited (Bobek & Hatfield, 2003; Xiao & Wu, 2006; Xiao & Wu, 2008; Croy et al., 2010a; Croy, Gerrans & Speelman, 2010b; Cohen, 2014; Croy, Gerrans & Speelman, 2015; Heraty & McCarthy, 2015). Nonetheless, TPB is able to explain the findings of this study, which makes it a valuable model of planning in Malaysia. It has also extended the understanding of the psychological relationships in financial planning for retirement behaviours.

Financial planning for retirement is a complex form of behaviour, encompassing a spectrum of activities and a wide variety of financial products and services. In order to study this behaviour, a variety of approaches, methodologies and sources are used. The objective is to ensure the validity and reliability of the study.

Consequently, careful considerations were taken in the development of the survey instrument. Two separate methods are used to develop the items (Morgan, 1998; Tashakkori & Teddlie, 1998; Fishbein & Ajzen, 2010). Attitudes, subjective norms and perceived behavioural control were adapted from existing validated scales. Beliefs (behavioural beliefs, normative beliefs and control beliefs) were developed from qualitative interviews, as they are unique and should be elicited from the population of study (Fishbein, 1963; Ajzen & Fishbein, 1980). Further tests using PLS-SEM were used to ascertain that the set of modal beliefs developed were significant indicators. Such steps provide a richer view of the complexity and uniqueness of financial human behaviour.

To ensure validity of the measures, a team of experts had been recruited to evaluate the items in the survey instrument based on consistency, representativeness, relevance and clarity. Results of their evaluations were used to calculate the Content Validity Index (CVI) and the Kappa Coefficient (Polit & Beck, 2006). Using these two criteria, decision rules were set, and items that do not meet with the rules specified were eliminated.

Many studies on human behaviour have non-normal data distributions. This study encounters similar challenges. Nevertheless, steps were taken to choose a statistical program that can cater to data that have normality issues. PLS-SEM is chosen as it allows structural equation modelling to be used with non-normal data. Its use of 5,000 re-samples in the bootstrapping technique and the robust validation process enhances the level of confidence on the findings.

This study offers practical contributions to policymakers, financial planning practitioners and the financial services industry. In recent years, many countries have witnessed the demographic phenomenon of an aging population (United Nations, 2002). As a result, policymakers have begun efforts to shift the responsibilities of retirement support to individuals. However, studies have also reported that the individuals are not

prepared for such responsibilities. This study suggests that policymakers can use TPB as a framework to understand planning at an individual level. The policies developed would then be more effective and helpful to individuals (Cohen, 2014).

The financial planning industry is currently underdeveloped. Nevertheless, it has shown promising market potential in Malaysia. Unfortunately, trust levels are low, as many are sceptical towards the industry players' commitment to resolving their clients' financial woes. For the industry to develop, it is vital to understand the reservations and mindsets of their clients. TPB offers a framework into the consumers psychology when planning for retirement. It helps in understanding the determinants that motivate their clients, thus enabling the development and marketing of more appropriate financial products or services.

1.7 Limitations of the Research

Researchers often struggle between parsimony and complexity. Although the constructs in TPB were able to predict intentions to plan, they do not fully address the complexity of human behaviour (Ajzen, 1991). The study recognises that there are other possible variables that play a role in influencing intentions, but operationalising all the variables would complicate the analysis (Ogden, 2015).

Every data collection method has its pros and cons. Similarly, survey methods also have its limitations. For this study, the initial low response rate was a challenge. Subsequent attempts to generate higher responses were successful, but was bounded by the policy and procedural requirements of the respective institutions (to collect data at selected locations in the universities, during working hours). Some of the survey forms collected were also unusable and had to be excluded. The data collected were based on the thoughts and opinions of the respondents. As in studies that involve self-reports, elements of bias may occur. Individuals may not be entirely honest in their responses or have no recollection of what they did some time ago. Some also perceived the questions as sensitive and personal and were unwilling to participate. Nonetheless, strict measures were taken to provide complete anonymity and confidentiality to the respondents.

This study is cross-sectional. Cross-sectional data do not reflect how individuals may change their expectations as they evolve through the life-cycle. Nonetheless, many cross-sectional studies on TPB have useful and insightful discoveries (Notani, 1998; Armitage & Conner, 2001; Schulze & Wittmann, 2003; Rivis & Sheeran, 2004; Rivis, Sheeran & Armitage, 2009). In addition, Fishbein and Ajzen (1975); Ajzen (1991) had maintained that beliefs and attitudes tend to remain stable over time.

1.8 Future Research

This study adopts a reasoned action approach to understanding financial retirement planning. Its framework, TPB postulate a limited set of determinants (attitudes, subjective norms, perceived behavioural control) to explain behaviour (Fishbein & Ajzen, 1975; Ajzen & Fishbein, 1977; Ajzen & Fishbein, 1980).

Yet, a multitude of determinants is known to influence planning. Many researchers have also questioned the sufficiency of TPB determinants in addressing intentions and behaviours (Ajzen & Fishbein, 2005; Fishbein & Ajzen, 2010; Klobas & Ajzen, 2015). Even the original authors of TPB have recognised the need for further research in predicting, understanding and changing behaviours. Such efforts may include future investigations into the origins of beliefs, exploring various demographic and personality variables (Conner & Armitage, 1998; Armitage & Conner, 2001; Rivis et al., 2009; Fishbein & Ajzen, 2010).

Another area for future research involves subjective norms. The role of families, adult children and the extended family network to provide support for elderly people can be explored further (Yoo & Kim, 2010). In many Eastern countries like Malaysia, the concept of filial obligation and responsibility is greatly valued and is perceived as a duty of adult children (Walker, 2002). Yet, research in this area is limited, but with the growing importance of retirement, this is an important area for further investigations (Lapinski, Kerr, Zhao & Shupp, 2017).

The sample used in this study consists of University employees in Malaysia. However, a different group or segment of the population may engender different findings. Future research could examine the influence of selected variables on different groups of individuals. This may provide more information for various parties to tailor their services according to the needs of different segments of the population.

1.9 Definitions of Key Terms

(a) Accessible beliefs

Belief represents a state of mind. A person may deem something to be true, despite evidence of the contrary. It is a mental representation, where a person is oriented towards what they think is likely to be true (Schwitzgebel, 2006). Beliefs may be formed directly from observation; indirectly from various sources or may be self-generated through inference processes. They are dynamic and may persist over time, be forgotten or new beliefs can be formed. Many different beliefs are formed about an object, but only a relatively small number determine attitudes at any given moment. Beliefs about object that come readily to mind are known as salient beliefs (Fishbein & Ajzen, 1975; Ajzen & Fishbein, 1980). However, the term salience has been replaced by accessible in recent social psychology literature (Fishbein & Ajzen, 2010). This term reflects an important fact, which is that beliefs can be activated spontaneously. One may consciously or unconsciously activate it, but it should come readily to mind when a person has reason to retrieve them.

(b) Behavioural beliefs

Behavioural beliefs link a behaviour to the subjective probability that it will produce a given outcome. The combination of all accessible beliefs with the subjective values of the expected outcomes will determine the attitude towards the behaviour (Ajzen, 2019).

(c) Normative beliefs

Normative beliefs refer to the perceived behavioural expectations of important referent individuals or groups in the population of study. The combination of a person's motivation to comply with the different referents will determine the subjective norm (Ajzen, 2019).

(d) Control beliefs

Control beliefs have to do with the perceived presence of factors that may facilitate or impede the performance of a behaviour. The combination of the perceived power of each control factor will determine the prevailing perceived behavioural control (Ajzen, 2019).

(e) Attitudes

Attitudes is defined as "a mental and neural state of readiness, organised through experience, exerting a directive or dynamic influence upon the individual's response to all objects and situations with which it is related" (Allport, 1935, p. 810). Eagly and Chaiken (1998, p. 269) define attitude as a "psychological tendency that is expressed by evaluating

a particular entity with some degree of favour or disfavour". Ajzen (1991); Fishbein and Ajzen (2010); Ajzen (2019) further defined attitudes as the latent disposition or tendency to respond with some degree of favourableness or unfavourableness to a psychological object.

(f) Subjective norm

Subjective norm is the perceived social pressure to engage or not to engage in a behaviour. Social pressure comes from the social environment of a person. Subjective norm has an injunctive and descriptive component. Injunctive norms are concerned with gaining approval (or suffering from disapproval) from important groups or individuals whereas descriptive norms are related to one's perception of what important others do in relations to the behaviour of study (Ajzen, 2019).

(g) Perceived behavioural control

Perceived behavioural control refers to the extent to which a person believe that they are capable and that they have control over their ability to perform a given behaviour (Ajzen, 2019). It takes into account a person's perception of the availability of information, skills, opportunities and resources required to perform the behaviour as well as possible barriers or obstacles that may have to be overcome.

(h) Intention

Intention is an indication of a person's readiness to perform a given behaviour, and it is considered to be the immediate antecedent of behaviour (Ajzen, 2019).

1.10 Overview

This thesis comprises six chapters. Chapter 2 presents a review of the literature on financial behaviour. As empirical studies related to financial planning are limited, the literature review incorporates studies in other financial areas such as savings, insurance, investment and money management. Nevertheless, the focus will be on examining influences among constructs related to the theoretical framework developed. Related underpinning theories are also discussed. Subsequently, the theoretical framework is presented, which also draws up the justification for the development of the hypotheses.

Chapter 3 discusses the research methodology in detail. The development of the survey instrument is outlined. Details of the development of each item in the survey are provided. The items developed are subjected to reliability and validity tests. A pilot test is also conducted before data collection. There is a detailed discussion of SEM using PLS along with the statistical techniques and guidelines. The chapter also justifies the methodology.

Chapter 4 analyses the data collected. This includes the respondents' profile and descriptive analysis of the responses to the items in the survey. In addition, preliminary data analysis is also conducted. Subsequently, both the measurement and structural models are evaluated. The measurement model provides more output on the reliability and validity of the study. As for the structural model, the path coefficient and variances are tested.

Chapter 5 presents a detailed discussion of the results in the previous chapter. All the paths in the theoretical framework are discussed and explained. In explaining the results, previous literature is reviewed.

Chapter 6 concludes the study by outlining the contributions, limitations and suggestion for future research. Finally, all supporting materials, such as the interview questions, survey instruments and content validation information are provided in the appendices.

28

CHAPTER 2: LITERATURE REVIEW

One distinct characteristic of humans is the ability to plan. This ability helps individuals to overcome their immediate impulse, enabling them time to evaluate the situation before taking action. Plans reflect the mental representations of what a person wants to do and their commitment to the effort (Friedman & Scholnick, 1997b). This ability helps a person to forms directions and solve problems. It provides a purpose for life. However, planning is a complex process. A person is required to gather their cognitive, emotional and motivational resources in the service of reaching desired goals. Although mentally taxing and behaviourally challenging, planning is critical to many activities in a person's life. This study aims to focus on one aspect of planning which involve an important phase in a person life; that is financial planning for retirement. This chapter review the epistemology of financial planning, the empirical works of previous authors and describe the theoretical framework that will guide this study.

2.1 Financial Planning for Retirement

Financial planning owes many of its initial developments to practitioners in the financial services industry. These practitioners comprise of bankers, insurance personnel or agents. By the nature of their occupation, they provide advice in many areas of finance to help their clients. However, such advice is dispensed in a fragmented way. The focus of these practitioners was in solving their clients' immediate problems and their allegiance to their professions.

However, some of the activities of these early practitioners drew negative responses; not only among their clients but from fellow practitioners in their industry. Many of the industry leaders opine that there is a requirement to build professionalism among its members. There are fiduciary responsibilities in what financial planners do, which encompass beyond just selling a particular insurance or financial product. With this in mind, a standardised code of ethics and practice standards is introduced among financial planners.

However, in developing these standards, these early developers face a challenge in defining financial planning. Financial planning involves a spectrum of activities and a wide variety of financial products and services. Nevertheless, a final decision was made to adopt a process view of planning, where planning is depicted to follow a rational and orderly path. Briefly, it involves gathering data on a person's current financial status and other non-financial information. Financial status includes related financial information such as assets, liabilities, insurances and investments. Non-financial information includes information such as marital status, family background and education level. Goals are then set based on the information available. A detailed roadmap is outlined on how to achieve the goals set. This depiction of financial planning has been incorporated into professional financial planning bodies and organisations in many countries (International Organization for Standardization, 2005; Financial Planning Association of Australia, 2014; Financial Planning Standards Board, 2015).

This processual adaptation is robust and is similar to how other fields have depicted planning. For example, in the field of strategic planning for businesses, planning is defined as the formulation of long-term plans in a business environment. This is further elaborated as the managerial process of developing a vision, setting objectives, designing, implementing and executing a strategy (Pearce & Robinson, 2011). Overton (2008) supported this by stating that financial planning is a form of strategic planning. While strategic planning focuses on managing the organisation resources with the aim of maximising the profitability of a business, financial planning's objective is to help individuals achieve their goals with the proper management of their financial resources.



Figure 2.1: Processual depiction of planning

The definition of financial planning provides a guide for many practitioners in the industry. But in time, weaknesses arise. A person is assumed to have access to all the information required to plan. Furthermore, they also know the repercussions of the alternatives chosen and can effectively rank the options available. Unfortunately, such assumptions can only be possible with time and information that are accurate and complete. Grable, Archuleta, Roy and Roudi (2011) highlighted that this approach to planning is idealistic and difficult to achieve. It gives an illusion that by following a linear form of action, financial goals can be achieved.

Whether one is planning for an individual or an organisation, planning is rarely simple and orderly. This is obvious in other fields that have adopted the processual depiction of planning. For example, in the field of strategic planning, Mintzberg (1967); Mintzberg and Waters (1985); Mintzberg (1990); Mintzberg and Westley (1992) have reported that many strategies are not internalised or are misinterpreted by the organisation, resulting in high failure rates. Likewise, many financial planners report that financial plans or recommendations developed are not read or understood by clients. Clients frequently display a lack of commitment to planning (Koh & Fong, 2000).

Such developments signal the importance of further understanding financial planning. There is a need to know why, how and when a person plan. Theories in this field is lacking and it is vital that specific models, that can be applied consistently across different context and among different populations guide the practice and development of the field (Harrison, 2005; Beal & McKeown, 2008; Grable et al., 2011). Financial planning is a complex process, which encompass beyond monetary considerations and is greatly dependent on the goals a person hopes to achieve. A person, when planning, must use their cognitive, emotional and motivational abilities to achieve their desired objectives. Hence, there is a myriad of psychological and contextual determinants that could potentially influence planning behaviours that has yet been explored.

2.1.1 Underpinning Theories

In economics, a model frequently used to understand an individual's planning, savings and consumption patterns is the life-cycle hypothesis (LCH). LCH stipulates that people make consumption decisions based on several considerations. Firstly, they would evaluate the resources available to them over their lifetime and secondly, they would also consider the resource requirements of their current stage of life.

With this in mind, individuals intend to even out their consumption in the best possible manner over their entire lifetimes, doing so by accumulating when they earn and dis-saving when they are retired. The key assumption is that all individuals choose to maintain stable lifestyles at all times where they would build up assets at the initial stages of their working lives for it to be used later during retirement. Hence, people constantly save for retirement and will alter their consumption patterns according to their needs at different stages. In times when earnings are higher than consumption, they will save the excess which are then used to support consumption when income declines. If they do consume more than their earnings, they will finance the differences via borrowings. Borrowings typically occur in a person's youthful years where many do not earn enough for consumption. The resources are accumulated during the middle age years whereas during old age, a person would use the assets accumulated earlier. Such pattern would mean that consumption would be even out during their lifetimes (Modigliani & Brumberg, 1954; Ando & Modigliani, 1963).

However, LCH has its weaknesses. It assumes individuals are rational beings with access to all available information to formulate and execute their savings and spending plans. They are able to make complex economic computations and deal effectively with financial markets. Unfortunately, many people do not have such expertise. Acquiring such knowledge also come with time and cost.

The situation is more complex as governments, which previously designed and implement retirement pensions and protection plans, begin to entrust individuals with the responsibility of saving, investing, and decumulation activities. Individuals, who do not need to devote much attention to retirement details before, now have to explore and be aware of their future retirement plans. Not only must they acquire knowledge for this purpose, they must also make assumptions on future earnings, health status and return on investments (Mitchell & Utkus, 2004).

This model does not fully explain people's behaviour. Findings reveal that the elderly do not dissave as quickly as has been said in LCH. They are cautious about unpredictable expenses that may occur in their lifetime, and as such, will save more. These savings are known as precautionary savings. Precautionary saving may be made for the probable event of living longer than expected (and hence having to provide for a longer retirement period) or the possibility of ill-health and huge medical expenses. Others may also save more to

leave bequests to their children (Mitchell & Utkus, 2004). Nevertheless, LCH is useful in making macro predictions such as a country's savings and growth rate.

One popular focus of planning for retirement research is on the demographics of the individual. There are significant differences in the degree of engagement in financial planning based on gender, ethnicity, income, education, occupation, age and marital status (Berger & Denton, 2004; Calasanti, 2010; Ekerdt, 2010; Fontes, 2011; Ekerdt & Baker, 2014). To explain such differences, the Cumulative Inequality Theory is used (Ferraro, Shippee & Schafer, 2009). Cumulative Inequality Theory uses a life course perspective which emphasises that structural opportunities and risks are bestowed by the initial social location of a person's life. It can magnify into advantage and disadvantage, which translates into later life inequalities. Social location is determined by the demographic profile of an individual such as family statuses and origins; gender, ethnicity; the communities where people grow up and grow old; educational, socioeconomic and historical circumstances. Individuals with greater income, wealth, and education face circumstances offering more choices, and fare significantly better in old age than their less educated, less wealthy, lower income counterparts (Ekerdt, 2010; Ekerdt & Baker, 2014). Advantaged structural locations provide these individuals with better arrays of choices and capacities.

However, in current times, traditional social structures and location are continually reinvented and it is unclear how it influences the individual. In addition, life course trajectories are not only shaped by social location. It can change when individual exercise human agency, resulting in intragroup differences. Human agency refers to the capacity of individuals to act independently and to make their own free choices without being grounded in particular circumstances or being sanctioned by their affinity groups (Reeve, Ryan, Deci & Jang, 2007; Hardré & Reeve, 2009). An individual, structurally disadvantaged in earlier life or in one particular life domain, can potentially compensate

this by exercising agency to improve in another life domain. Choices individuals make reflects the variation in individual capacities and provide chances for a good life (Greenwell & Bengtson, 1997; Hellevik & Settersten, 2013).

However, little is known on how individuals can actually make sense of their human agency for their future economic security. Even less is known about how individuals account for the factors that inhibit them from starting financial preparations or prompt them to begin (Kemp, Rosenthal & Denton, 2005). Many of the assumptions in Cumulative Inequality Theory are implicit in nature and inferred from outcomes which has yet been tested empirically (Giele & Elder, 1998; Johnson, Crosnoe & Elder, 2011).

Exploration into planning continues in various areas. For Friedman and Scholnick (1997b), these explorations were initially intended to resolve conflicting explanations on adolescents' behaviour. Claims were made that societal problems associated with this age group stems from their cognitive limitations in the area of planning. However, further investigation reveals that planning itself involves a complex set of psychological operations. An individual has to use its cognitive, motivational and emotional resources in order to achieve the goals set out. How these resources are used reflect an individual's perception of the past, their knowledge of the present and hopes for the future.

Friedman and Scholnick discussed a framework, which outlines the various determinants that influence planning. However, what is interesting is that Friedman and Scholnick (1997b), acknowledges the importance of the 'decision to plan'. They stress that planning is not possible if a person is not psychologically motivated to do so. The decision to plan signifies an individual's intention, that is, their commitment to their planning initiatives. Many cognitive and rational model of planning merely assumes individual would be motivated to ensure that the planning effort is successful. The 'decision to plan' is frequently ignored. However, it is this commitment towards planning that is important.

35

Individuals must believe that planning is essential and that goal developed are important. The goals provide visualisation of the future that is appealing to the cognitive senses, which in turn, motivates them (Friedman & Scholnick, 1997a).

Yeske (2010) voiced similar concerns as decisions made in each step is influenced by a person's psychological and environmental circumstances. He reiterated that studies that emphasis on the "interior dimension" of motivations and values are important. A deeper understanding of beliefs, values, and motivations help to improve activities in financial planning, which aid the 'discovery' process for an individual. Success to any initiatives must match the various challenges in a person's personality, belief system, and personal history. It is these plans and strategies that will have the highest probability of success.

Using this framework, Hershey (2004); Hershey et al. (2007) extended the determinant of planning to financial planning for retirement. Financial planning predictors are grouped into four contributory factors. The first consists of psychological influences, which outline the cognitive, personality and motivational resources required. Cognitive resources are identified as knowledge and perception of task relevance. Personality influences refer to individual preferences such as time, risks and emotional stability. Motivational influences consist of factors such as the clarity of the goals, personal values and a person's confidence.

The second contributory factors consists of characteristics of the planning task. These include aspects such as the complexity of the task and whether there are options available. Task components such as activities that are long range in nature and daily monitoring of one's portfolio also influence a person's decisions.

The third group of factors refer to the environmental influences to planning such as availability of financial resources and economic forces. Financial resources come from an individual income base, current savings, assets and discretionary income. It also includes sources such as informational, educational and technological resources which support planning behaviours. Economic forces reflect the conditions and future income expectations that influence financial planning for retirement.

The final group of factors are the sociocultural influences or the cultural ethos. It represents a collection of social forces from society, family and peer norms that influence the individual. It shapes the individual psychological tendencies and predisposition.

Studies on the various determinants engendered useful findings. More importantly, it provided empirical support to the various determinants of financial planning for retirement (Hershey et al., 2007; Hershey & Jacobs-Lawson, 2009; Van Dalen, Henkens & Hershey, 2010; Hershey et al., 2013; Gutierrez & Hershey, 2014; Hershey & Henkens, 2014; Gerrans & Hershey, 2016; Gupta & Hershey, 2016; Kiso & Hershey, 2017). However, the decision or intention to plan and individual motivations for financial planning for retirement has yet been fully explored. Intentions indicates how hard people are willing to try, and the effort that they put in to perform a behaviour. The stronger the intentions, the more likely a person will perform the behaviour.

Thus, to understand intentions, the Theory of Planned Behaviour (TPB) is explored. TPB is a psychological and motivational theory which offers a theoretical approach to predict and explain human behaviour by using selected determinants (attitudes, subjective norms, perceived behavioural control) (Fishbein & Ajzen, 1975; Ajzen & Fishbein, 1977; Ajzen & Fishbein, 1980). It assumes that human behaviour follows a reasonable and often spontaneously path from the information or beliefs people possess about the behaviour. These beliefs come from a variety of sources, such as personal experiences, formal sources of information, interactions with family and friends. People may also form beliefs from the ways they interpret and remember information. These beliefs then guide the decision to perform or not perform a behaviour. Although this theory assumes that people behave in similar ways and are influence by a limited set of constructs, it also recognises the importance of domain-specific constructs in understanding behaviour. Its integrative framework has stimulated a great deal of empirical research and has been used to explain a wide range of behaviours such as health behaviour (Conner & Sparks, 2005), leisure choices (Ajzen & Driver, 1992), weight loss (Sparks, Shepherd, Wieringa & Zimmermans, 1995), fertility intentions (Ajzen & Klobas, 2013) and information technology usage (Taylor & Todd, 1995). Also, TPB's theoretical validity and robustness were further substantiated from the meta-analysis studies conducted (Ajzen, 1991; Godin & Kok, 1996; Armitage & Conner, 2001).

However, there is limited research on its use in financial and retirement behaviour studies. Nevertheless, existing studies using TPB in this area such as credit counselling (Xiao & Wu, 2006), tax compliance (Bobek & Hatfield, 2003), debt management (Xiao & Wu, 2008), savings behaviour (Croy et al., 2010a; Croy et al., 2010b; Croy et al., 2012) and retirement planning (Cohen, 2014; Heraty & McCarthy, 2015) had shown promising results. To determine the suitability of this framework, literature predicting and explaining financial and retirement behaviours are reviewed in greater detail.

2.2 Attitudes and Beliefs

In investigating the various determinants in financial behaviour studies, attitudes have been a popular focus of research. Attitudes are described as a person's predisposition to an object of study (Fishbein & Ajzen, 1975; Ajzen, 2005b). This predisposition influences a person's response and behaviour. The object of study may encompass people, events, activities or ideas. Allport (1935, p. 810) further reiterated this in the following definition, that attitudes are "a mental and neural state of readiness, organised through experience, exerting a directive or dynamic influence upon the individual's response to all objects and situations with which it is related". Similarly, Eagly and Chaiken (1993, p. 1) have described it as "a psychological tendency that is expressed by evaluating a particular entity with some degree of favour or disfavour." This is further supported by Crano and Prislin (2006, p. 347); where attitudes are perceived as "...evaluative judgements that integrate and summarise... cognitive or affective reactions".

From the definitions, the study of attitudes offers a multitude of possibilities for research. It is a predominant reality that individuals differed in their attitudes pertaining money and finances. Not only that, they also differ in how they plan and the perspective they have on retirement issues. Some are vigilant, well organised and maintains control of their finances. However, there are also others that do not keep account of the money spent or have plans to deal with future or unexpected expenditures. In a research by Atkinson and Messy (2012) on financial literacy, different financial attitudes towards money and planning for the future were documented. Fourteen countries participated in this study. Although most countries reported favourable financial attitudes, results from Armenia and Poland were different. Only 1 in 10 Armenian have positive attitudes. For Poland, only 27% are reported to have positive attitudes. The study also documented that only 53% of Malaysians have positive attitudes, which is worrisome.

Various studies had been conducted to explain these differences in attitudes. One such attempt involves understanding about money attitudes. In the studies, researchers have discover a variety of psychological meanings of money. Subsequently, scales have been developed to measure these meanings. These meaning ranges from positive money attitudes (such as 'comfortable security') to negative ones (such as 'shameful failure') (Wernimont & Fitzpatrick, 1972). Further studies in this area also uncover other meanings. For example, money has been used to measure attitudes in relations to status, respect and freedom (Goldberg & Lewis, 1978); perceptions (Yamauchi & Templer, 1982), behaviour

and beliefs (Furnham, 1984) and ethics (Tang, 1992). Money, by itself may be seen as an inert object, but due to differences in one's experiences and situations, it has been empowered with special powers and meanings. Unfortunately, research into money attitudes was limited to aspect pertaining to the demographic profile of a person and variables such as spending habits of the individual (Furnham, 1984). The motivational influence of money on healthy financial behaviours such as financial planning for retirement is not fully explored.

Attitudes towards planning were also studied by various researchers. Van Deventer et al. (2014) investigated attitudes on personal financial planning of South African students. Findings reveal that the respondents have positive attitudes towards personal financial planning and placed importance on specific areas such as credit, insurance, investment and estate planning. This is in contrast to Tan, Hoe and Hung (2011) where Malaysia respondents displayed neutral perceptions on critical components of personal financial planning. What is particularly disturbing is the area of retirement management. Although respondents were aware of the amount of money they would need for retirement, they have yet started with their retirement planning nor do they have any formal retirement plan.

Similarly, attitudes toward various financial activities were also analysed. One of these involved attitudes towards savings. Savings is a wealth accumulation activity where a person's manages his assets with the intentions creating wealth. Croy et al. (2010b), in studying the motivations of retirement savings among fund members in Australia, discovered that attitudes have a positive influence on intentions. Other variables such as perceptions of planning and planning preparedness also have powerful influences. Similarly, Joo and Grable (2005) discovered that positive financial behaviours and attitudes increase the likelihood of a person having a savings programme.

Investment is another area of interest among researchers. Investments are activities where one can grow or increase wealth. Sivaramakrishnan, Srivastava and Rastogi (2017), in studying the various determinants of investments decisions, discovered that investors attitudes have a powerful influence on intentions to invest in equity markets. Other influential aspects include regulatory perceptions, avoidance of risk and various hassle factors (such as procedural complexity, expenses, and time) are important attitudinal considerations when investing. Research into understanding consumers investment decisions also involved an understanding of other influences such as inflation (Armantier, de Bruin, Topa, van der Klaauw & Zafar, 2015). Inflation measures the rate of price increases over time and is a significant risk associated with the effort to grow investment. Most investors would judge the success of their investment decisions by comparing the rate of return of any investment portfolio with the inflation rate. The result supports the economic theory and reaffirms the relevance of inflation expectations.

Notwithstanding, research also dwell into various risks attitudes. In a study by Yilmazer and Lich (2015), risk preferences are discovered to be an important influence on household investment decisions. In addition, the risk tolerance of spouses also influences the assets allocation, where the spouse with the higher bargaining power is more likely to influence the proportion of risky assets in the investment portfolio.

Risk attitudes are also studied in the purchase of insurances. In Furnham and Goletto-Tankel (2002), young people showed a poor understanding of life assurances. Topics on life assurances were deemed boring, as it appears irrelevant to their lives. However, those with an understanding of life assurance have a firm grasp on the importance of savings and arranging for their financial future. Owusu-Frimpong, Omar and Mmieh (2011) in their study on attitudes towards financial products in Ghana revealed a low level of subscription to life insurance policies. The researchers attribute this to the reliance on help from family and relatives in time of financial distress. Also, many could not afford to purchase insurance given their low-income level.

In summary, there is a myriad of attitudes that influence financial, money and retirement behaviours. Such attitudes may stem from their beliefs about money, planning, the range of financial products or even on retirement goals. Due to the evaluative nature of attitudes, individuals could take a position with respect to any of these aspects on a dimension that ranges from negative to positive. For example, planning for retirement is perceived differently by different people. For those who have successfully accumulated adequate resources to fund this period, they evaluate the whole activity positively, and may even look forward to retiring. Others, who have inadequate resources, may see it negatively (Kim & Moen, 2001). Hence, differences in how one evaluate the different activities in financial planning for retirement (finances, money, planning and retirement) have different effect on intentions (Lusardi, Schneider & Tufano, 2011). How one evaluates these different aspects have not been explored well, and current findings are fragmented. There is a need to further investigate the overall attitudes on financial planning for retirement, and its effect on different populations and context (Ajzen, 1991).

2.3 Contextual Environment

The social environment influences planning behaviours. This influence is captured in TPB in the construct subjective norms, and refers to what is acceptable or permissible behaviour in a group or society (Fishbein & Ajzen, 2010). Social norms influence performance of a particular behaviour; and in TPB, it is viewed as perceived social pressure to perform (or not to perform) a given behaviour.

Studies done in this area has identify two main sources of social pressures. The first are proximal influences from the immediate surroundings of the individual. These influences may be from the family unit, friends or colleagues. The second source represent distal influences, which are not immediately present to the individual. These influences may come from cultural values, customs and laws or various professionals and social groups to which an individual belongs to.

A common source of social pressure to the individual comes from family members. Friedman and Scholnick (1997a) discuss the significant effects it have on the individual. Families provide the knowledge base for planning in their daily activities. For example, family routines are forms of daily anticipated plans, which build expertise, that is observed and internalised by family members.

As such, there is a strong and significant influence of family members on intentions to plan, particularly for retirement (Binswanger & Carman, 2012; Wang, Worsley, Cunningham & Hunter, 2014). Wang and Shi (2014) discussed how family and social network factors influence allocations of retirement investments in defined contributions plans. The findings indicate that the risk attitude of the spouse influences retirement investments allocations. Yang and Devaney (2012) also reported that spouses influence retirement wealth. They coordinate their pension and investments decisions and would rather opt for similar rather than diversified investments (Wang, Henkens & Van Solinge, 2011). Hence, the amount of retirement assets a person holds is also influenced by the bargaining power of spouses.

Notwithstanding that, many spouses also coordinate their leisure and social activities together. It is common for one to accelerate a retirement plan in order to ensure that they retire together with their spouses (Street & Desai, 2011). Couples may also stall their retirement plans depending on their dependent-care situations. Individuals with dependent

children are reported to postpone retirement plans. Similarly, others may retire early to take care of dependents with illness or grandchildren (Wang & Zhan, 2012; Wang & Shi, 2014). Some retirees also engage into some form of bridge employment if their resources are inadequate. But these are frequently seen as forced rather than wanted (Kim & Moen, 2001).

Children faced with similar planning situations would emulate their parents (Koposko & Hershey, 2014; Koposko & Hershey, 2016; Koposko et al., 2016). Kimiyaghalam, Mansori, Safari and Yap (2017) discovered that parents influence have a significant effect on retirement planning. Tang, Baker and Peter (2015) also reported that parents influence responsible financial behaviour. The thoughts and feelings of parents are important as it serves as a reference point on how to handle the issues at hand. Parents provide support structures and some even share responsibilities with the individual.

In addition to the family, peer groups such as colleagues and friends also play a role in influencing financial planning for retirement. Bayer, Bernheim and Scholz (2009) reported that individuals seem more likely to plan after being exposed to financial preparation seminars with colleagues in their workplaces. Individuals also have more favourable expectations towards retirement (Taylor-Carter, Cook & Weinberg, 1997). Many are more willing to engage in financial preparation with the basic financial planning skills and principles acquired during the seminars (Hershey, Walsh, Brougham, Carter & Farrell, 1998; Hershey, Mowen & Jacobs-Lawson, 2003). Croy et al. (2012) reported that peer effects might influence saving decisions and the formulation of saving plans. In the United States, many new employees are reported to be influenced by the decisions of peers or seniors when it comes to decisions on their retirement such as participation and contribution levels. Friends also play a role in influencing various aspects of retirement.

They provide informational support, be role models and sometimes even offering monetary assistance (Wang & Shultz, 2010; Zheng, Wang & Gao, 2012; Zhan, Wang & Yao, 2013).

Similarly, a person may be influenced by factors not immediately present in their situation. These distal influences may come from cultural values, customs, and laws to which an individual belongs. This would include various professionals and social groups.

Distal influences have also affected financial and retirement behaviour. Joo and Grable (2005); Grable et al. (2011); Marsden, Zick and Mayer (2011) discussed the roles of professional advisers (such as financial planners, lawyers, accountants and counsellors) in dispensing advice to their clients. Clients have reported to be more confident in making a financial decision after discussions with their advisors. They are more informed and gain a clearer perspective of how they can achieve their objectives. These professionals are also in a position to assess their client's current situation and provide suggestions on how best to plan.

There also appear to be a correlation between greater wealth and the financial advice dispensed by these professionals, but the direction of causality is difficult to ascertain (Kramer, 2012). Croy et al. (2012) reported that although the belief towards the financial advisers may be strong, many are not motivated to comply due to distrust. Financial professionals also have their interest, and their efforts may not always improve the financial situation of their clients.

Influences from governments via their policies and social protection initiatives play an important role in influencing planning and retirement behaviours. However, the extent to which the government plays a role is also dependent on its political structures. For example, some countries emphasised on their social protection measures. Known as welfare states, these countries may adopt a universal system, with provisions that cover all citizens. Such

protection helps an individual in its retirement planning. Croy et al. (2012) revealed that although belief towards the government is strong, motivation to comply towards its initiatives is questionable.

In summary, the normative influences in a person's life are vast and diverse. These influences may be proximal or distal in nature. While it cannot be denied that people are more receptive and trusting towards individuals close to them (family members, friends, peer groups), distal influences (government, professional advisors) also have effects on intentions. Hence, identifying sources of normative influences is important to explain the social pressures influencing the environment of an individual.

Besides that, people are not only influence by what the important referents expects, but also in what they do. Studies of such nature (Koposko & Hershey, 2014; Koposko & Hershey, 2016; Koposko et al., 2016) stressed the importance of both the injunctive and descriptive component in subjective norms. Yet, who and the extent to which these influences play a role in a person's financial decisions differs. Early interpretations of subjective norms were questionable as they only included injunctive components. These findings also frequently assumed subjective norms components are the weakest, resulting in numerous debates on its role in TPB (Fishbein & Ajzen, 1975; Godin & Kok, 1996). Although later studies by Fishbein and Ajzen (2010); Ajzen and Klobas (2013); Klobas and Ajzen (2015) discover that different antecedents of intentions are expected to have different effects, empirical evidences are required to integrate the study of descriptive norms into TPB.

2.4 Perceptions of control

The previous sections have discussed the various literature related to attitudes and subjective norms. However, intentions are also influence by factors beyond a person's control. People cannot act on their intentions if they lack the skills or resources or if external factors prevent them from doing so. Hence, for a person to successfully carry out their intentions, they must have sufficient volitional control. Unfortunately, measures of actual control are not always possible, as it is not always known to the individual all the relevant internal and external factors that may facilitate or impede performance of a behaviour. Nevertheless, a person's perceptions of their control over a behaviour can accurately serve as a proxy for actual control and is frequently used in the prediction of behaviours (Ajzen, 1991; Fishbein & Ajzen, 2010; Klobas & Ajzen, 2015; Ajzen, 2019).

2.4.1 Cognitive Requirements

Cognition is an important aspect of planning, especially for complex matters. Cognition is mental processes in the acquisition, storage and use of knowledge. They are thinking skills, which influence the quality of decisions and perceptions. With regards to planning for finances and retirement, knowledge in the area of personal finance, investing, risk management and savings are of particular importance (Hershey et al., 1998; Hershey & Walsh, 2000; Hershey, Jacobs-Lawson & Walsh, 2003). Hershey (2004) further elaborated on two particular dominant forms of knowledge in financial planning for retirement. The first is the domain-specific declarative knowledge which includes knowledge in critical financial and investing concepts or dynamically changing information such as prime lending rate or major stock market indices. Next is procedural knowledge, which includes knowledge of specific procedures. This may include aspects such as the important steps in the retirement planning process or procedural requirements to check online for a person's retirement savings status. Both types of knowledge are good predictors of involvement in the financial and retirement planning process.

One particular area of focus in relations cognition is financial literacy. Financial literacy is the knowledge and ability to use to manage financial resources (Beal & Delpachitra, 2003; Hung, Parker & Yoong, 2009). Skills to manage financial resources is important, as it ensures that a person's well-being is taken care of in their retirement days (President's Advisory Council on Financial Literacy, 2008; Jump\$tart Coalition for Personal Financial Literacy, 2018). Mandell (2007) also stressed that these financial skills are essential for making more informed decisions for the future.

The importance of financial literacy is recognised by policymakers. It has been embraced as a solution, especially in light of the complexity of financial products in recent times and the increasing responsibility of saving, investing and planning for retirement entrusted to workers and retirees (Fernandes, Lynch & Netemeyer, 2014).

Financial literacy has an influence on financial behaviour. The empirical studies by Joo and Grable (2005); Hershey et al. (2007) reveal that financial knowledge help increases savings level. Those with financial knowledge have a higher perception of the adequacy of their savings. Similar results are also sighted in research in the Netherlands, Hong Kong (Chou, Yu, Chan, Wu, Zhu & Lou, 2015) and in Mexico (Koposko, Bojórquez, Pérez & Hershey, 2016). Financial literacy also has strong links with investment, cash and credit management (Hilgert, Hogarth & Beverly, 2003) and engagement in healthy financial practices (Martin, 2007; Bayer et al., 2009; Collins & O'Rourke, 2009). Similarly, Courchane and Zorn (2005); Courchane, Gailey and Zorn (2008) in examining links between financial knowledge, financial behaviours and credit outcomes, discovered knowledge to be a key explanatory variable for behaviour and credit outcomes. Subsequent research has also positively correlated financial literacy to retirement planning and wealth

accumulation activities (Ameriks, Caplin & Leahy, 2003; Van Rooij, Lusardi & Alessie, 2011; Van Rooij & Teppa, 2014; Lusardi & Mitchell, 2017).

Many investment decisions and activities are dependent on financial literacy. Van Rooij, Lusardi and Alessie (2012) reported that financial literacy has influences on stock market participation. The investments choices and the evaluation of investment fees are also influenced by how financially literate a person is (Christelis, Jappelli & Padula, 2010; Van Rooij et al., 2011; Lusardi & Mitchell, 2011b; Choi, Kariv, Muller & Silverman, 2014). In contrast, negative financial behaviour such as debt accumulation, poor loan choices and inability to service monthly debt instalments are related to low financial literacy (Lusardi et al., 2011).

The importance of financial literacy and its effects on the financial well-being of a person is undeniable. Unfortunately, many people are still financially illiterate. Results from the 2004 Health and Retirement Study in the United States indicates low scores in several measures of financial literacy. Only about half knew how to compute compound interest and inflation rate. 34.3% could answer all financial literacy questions correctly, 35.8% has only two responses correct, 16.3% has one response correct, and 9.9% got all answers wrong (Lusardi & Mitchell, 2011a). Such responses were quite revealing as these groups are of the age where there would have to make important financial decisions that would influence their retirement future. They have also witnessed numerous financial and economic challenges which have provided them with the exposure and information on financial risk.

This survey is replicated to include different groups such as young respondents between ages 23 to 28 (2007–2008 National Longitudinal Survey of Youth) (Lusardi, Mitchell & Curto, 2010) and respondents of all ages (RAND American Life Panel) (Lusardi & Mitchell, 2010). In 2009 and 2012, the National Financial Capability Study was also conducted (Lusardi & Mitchell, 2011c). All the findings concluded that the level of financial literacy in the United States is low.

Similar results were also discovered in other countries. This was regardless of the stage of economic development or the political state of the country (Millar & Devonish, 2009; Chinen & Endo, 2012). For example, countries with developed financial markets such as Germany, Netherlands, Australia and Japan have reported low literacy rates. Similarly, the Organisation for Economic Co-operation and Development (OECD) have highlighted the lack of financial literacy among its member countries (Christelis et al., 2010). Nations with rapidly changing financial markets such as Russia and Romania also reported low levels of financial literacy rates (Lusardi & Mitchell, 2014). Atkinson and Messy (2011); Atkinson and Messy (2012) discovered patterns of financial illiteracy in 14 countries stretching across four continents. These countries are at different stages of development, and the study was conducted using a similar set of financial literacy questions.

The lack of financial literacy is of grave concern to many parties, especially policymakers. It indicates that many have problems in financial matters. They would not be able to avoid financial mistakes or may not be aware of financial practices that can help them cope with possible economic shocks.

Lusardi and Mitchell (2007); Lusardi and Mitchell (2011c); Lusardi and Mitchell (2014) highlighted that financial matters are cognitively challenging and time-consuming. The collection, analysis and evaluation of different information, facts and alternatives is an expensive process. To many, planning is only useful when the expected gains from the

effort made exceed its costs. However, this is not always the case, which is why some opt not to plan, as they think it is not worth the time and effort (Agnew, Szykman, Utkus & Young, 2012).

Correspondingly, Remund (2010) and Fernandes et al. (2014) revealed that many other factors also influence financial literacy besides knowledge, skills and ability. The metaanalysis studies of Fernandes et al. (2014) found that interventions to improve financial literacy could only explain about 0.1% in the variance of financial behaviours. The results revealed that financial literacy effects deteriorate with time. Large scale intervention efforts also have similar effects. This is especially true as many things taught are not immediately acted upon.

Likewise, many researchers highlighted that there is inconclusive evidence to demonstrate that a causal relationship exists between financial literacy and improved financial outcomes (Willis, 2008; Willis, 2011; Hastings, Madrian & Skimmyhorn, 2013; Fernandes et al., 2014; Holzmann, 2015). Similarly, although Hilgert et al. (2003) have evidence of a link between financial education and sound financial decisions, they are also of the opinion that the findings are inconclusive. Other factors such as biases, heuristics and nonrational determinant could also influence decisions. The intrinsic personal characteristics of individuals, to want to improve their financial situation, have also been known to cause people to seek financial education.

Hadar, Sood and Fox (2013) argued that knowledge in financial matters could be in an objective or subjective form. Objective knowledge refers to accurate information that is stored in memory. Subjective knowledge refers to a person's assessment of their knowledge. In a series of experiments conducted, respondents who felt more knowledgeable were more confident in their decisions and take proactive financial action. The results demonstrated that financial information influences a person's decisions via

subjective knowledge. They concluded that most financial literacy programmes fail because of the focus on objective knowledge instead of subjective knowledge. Subjective knowledge is important, as they are associated with consumer confidence, which drives behaviour.

Meanwhile, in a research on the dimensions of financial planning for retirement, Huston (2010) revealed that a financially literate person would not exhibit the predicted positive financial behaviours if they lack the confidence to use the knowledge acquired. This additional application dimension stress that information is not the only important dimension. Ability and confidence to apply the information appropriately should not be ignored.

Similar studies have documented the importance of confidence in financial planning for retirement (Organization for Economic Co-Operation and Development, 2005; Lusardi & Mitchell, 2007; Hung et al., 2009). People, especially those who do not have the financial knowledge, depend heavily on how much they think they know (Lusardi & Mitchell, 2007; Hung et al., 2009). Confidence is reported to be more predictive of financial behaviours. Sometimes, the influence of confidence is even greater than actual knowledge (Hung et al., 2009).

Willis (2008); Willis (2011) argued that effective financial literacy would be difficult to achieve. Due to the dire straits of financial literacy in many countries, effort in educating people has to be extensive. Also, the changing financial landscape would mean that the ordinary persons are left to their own devices when navigating through the ever-changing cornucopia of financial products. An individual is only armed with financial education as he or she weaves through this sea of challenges.

The dynamic financial marketplace also makes it difficult for many to keep up with the changes. Effective implementation is hard given the constant change. Furthermore, there is no right answer for financial matters. What is right for one person, may not be the solution for another. As such, massive education campaigns on financial literacy may not yield the expected benefits. Instead, education that is tailor-made to a specific audience may be more beneficial.

In summary, whilst cognitive abilities in economic and financial matters are essential in financial decision making and management, it is not without its challenges. This is evidenced as most countries reported facing financial illiteracy. In addition, findings on the influence of financial literacy on improving financial behaviours are mixed (Fernandes et al., 2014). Various other factors such as confidence and the personal characteristics of individuals are equally influential. In addition, the massive effort and costs to educating populations do not necessarily yield expected results.

2.4.2 Time elements

One important aspect of planning is that it involves projections into time. Indirectly, it represents a person's intent to control their future. However, such attempts are challenging as the future is unknown and hard to predict. To draw up a plan of action of a future life demands creativity, innovation and foresight. As such, many streams of research have investigated the challenges and obstacles presented by the time elements in planning.

One such research investigates self-control or self-regulation failure. Self-control refers to a broad category of conscious and nonconscious effort a person exerts to control behaviour. A volitional self-regulated act involves restraint over an impulse. To counter the impulse, a person has to alter its states and responses. These skills involve patience, impulse control and willpower as they grapple over the constant struggle to control their urges. Self-control failure happens when a person fails to exert the desired action. Such lack of self-control ability happens when individuals delay or procrastinate a positive behaviours and indulge in negative impulses (Pham, Lee & Stephen, 2012; Ryack, 2012; Van Schie, Dellaert & Donkers, 2015)

In planning financially for retirement, a person is faced with the decision to delay gratification. They have to override an immediate impulse of enjoying a tempting stimulus now and in return for a delayed but better reward (O'Donoghue & Rabin, 1999; O'Donoghue & Rabin, 2001). Plans and strategies reflect the future which may or may not happen whereas temptations (to spend) are an immediate reality. The timing of rewards and punishments are temporally far apart, making efforts at self-control a constant struggle (Gollwitzer & Oettingen, 2011; Koch & Nafziger, 2011; Townsend & Liu, 2012; Dzhogleva & Lamberton, 2014). These are the reasons why people with high credit card debt and low pre-retirement wealth (Ainslie, 2010; Charlton, Gossett & Charlton, 2011; Choi & Kim, 2014).

The tendency for people to have a stronger preference for more immediate payoffs relative to later payoffs is described as temporal myopia in the field of economics. Such studies reveal the myopic tendency of human to give in to impulses. To explain such tendencies, people are assumed to discount more on rewards that are further into the future in comparison to shorter delays, which reflects the inconsistent choices made over time (Ainslie, 2005; Farmer & Geanakoplos, 2009; Ainslie, 2010; Charlton et al., 2011; Teuscher & Mitchell, 2011; Choi & Kim, 2014). This results in people regretting when they make short term choices, and discovering later that waiting would result in a more substantial reward. Such short term choices have been thought of as impulsive or rash and it reflects the inability of people to consider the long-term outcomes of an action when making a choice (Ainslie, 2010; Charlton et al., 2011; Choi & Kim, 2011; Choi & Kim, 2014).

The preference for immediate pleasure instead of a future reward is not new. Such preferences result in them discounting the future reward, making it less attractive. Ainslie (2010) also describe it as the most basic impulse. However, what causes these impulses is debatable. Sometimes it is attributed to naiveté, or inability to estimate the values of the contingencies involved. It also shows that people tend to act impulsively when given choices involving time.

2.4.3 Resources

With increasing age, individuals would generally withdraw from the labour force, either voluntarily (through retirement) or involuntarily (through the loss of a job or for health or caregiving reasons). Without an active link to the labour market, this means a decrease in income for most. For a substantial minority of older individuals, social welfare programs implemented by governments are the primary source of income and they may find themselves living in low-income situations. Hence, in many situations, those that are older would prepare to avoid such circumstances. They may also plan more because finances become more salient as they approach retirement or because owing to life cycle factors (higher income, decreased childrearing expenses), they have the available resources (Kim & Moen, 2001). Such efforts would free a person psychologically from the stress of old age, enabling one to have the freedom and autonomy to make life choices (Harrison, 2005; Grable et al., 2011; Noone, O'Loughlin & Kendig, 2012).

Hence, resources (especially those that are financial in nature) is of great interest (Ekerdt et al., 1996; Ekerdt, 2010; Ekerdt & Baker, 2014). These variables correlate well with retirement savings and financial planning activities. High income individuals are reported to be more actively engaged in planning as they have more resources (Hira, Rock & Loibl, 2009; Fernández-López, Otero, Vivel & Rodeiro, 2010). They are also more
knowledgeable, have clearer financial goals and may engage the services of a professional financial planner. As such, they have a higher perception of the adequacy of their retirement effort (Balasuriya, Gough & Vasileva, 2014; Chou et al., 2015).

However, resources of a person are not limited to its income generating capacity. Different resources and their availability can be identified along demographic and psychological lines. Some of these differences (in equalities or inequalities) could be due to an individual's initial social location in life, which magnifies into social and economic opportunities and risks in old age (Berger & Denton, 2004; Ferraro et al., 2009; Calasanti, 2010; Ekerdt, 2010; Fontes, 2011; Ekerdt & Baker, 2014). For example, people with greater income, wealth and education face circumstances offering more choices and fare significantly better in old age than their less educated, less wealthy, lower income counterparts (Ekerdt, 2010; Ekerdt & Baker, 2014).

In analysing these differences, age is frequently the demographic variable used. Findings have reveal that older individuals are more engaged in planning in comparison to their younger counterparts (Hershey et al., 2007; Hershey et al., 2007; Padawer et al., 2007). Phua and McNally (2008) also reported age differences in their study of men's attitudes towards retirement planning where older men are more likely to save for retirement. Older workers are normally paid more (and hence have more resources to plan) or hold higher positions in their respective organisations. Being older; most would have also settled their debts and have the time to be engaged in planning activities (Ekerdt, Hackney, Kosloski & DeViney, 2001). They may also be more knowledgeable in financial matters and less tolerant to risks to their investments (Dulebohn, 2002; Dulebohn & Murray, 2007). In addition, retirement objectives and realities are clearer as one ages, which indirectly motivates a person to plan.

Gender also has its influences, with men reported to be more likely to engage in financial planning and savings activities (Glass & Kilpatrick, 1998; Moen, Huang, Plassmann & Dentinger, 2006; Noone, Alpass & Stephens, 2010; Noone et al., 2012). They are also more proactive in growing wealth via investments (Glass & Kilpatrick, 1998) and have more specific retirement goals in comparison with women (Hershey, Jacobs-Lawson & Neukam, 2002). Various reasons are cited for such differences, namely the fact that men earned more as they hold permanent jobs and have access to a more illustrious career. Traditional sexrole patterns have also dictated that men are responsible for the family financial planning activities.

Relative to men, women are at greater risk of poverty in later life, especially widowed or divorced women (Lown, 2008; Rowley, Lown & Piercy, 2012). Women are expected to forsake their careers to take care of the family and with that many have discontinuous work patterns. Women also have a longer lifespan relative to their male counterparts which makes them more susceptible to the risk of aging (such as disease and disability). For some, particularly those without access to social and material resources, this may mean a loss of independence and a risk of isolation and loneliness.

Education level also influences retirement planning. Highly educated people have more capacity and resources to plan due to their knowledge and exposure (DeVaney & Chiremba, 2005). They also have a better understanding of financial matters and have the cognitive resources to make an informed decision (Harris, Loundes & Webster, 2002; Lai & Tan, 2009; Folk, Beh & Baranovich, 2012; Hershey & Jacobs-Lawson, 2012).

However, the use of demographic indicators has been critique. The precise reasons why certain demographic variables appear to have strong links to financial preparation are not always clear or direct. Demographic factors such as culture and religion has not been fully investigated. At the same time, there appear to be intragroup differences. Many of the assumptions used are also implicit in nature and inferred from outcomes (Giele & Elder, 1998; Johnson et al., 2011). Hershey et al. (2013) states that these indicators are not explanatory variables and have limited use in theory development.

Hence, there are other plausible reasons for such differences. Foremost, is the belief that a person has the ability to manoeuvre across the various constraints of inequalities in life by exercising human agency. Human agency refers to the capacity of individuals to act independently and to make their own free choices (Reeve et al., 2007; Hardré & Reeve, 2009). A person structurally disadvantaged in earlier life or in one particular life domain can potentially compensate this by exercising agency in another. Hence, an individual may have low social economic status, but may experience well-being and happiness through planning behaviours that capitalize on advantages in other domains (such as a nurturing family relationship or a large network of friends for social support).

When planning, multiple psychological mechanisms that build up the motivational resources of a person is at work. Firstly, in planning, an abstract goal is transformed into specific sets of action (Townsend & Liu, 2012). This builds clarity which enables a person to set guidelines and contingencies to make better decisions (Thaler & Benartzi, 2004; Benartzi & Thaler, 2007). Risk of uncertainties are reduced, and individuals can cope better with arising challenges in life. Atkinson, McKay, Collard and Kempson (2007) discussed that by planning, a person would be able to cope with various unexpected emergencies. Certain financial planning activities like budgeting help individual to monitor and control, enabling them to live within their means. It helps identify expenditure that can cause real problems (where one has to borrow in order to meet the commitment) or deal with commitments that fell outside the typical pattern of expenditures.

When planning, a person also considers the context one is in. Such consideration enables a person to evaluate things in different perspectives. For example, one may visualise retirement as a positive or negative phase in their lives. Those who sees it as an opportunity for a 'new start' are motivated to make early preparations (Wang, 2007; Wang et al., 2011; Wang & Shi, 2014) while others who view it as an unwelcome, imposed disruption may dread the uncertainty and procrastinate in planning (Frieze, Olson & Murrell, 2011; Karpinska, Henkens & Schippers, 2011; Van Solinge, 2014).

Hence, the various planning initiatives help a person to be more organised and solve problems (Hayes-Roth & Hayes-Roth, 1979; Friedman & Scholnick, 1997a). This builds commitment towards the 'decision to plan' (Stawski, Hershey & Jacobs-Lawson, 2007). Research suggests that those who plan and have clear financial goals are more engaged in financial preparations (Neukam & Hershey, 2003; Stawski et al., 2007; Hershey, Austin & Gutierrez, 2015).

In summary, resources are an important consideration in financial planning for retirement. Wang (2007); Wang and Shultz (2010); Wang and Shi (2014) postulate that a person's adjustment to retirement depends on their resources. However, these resources are not limited to financial resources. They can extent along demographic and psychological lines. In most demographic analysis, by virtue of their social location in life, older, higher educated males appear to be more prepared financially for retirement. Yet, such assumptions are challenged especially when there are also intragroup differences amongst the groups. Individuals have the freedom to mapped out strategies to overcome life challenges and build resources. They need not fall in line with traditional social structures. For example, a person may compensate an inequality in one domain of their lives by planning to capitalize the advantages in other domains. Such efforts come in many forms,

which makes it a rich landscape for research (Wang & Shultz, 2010; Wang et al., 2011; Wang & Shi, 2014).

2.5 Theoretical Framework

This study adopts a reasoned action approach to understanding financial retirement planning. TPB postulate that behaviour will always follow a reasonable and spontaneous path from the information or beliefs people possess. Hence, it can be predicted and explained by using a limited set of determinants (attitudes, subjective norms, perceived behavioural control) (Fishbein & Ajzen, 1975; Ajzen & Fishbein, 1977; Ajzen & Fishbein, 1980).

Historically, the early formulation of this approach recognised intentions as the immediate antecedents of behaviour and that intentions are, in turn, a function of attitude and subjective norms (Fishbein, 1963; Ajzen & Fishbein, 1969; Fishbein & Ajzen, 1975). This early formulation was referred to as Theory of Reasoned Action (TRA).

Early studies using TRA have demonstrated its applicability in various situations and behaviours (Beck & Ajzen, 1991). Meta-analysis review by Sheppard, Hartwick and Warshaw (1988) also revealed that TRA was able to predict across a range of behaviours. However, TRA is only applicable to behaviours must be under the complete volitional control of the individual. This would mean that a person must have consciously thought of all related aspects of the behaviour and can decide whether to perform or not perform the behaviour. Hence, not all behaviours can be explained by TRA (Ajzen & Fishbein, 1980; Ajzen, 1985).

This had led to the introduction of the construct Perceived Behavioural Control (PBC), an additional predictor of both intention and behaviour. PBC is defined as the individual's perception of ease or difficulty in performing a behaviour. It assumes the performance of a behaviour is contingent on the presence of opportunities or resources (Ajzen, 1991; Ajzen, 2002b; Fishbein & Ajzen, 2010). With this extension, TRA is known as the Theory of Planned Behaviour (TPB). TPB cater to the non-volitional aspects of behaviour (Ajzen, 1985; Sheppard et al., 1988; Ajzen, 1991). In other words, for behaviour not purely under volitional control, perceptions of control would influence intentions and behaviour (Ajzen & Fishbein, 1980; Ajzen, 2005a).

TPB was tested across a wide range of behaviours. Findings reveal that it was able to explain well the variances in intentions and behaviour. The average correlation for intentions ranges from 0.59 to 0.66 (Notani, 1998; Armitage & Conner, 2001; Schulze & Wittmann, 2003; Rivis & Sheeran, 2004; Rivis et al., 2009). Armitage and Conner (2001) also found that the influence of PBC is significant even after controlling the effects of attitudes and subjective norms. Several meta-analyses also reveal positive findings where PBC has added power to predict behavioural intention and behaviour (Conner & Armitage, 1998; Armitage & Conner, 1999; Armitage & Conner, 2000; Armitage & Conner, 2001).

Research was also undertaken to compare the prediction of intentions and behaviour using TRA and TPB (Madden, Ellen & Ajzen, 1992). The results demonstrated that TRA was adequate in explaining behaviour within volitional control (Ajzen, 1991). For behaviours not under full volitional control, TPB was superior in explaining the variance in intentions and behaviour. This is also supported by meta-analysis reviews in different fields (Armitage & Conner, 2001). Perceptions of control have demonstrated an influential effect on intentions and behaviours. However, in later studies (Fishbein & Ajzen, 2010; Ajzen & Sheikh, 2013; Klobas & Ajzen, 2015) the relative importance of the constructs (attitudes, subjective norms and PBC) are found to vary. Such findings are expected, as intentions and behaviours differ among individuals and populations. Some populations may be affected by attitudes and less so by subjective norms and PBC. Similarly, one type of behaviour may be more under the influence of attitudinal than normative or control factors. The reasoned action approach provides a conceptual framework that accommodates the theoretical constructs influencing behaviour. In the following sections, each of the construct in TPB is reviewed in greater detail to gain a better understanding of the role they play in explaining behaviours.

2.5.1 Attitudes and Behavioural Beliefs

In TPB, one of the key elements in determining intentions to engage in a targeted behaviour is attitudes (Armitage & Conner, 2001). Attitudes is defined as a disposition or tendency to respond with favourableness or unfavourableness towards an object or behaviour (Ajzen & Fishbein, 2005; Ajzen, 2005a; Fishbein & Ajzen, 2010). A person may evaluate an object or behaviour negatively, positively or even take a neutral stance (Ajzen, 2019). Hence, a bipolar evaluative dimension known as semantic differential scales is used to measure attitudes. It consists of adjective pairs placed on opposite ends of a scale; such as good-bad, pleasant-unpleasant and harmful-beneficial (Osgood, Suci & Tannenbaum, 1957).

Similarly, studies by Ajzen and Fishbein (1980); Ajzen and Driver (1991); Ajzen and Driver (1992) also discover different types of attitudes. Attitudes can be cognitive or affective in nature. Such distinctions has its roots in an earlier theory, that is the

multicomponent view of attitude (Rosenberg & Hovland, 1960). However, Fishbein and Ajzen (2010) hold different opinions on the use of the term 'cognitive' and 'affective'. These terms can be misleading as they proposed a unidimensional view of evaluation. As such, more neutral terms such as 'instrumental' and 'experiential' is proposed. Instrumental items reflect perceived positive or negative consequences of a behaviour whereas experiential (or in some literature, affective) items reflect the positive or negative experiences associated with performing the behaviour. These form the direct measures of attitudes.

These standardised direct measures are good indicators of an underlying attitude, but they do not provide information on causes of such attitudes. For this purpose, beliefs structures must be examined. Beliefs are formed by associating the object or behaviour with various characteristics, qualities and attributes. A person experience various belief during their lifespan. It may be formed by direct observation; indirectly from friends, teachers, the media and other outside sources; or self-generated through inference processes. Some beliefs persist over time, some are forgotten and new beliefs can be formed. These changes would produce corresponding changes in attitudes.

The way beliefs influence attitudes is best described by the Expectancy Value theory (EVT) (Ajzen & Fishbein, 1980; Ajzen, 1991). EVT is a model of attitude formation and it provides a framework for understanding attitudes and beliefs (Conner & Sparks, 2005). Each behaviour or object have various attributes, which has its respective expectancy and value component (Ajzen & Fishbein, 1980; Eagly & Chaiken, 1993). The expectancy component refers to the level of confidence a person holds about the consequences of performing a behaviour. It represents specific beliefs a person holds regarding their success of the task that they perform. In EVT, it is termed as behavioural beliefs (b_i). The value component refers to how important or useful an individual perceives the task. It answers

the question as to why they would want to perform the task (Wigfield & Eccles, 2000). It forms the evaluation of the outcome (e_i) .

To predict attitudes, the expectancy and value component must be multiplied. The total of all these attributes in relations to the behaviour will form the overall attitudes. EVT assumes that people will only be motivated to perform certain behaviour if they deem that the results are highly valued. If they do not believe that it will lead to the desired outcome or will lead to outcomes that are not valued, the person will not perform it.

The components considered are illustrated using the following equation:

$$A = \sum_{i=1}^{i=p} b_i \cdot e_i$$

where

- A: Attitude towards the task or targeted behaviour
- b_i : The behavioural belief of the targeted behaviour, which lead to outcome i
- *e_i*: Evaluation of the outcome *i*
- *p*: Number of accessible consequences

The above equation attempts to explain attitude formation by discussing details on the 'calculations' that are considered. In reality, individuals do not make these computations when making decisions on a behaviour (Conner & Sparks, 2005). Instead, attitudes emerge automatically and spontaneously as beliefs are formed. These attitudes are thought to be immediately available when a person is confronted with the attitude object or behaviour. As such, the equations merely provide an illustration of a person's attitude and its underlying behavioural beliefs.

Similarly, many researchers also assumed that the reasoned action approach consider people to be rational and to deliberate at length before engaging in any behaviours. Contrary to that, beliefs can be biased. There are possibilities of invalid information, accessing only selective information, be self-serving or information that fail to reflect realities of life.

People hold numerous beliefs about an object, attribute or event. However, only a small set is accessible at a particular time (Fishbein & Ajzen, 1975; Ajzen, 2011). Accessible beliefs are activated spontaneously. Its activation occur below conscious awareness, but come readily to mind when a person retrieves them (Beck & Ajzen, 1991; Fishbein & Ajzen, 2010; Ajzen, 2012).

Hence, in order to understand why people, hold certain attitudes, it is necessary to elicit their accessible beliefs. This would involve asking a representative sample of individuals from the population of interest to list the advantages and disadvantages of performing the behaviour under investigation. A content analysis, where responses are organized to identify beliefs held with the greatest frequency in the population, is performed (Conner & Sparks, 2005; Fishbein & Ajzen, 2010). This is the recommended pilot work before beginning any form of study and procedures (Ajzen & Fishbein, 1980; Ajzen, 2011). It is insightful as the elicitation procedure usually produces sets of beliefs specific to the population and context that is studied (Fishbein & Ajzen, 2010).

2.5.2 Subjective Norms and Normative Beliefs

The social environment of a person exerts strong influence on a person's intentions and actions. Subjective norms reflect social norms, what is deemed acceptable or permissible behaviour in a group or society. It is defined as the social pressure to perform (or not to perform) a given behaviour. The stronger the perceived social pressure, the more likely an

intention to perform the behaviour will be formed (Ajzen & Fishbein, 2005; Ajzen, 2005a; Fishbein & Ajzen, 2010).

Two types of social norms can be identified using a reasoned action approach. They are injunctive and descriptive norms (Cialdini, Kallgren & Reno, 1991; Cialdini et al., 2006; Cialdini, 2007). Injunctive norms refer to perceptions of others concerning what should or ought to be done with respect to performing a given behaviour, whereas descriptive norms refer to perceptions that others are or are not performing the behaviour in question. People may be motivated to behave in accordance with what they believe others think or do for various reasons. It could be that these persons or groups has the power to reward or punish them, has the right to request it, is an expert, or simply because they want to be like them.

Earlier conceptualisation of subjective norms is limited to measures that are injunctive in nature, which is what important referents think that the person should do. Hence, social pressure is only from the approval (or disapproval) of people important to the individual. As a result, in some studies, subjective norms displayed weak results (Sheeran & Orbell, 1999; Armitage & Conner, 2001). It is subsequently recognised that injunctive norms are not the only source of social pressure. Subjective norms measure the influence of the social environment, hence, social influences that directly and indirectly influenced the thoughts, feelings and actions of a person should be considered (Turner, 1991; Ajzen, 2005a).

Social pressure also exist when a person believe that important others are also performing (or not performing) a particular behaviour (Fishbein & Ajzen, 2010). These forms of norms are known as descriptive norms. It measures the attitudes and behaviour of people important to the individual. Many use important referents as examples of wise decisions. They serve as a reference point for people when they are making decisions for themselves (Cialdini et al., 1991; Cialdini et al., 2006; Cialdini, 2007). Donald and Cooper (2001) opined that, by the definition of social influence, there are two social processes that

should be considered that influence intentions and behaviours. The first concern the approval of others, and the second, other indirect social influences, such as descriptive norms. Observing the actions and behaviours of the important referent is a strong motivator, which is at times stronger than complying to the perceived expectations of significant people in a person's life.

To further understand the causal determinants of subjective norms, it is important to examine beliefs, in particular normative beliefs. Normative beliefs are beliefs of specific important referent individuals or groups in a person's life. It considers whether these referents would approve or disapprove of them performing the behaviour; as well as whether these referents are performing or not performing the behaviour. If people perceive that people important in their lives encourage and support the behaviour, it would likely result in positive intentions (Ajzen & Fishbein, 1980). In totality, normative beliefs produce subjective norms, that is, perceived social pressure to engage or not engage in the behaviour.

The way normative beliefs influence subjective norms is best described by the following formulation (Ajzen & Fishbein, 1980; Ajzen, 1991):

$$SN = \sum_{j=1}^{j=q} nb_j. \ mc_j$$

where

- SN: The subjective norm in relation to a target behaviour
- nb_j : The normative belief of referent j
- *mc_j*: Motivation to comply with referent *j*
- *q:* Number of accessible referents

In the formulation, subjective norms (SN) is a function of normative beliefs (nb_j) and motivation to comply (mc_j) . Normative beliefs (nb_j) is the perception of individual on whether the important referent would expect them to perform a behaviour and are performing the behaviour. Motivation to comply represent the extent in which the individual is motivated to conform to the important referents' expectations (mc_j) . Each measure of normative beliefs is multiplied by the motivation to comply in order to ascertain the subjective norms. Hence, the social pressures to behave are only when the individual is motivated to comply with that particular referent (Ajzen & Fishbein, 1980).

Similar to behavioural beliefs, there are numerous groups and individuals that influence a person's normative beliefs. The list of important groups and individuals can be elicited from a representative sample of the population. This would then form a modal set of accessible normative beliefs. It is also the recommended pilot work before beginning any form of study (Ajzen & Fishbein, 1980).

2.5.3 Perceived Behavioural Control and Control Beliefs

In addition to attitudes and subjective norms, perceived control over performance of a behaviour also influence intentions. PBC reflects the extent to which people believe that they have the capability and control in performing a behaviour. It considers the information, skills, opportunities and resources required to perform the behaviour. It also takes into account the possible barriers or obstacles that have to be overcome. If people believe that they do not have control over performance of a behaviour, they may not form strong behavioural intentions even if they hold positive attitudes and perceive strong social pressure to do so (Ajzen & Fishbein, 2005; Ajzen, 2005a; Fishbein & Ajzen, 2010).

Another issue of concern regarding PBC is the conceptualisation of the control variables. Most researchers are of the opinion that PBC and self-efficacy are similar or equivalent. Both refer to a person's evaluation of their ability to perform a behaviour. However, there are others who have highlighted subtle differences (Bandura & Pallak, 1982; Ajzen, 1991; Sheeran, 2002). PBC is perceived as the difficulty in performing a behaviour while self-efficacy is the perceived confidence in performing the behaviour (Garcia & Mann, 2003). Such confidence is deemed important to determine how well one can execute a course of action in any prospective situation. A person with self-efficacy is confident of their ability to perform a behaviour. This influence a whole spectrum of activities, from the choice of activity to preparation and effort directed towards completion of an act. As such, many researchers also argue that both concepts should be clearly defined and outlined in the field of study (Giles, McClenahan, Cairns & Mallet, 2004).

The influence of PBC on behaviour can be direct or through intentions. Direct influence is possible when the individual is aware of all factors that will ease or impede behaviour. This is known as actual control, where there is a direct path from PBC to behaviour. Under such circumstances, the perceptions of control also reflect a person's actual control (Beck & Ajzen, 1991). These factors can come in many forms such as opportunities, ability, resources or skills to perform the behaviour of study (Ajzen, 1985; Beck & Ajzen, 1991).

There are also situations where an individual is not fully aware of the opportunities, resources or skills required to perform a behaviour (Beck & Ajzen, 1991). Perceptions of control, reflected by the construct PBC, are used to predict future behaviour instead of actual control. In TPB, the path from PBC to intentions reflect the motivational influence of the control component (Madden et al., 1992).

Determinants of PBC, can be examine further by identifying beliefs, that is control beliefs. The relation between control beliefs and perceived behavioural control is expressed using the following equation:

$$PBC = \sum_{k=1}^{k=r} \mathbf{c}_{k} \cdot p_{k}$$

where

PBC: Perceived Behavioural Control in relation to a target behaviour

 c_k : Perceived frequency or likelihood of occurrence of factors k

 p_k : Perceived facilitating or inhibiting the power of the factor k

r: Number of control factors

PBC has an expectancy and value component. The expectancy component is represented by the perceived frequency or likelihood of occurrence of factors that facilitate or impede the performance of a behaviour (c_k). The value component is represented by the perceived power of the facilitating or inhibiting factor (p_k). PBC is computed by multiplying the frequency of occurrence of the factors with the perception of the power of each factor that facilitate or inhibit behaviour (Conner & Sparks, 2005). In summary, if a person perceives that the behaviour requires a resource or skill, and believe that they have what is required, they will be motivated to attempt the behaviour.

Readily accessible beliefs regarding these control factors are assumed to determine the overall level of PBC. Similar to behavioural and normative beliefs, control beliefs are elicited from a representative sample of respondents. These participants are asked to list the factors they believe would enable and impede them when performing the behaviour. A set of modal accessible control beliefs is then constructed by selecting the most frequently mentioned factors.

2.5.4 Intentions

Intention is a central concept in the studies of TRA and TPB. It represents the motivational factors that influence the behaviour studied and how cognitively ready a person is to perform a behaviour (Ajzen & Fishbein, 1980). It also reflects how hard they are willing to try and the degree of effort they are willing to exert to perform the behaviour (Ajzen, 1991). It can be deemed as instructions one gives themselves to act (Triandis, 1977).

In both TRA and TPB, intentions precede behaviour, making it a good predictor. It has also been used as a proxy to behaviour, especially when there is no available measure. The stronger an individual's intention to engage in a behaviour, the more successful they are in performing the behaviour (Ajzen, 2005b).

In TRA, intentions (*BI*) are influenced by attitudes (*A*) and subjective norms (*SN*). However, for TPB, intentions (*BI*) are not only influenced by attitudes (*A*) and subjective norms (*SN*) but also perceived behavioural control (*PBC*) (Ajzen, 1985; Ajzen & Klobas, 2013; Klobas & Ajzen, 2015). Also, PBC is also able to influence behaviour directly (Armitage & Conner, 1999; Armitage & Conner, 2001). In summary, the more favourable the attitudes, subjective norm and PBC is, the stronger the individual's intention to perform the behaviour under consideration.

These influences can be presented in the following equation:

$$B = BI = w_1 A + w_2 SN + w_3 PBC$$

where

B: Behaviour of interest

BI: Behavioural intent

A: Attitude towards performing the behaviour

SN: Subjective norm

PBC: Perceived behavioural control

 w_1 , w_2 , w_3 are the relative weights of attitudes, subjective norms and PBC

Also, as previously discussed, the antecedents of attitudes, subjective norms and perceived behavioural control can be explained by beliefs. For attitudes, the antecedent is behavioural beliefs, which are beliefs on the consequences of the behaviour. Subjective norms antecedents are normative beliefs which list the important referents in a person's life. As for PBC, the antecedents are control beliefs, which is a person's belief of the factors that may ease or impede the performance of a behaviour.

For studies of intentions and behaviour, intentions are good predictors of behaviours (Sheeran, 2002). The results have recorded a mean correlation between 0.44 to 0.62 (Sheppard et al., 1988; Notani, 1998; Armitage & Conner, 2001; Hagger, Chatzisarantis & Biddle, 2002; Schulze & Wittmann, 2003). Similarly, in meta-analysis study by Webb and Sheeran (2006), the interventions resulted in an average effect size of 0.66 for changes in intentions and 0.36 for changes in behaviour.

2.6 Hypotheses Development

This study intends to use a reasoned action approach to investigate the influence of beliefs, attitudes, subjective norms and PBC (Fishbein & Ajzen, 1975; Ajzen & Fishbein, 1977; Ajzen & Fishbein, 1980) on intentions to plan financially for retirement. These determinants are drawn from TPB, a social psychology model which has been used to explain a wide variety of behaviours. It recognises that each of the construct will have varying effects among different individuals and groups. In a similar manner, behaviours

may also be influenced differently by the constructs (Fishbein & Ajzen, 2010; Ajzen & Sheikh, 2013; Klobas & Ajzen, 2015).

The first determinant, attitudes, pertains to a person's predisposition to financial, money and retirement matters (Fishbein & Ajzen, 1975; Ajzen, 2005b). A person may take a positive, neutral or negative stance in any of these matters. Generally, in existing studies, individuals are more likely to perform behaviours that are evaluated positively (Princeton Survey Research Associates International, 2013; KRC Research, 2016). Conversely, if they evaluate the outcome negatively, they are less likely to engage in activities pertaining the behaviour. However, studies in financial behaviours also reveal a myriad of influence (Furnham & Goletto-Tankel, 2002; Joo & Grable, 2005; Croy et al., 2010b; Owusu-Frimpong et al., 2011; Tan et al., 2011; Van Deventer et al., 2014; Armantier et al., 2015; Yilmazer & Lich, 2015; Sivaramakrishnan et al., 2017). Hence, the following hypothesis is developed to test the effects of attitudes on intentions to plan financially:

<u>Research Question 1</u> - What is the influence of attitudes on the intention to plan financially for retirement among University employees?

Hypothesis 1 (H1)

Attitudes have a positive influence on intentions to plan financially for retirement among University employees.

Perceived social pressures from important referents influence a person's intentions. In TPB, this is reflected in the subjective norms construct. Subjective norms have injunctive and descriptive components (Cialdini, Reno & Kallgren, 1990). Injunctive norms are the perception of important referent's expectations whilst descriptive norms measure the perception of important referent's actions in relations to the targeted behaviour. These norms, in aggregate, would represent the social pressure from the environment.

However, the influence of these norms varies. Different populations and context of study are influenced by different important referents (Fishbein & Ajzen, 2010; Klobas & Ajzen, 2015). Nonetheless, individuals are expected to be influenced positively if they believe that important referents in their lives think that they should be planning financially for retirement (Fishbein & Ajzen, 2010). Similarly, if individuals believe that the important referents are also engaging in the behaviour, they will follow suit (Rimal & Lapinski, 2015; Lapinski et al., 2017). Hence, the following hypothesis is proposed to test the effect between subjective norms and intentions:

<u>Research Question 2</u> - What is the influence of subjective norms on the intention to plan financially for retirement among University employees?

Hypothesis 2 (H2)

Subjective norms have a positive influence on intentions to plan financially for retirement among University employees.

PBC measures an individual perception of their ability to perform a behaviour. If an individual is confident of having the relevant skills, opportunities and resources of performing the desired behaviour, they will have a high degree of PBC (Ajzen, 1991). Conversely, if they are not confident of their ability to control the opportunities or resources, they will have a low level of PBC.

A majority of studies have shown that intentions can be predicted with considerable accuracy using PBC (Armitage & Conner, 1999; Armitage & Conner, 2001; Ajzen & Albarracin, 2007). However, the influence of PBC also differ depending on the population and behaviour. There are studies where PBC have no direct correlation with intentions (Fishbein & Ajzen, 2010). Hence, to determine the influence of PBC on the intention to plan financially for retirement, the following hypothesis is developed:

<u>Research Question 3</u> - What is the influence of perceived behavioural control on the intention to plan financially for retirement among University employees?

Hypothesis 3 (H3)

Perceived behavioural control has a positive influence with intentions to plan financially for retirement among University employees.

To further explain the basic construct of TPB (attitudes, subjective norms and PBC), beliefs about financial planning for retirement are explored. These beliefs could be formed through personal experiences, formal sources of information, interactions with family and friends or the way information is interpreted. Individuals hold many beliefs, but only a small set are accessible at any one time (Fishbein, 1963; Fishbein & Ajzen, 1975; Ajzen, 2011). These beliefs then guide the decision to perform or not perform a behaviour.

According to TPB, three types of beliefs can be identified. These are behavioural beliefs, normative beliefs and control beliefs. Behavioural beliefs influence attitudes; normative beliefs influence subjective norms while control beliefs will help in understanding PBC. The following hypothesis is tested:

<u>Research Question 4</u> – What is the influence of beliefs on attitudes, subjective norms and perceived behavioural control among University employees?

Hypothesis 4a (H4a)

Behavioural Beliefs has a positive influence on attitudes among University employees.

Hypothesis 4b (H4b)

Normative Beliefs has a positive influence on subjective norms among University employees.

Hypothesis 4c (H4c)

Control Beliefs has a positive influence on perceived behavioural control among University employees.

In brief, the hypothesis developed is shown in the following schematic representation.



Figure 2.2: Research Hypotheses

2.7 Summary

This chapter discuss the development of financial planning for retirement and the importance of intentions and motivation in planning. The underpinning theories that has been used to understand financial, planning and retirement matters are also examined. Ultimately, TPB was used as the theoretical framework for this study as it offers a comprehensive organising framework to study the different psychological and environmental circumstances faced by an individual.

Various financial behaviour literatures were also reviewed to understand the influence of attitudes and beliefs, contextual environment and resources. Although, generally the importance of the constructs in TPB (attitudes, subjective norms and PBC) is recognised, there are gaps, where findings are inconclusive and fragmented. Some studies recognise the importance of a determinant, yet others find that it has no influence in their research. As such, hypotheses were developed to test its effects on intentions to plan financially for retirement.

university

CHAPTER 3: RESEARCH METHODOLOGY

The earlier chapters had analysed the literature on beliefs, attitudes, subjective norms and perceived behavioural control of financial planning for retirement. The problem statement and theoretical framework has been discussed to develop the research questions, model and hypotheses. It is the intent of this thesis to study the determinants influencing financial planning for retirement. The Theory of Planned Behaviour (TPB) is used as the underpinning theory where financial planning behaviours are assumed to form from the beliefs people possess. TPB offer a theoretical approach to predict and explain human behaviour by using selected determinants (beliefs, attitudes, subjective norms, perceived behavioural control) (Fishbein & Ajzen, 1975; Ajzen & Fishbein, 1977; Ajzen & Fishbein, 1980). However, there is limited research on its use in financial and retirement behaviour studies, hence, the constructs in the theoretical framework must be tested.

This study adopts a quantitative methodology in its research design. It would be the principal method used to answer the research questions and to test the hypothesis. For this thesis, primary data is collected via a survey. Surveys are common in financial planning research. They have been used to elicit responses for constructs such as attitudes and other determinants (Tang et al., 2015; Nosi, D'Agostino, Pagliuca & Pratesi, 2017; Topa, Lunceford & Boyatzis, 2018). It is cost efficient and suitable for studies that involve a larger sample. Respondents of the survey can be anonymous, which is crucial if it involves personal and sensitive questions (Creswell, 2014).

Studies conducted using surveys are also supported by Armitage and Conner (2001), where the results are deemed more superior than observations. Yin (1994) further supports this methodology. It is particularly suitable for studies that use the 'what-type' research questions.

3.1 Development of Instrument

In a survey, the development of the survey instrument is important. The survey instrument comes in the form of a questionnaire, which is a preformulated written set of questions to which respondents record their answers (Sekaran, 2003). The items in the survey instrument is developed via two separate methods (Morgan, 1998; Tashakkori & Teddlie, 1998; Fishbein & Ajzen, 2010). Items to test Hypotheses 1, 2 and 3 (attitudes, subject norms and perceived behavioural control) are adapted from existing validated scales from the literature. The authors of TPB has developed the items to measure the constructs and its validity have been proven in many studies. However, the authors invites researchers to use the current of knowledge and scales developed in their research to extent the body of knowledge in understanding behaviour (Ajzen, 2006; Fishbein & Ajzen, 2010). Meanwhile, items addressing Hypotheses 4a, 4b and 4c (behavioural beliefs, normative beliefs and control beliefs) is developed via qualitative interviews. This is because it is difficult to specify any fixed beliefs at any given context. A person holds numerous beliefs and there are possible differences among the different sets of population. Hence, it is necessary to elicit these beliefs from interviews with a modal set of participants of the population of study.

The behaviour of interest must be defined clearly in studies using TPB (Ajzen & Fishbein, 1977; Ajzen, 2006). For this purpose, the principle of compatibility is used. In this principle, behaviour is defined according to four considerations. The first is the target. It identifies the target in which the action is directed. The second is action, which defines the activity itself. The third explains the context in which the action takes place. The final elaborates on the time at which the act is performed. This principle is also known as 'TACT' which is the acronym for Target, Action, Context and Time. In developing this

principle, the researcher acknowledge that it may be difficult for some behaviour to specify all considerations. In such cases, the minimum specification should be the action and time frame (Conner & Sparks, 2005).

For this study, the behaviour in question is intentions to plan. It relates to money related planning activities, such as savings or investments, with the purpose of meeting a person's retirement goals (for example financial independence in old age). Using TACT, the behaviour is defined as the 'intention to plan financially for the next 12 months', and adapted across the measures developed. These elements are important and must be consistently applied to the items developed in order to ascertain the statistical relationship between them.

3.1.1 Adapted items

(a) Attitudes

Attitude is one of the determinants of a person's intentions (Ajzen & Albarracin, 2007). It is defined as the degree to which performance is positively or negatively valued and reflect the reaction to any object that is of significance to a person (Osgood et al., 1957).

There are two different components of attitudes, that is the instrumental and experiential components (Ajzen & Driver, 1992). Instrumental components consider the rewards (or punishment) to the attitude. The experiential component refers to a person's feelings or emotions. Ajzen and Fishbein (2000); Ajzen and Fishbein (2005) have indicated that attitude measures in TPB should have items representing both the instrumental and experiential components. At the same time, these items should be measured using semantic differential scales (Ajzen & Fishbein, 1980).

Five items were adapted to measure the evaluation of these components in attitudes. Two of these items measure the instrumental components, that is 'worthless-useful' and 'beneficial-harmful' while another two measures experiential or affective aspects, that is reflected by the scales 'enjoyable-uninteresting' and 'unpleasant-pleasant'. A final question that measures the overall evaluation, 'good-bad', is also included.

Measures	Items	Source	
Overall	Goodbad	Ajzen (2006), Bearden, Netemeyer and	
Experiential	Enjoyableuninteresting	Haws (2011), Bruner II (2009), Francis,	
Instrumental	Worthlessuseful	Eccles, Johnston, Walker, Grimshaw, Foy, Kaner, Smith and Bonetti (2004),	
Instrumental	Beneficialharmful		
Experiential	Unpleasantpleasant	(1992) (Usgood et al. (1957), Ajzen and Driver	

Table 3.1: Adapted Scale items for attitudes

(b) Subjective norms

Subjective norms refer to the influence of the social environment on the individual. It relates to a person's perception of the social pressure to engage (or not engage) in a behaviour.

Measures of subjective norms in TPB have injunctive and descriptive qualities. However, initial studies on subjective norms have only included the injunctive qualities. This results in many studies concluding that there are inadequacies of this form of measurement (Sheppard et al., 1988; Godin & Kok, 1996; Armitage & Conner, 2001; Fishbein & Ajzen, 2010). Generally, important referents are expected to approve desirable behaviours and disapprove undesirable ones. Hence, injunctive norms alone are not reflective of the pressures faced from the social environment. It merely address motivating action through the approval of others (Cialdini et al., 1991). In order to further study the influence of social environment, Ajzen and Fishbein (2005); Fishbein and Ajzen (2010) suggested the inclusion of descriptive norms. Descriptive norms describe what important referents do. It is also an important indicators social pressure. Rivis and Sheeran (2004) reported that across 14 tests, a descriptive norm-intentions of 0.46 was achieved. It was also able to explain an additional 5% variance in intentions, after taking into consideration attitudes and PBC. Conner and Sparks (2005) further reiterated that this is an interesting area of research.

To address the injunctive norms, questions pertaining to the opinions of important referents and whether they think the individual should do what important referents want are adapted. The first is the statement 'I feel the social pressure to plan financially for retirement', and the second 'Most people who are important to me think I should plan financially for retirement'. As for descriptive norms, the statement 'Most people who are important to me plan financially for retirement' is adapted.

Table 3.2: Adapted Scale items for subjective norms

Measures	Items	Source
Injunctive norms	I feel the pressure to plan financially for	Ajzen (2006), Bearden
	retirement.	et al. (2011), Bruner II
Injunctive norms	Most people who are important to me think	(2009), Francis et al.
	I should plan financially for retirement.	(2004)
Descriptive norm	Most people who are important to me plan	
	financially for retirement.	

(c) Perceived Behavioural Control (PBC)

PBC reflects the extent to which people believe that they have the capability and control in performing a behaviour. It considers the resources and barriers to a behaviour. When people believe they have control over the performance of a behaviour, they will form strong intentions (Ajzen & Fishbein, 2005; Ajzen, 2005a; Fishbein & Ajzen, 2010).

Two aspects are important considerations in the measures of PBC. The first is selfefficacy. Self-efficacy is defined as how easy or difficult it is to perform a behaviour. Selfefficacy is similar to Bandura (1995) definition, which is defined as the degree of confidence a person has. One must be convinced that they have the ability and resources to execute a behaviour in order to produce the outcome desired. In order to assess selfefficacy, the following questions are adapted: 'I am confident that I can effectively plan financially for retirement' and 'For me to plan financially for retirement is easy'.

The second consideration is perceived controllability (Ajzen, 2002b). It is defined as a person's perception of their ability to control behaviour and that the performance (or non-performance) is entirely up to them. The item 'The decision to plan financially for retirement is beyond my control' is used for this purpose. Another question to assess controllability is 'Whether I plan financially for retirement or not is entirely up to me' is also used. It is recommended that in measuring PBC both self-efficacy and controllability is used. Hence, a total of four items were used for PBC.

Items	Source
I am confident that I can effectively plan financially for	Ajzen (2006),
retirement.	Bearden et al.
For me to plan financially for retirement is easy.	(2011), Bruner II
The decision to plan financially for retirement is beyond my control.	(2009), Francis et al. (2004)
ollability Whether I plan financially for retirement or not is entirely up	
	ItemsI am confident that I can effectively plan financially for retirement.For me to plan financially for retirement is easy.The decision to plan financially for retirement is beyond my control.Whether I plan financially for retirement or not is entirely up to me

 Table 3.3: Adapted Scale items for Perceived Behavioural Control

(d) Intention to plan

Intention to plan financially for retirement indicates a person's readiness, which in this case is the readiness to plan financially for retirement in the next 12 months (Ajzen & Fishbein, 1980). This behaviour is measured using three items. The first item 'I want to plan financially for retirement in the next 12 months' outlined the desire to act, while the second item 'I intend to plan financially for retirement in the next 12 months' and third item 'I will try to plan' reflects intention to act, which direct action (Conner & Sparks, 2005; Ajzen, 2006).

The items are supported by various literature. Studies by Armitage and Conner (2001) also reveal a very high level of correlation among these items. A majority of studies reviewed used a mixture of these items (Conner & Armitage, 1998; Armitage & Conner, 2001; Ajzen, 2006).

Measures	Items	Source
Desire to act	I want to plan financially for retirement	Ajzen (2006), Bearden et al. (2011),
	in the next 12 months.	Bruner II (2009), Armitage and
Intentions	I intend to plan financially for retirement	Conner (2001); Francis et al. (2004)
	in the next 12 months.	
Intentions	I will try to plan.	

Table 3.4: Adapted Scale items for intentions

In summary, the survey instrument measuring attitudes, subjective norms, perceived behavioural norm and intentions consists of 15 items. Attitudes are measured using a semantic differential scale of 1 to 5 (Osgood et al., 1957). Subjective norms, perceived behavioural norm and intentions are, in turn, measured by a set of Likert scales. The level of the agreement is reflected by one which indicates 'Strongly disagree', two indicates 'Disagree', three indicates 'Neutral', four indicates 'Agree' and five indicates 'Strongly agree' (Vagias, 2006). The following table summarises the adapted scale items.

Constructs	Number of Items
Attitudes	5
Subjective Norms	3
Perceived Behavioural Control	4
Intentions	3
Total	15

Table 3.5: Summary of Adapted Scale Items

3.1.2 Qualitative Interviews

Items to measure attitudes, subjective norms and PBC are adapted from numerous validated scales available in the literature. These constructs are, in turn, influenced by beliefs. A person possesses many beliefs but only a relatively small number are accessible at any given moment. These accessible beliefs are important, as they are the ones that guides behaviour (Conner & Sparks, 2005; Fishbein & Ajzen, 2010).

However, it is difficult to specify any fixed accessible beliefs at any given context as a person hold numerous beliefs and possible differences exists among the various sets of the population. Therefore, for this study, a qualitative method is used to identify the beliefs. This method involves using interviews, that is conducted among members of the population of interest (Ajzen, 1991; Ajzen, 2002a; Ajzen, 2006). A modal of accessible beliefs is then elicited from these interviews and used in developing the beliefs sections of the questionnaires.

These interviews are also useful to explore and derive new insights to support the existing literature (Ajzen, 2011; Ajzen & Klobas, 2013; Ajzen & Sheikh, 2013; Ajzen, 2015; Ajzen & Dasgupta, 2015; Klobas & Ajzen, 2015). Currently, there is a lack of studies related to financial planning for retirement, particular in the Malaysian context. Such efforts are particularly useful to clarify an understanding of the problem and ascertain that the research is worth pursuing.

The interview questions were scripted using guidelines recommended (Ajzen, 2006; Klobas & Ajzen, 2015) and were conducted to the point of saturation. Nevertheless, respondents were also given the freedom to reflect on their views. This is especially important to respondents that may be unfamiliar or have not thoroughly thought of their future. Before beginning the interviews, a brief explanation is given on the background and objective of the study (Francis et al., 2004).

In brief, to elicit behavioural beliefs, participants were asked the following questions: 'What do you see as the advantages of you planning financially for retirement, for the next 12 months?', 'What do you see as the disadvantages of you planning financially for retirement, for the next 12 months?', 'What else comes to mind when you think about planning financially for retirement, for the next 12 months?'.

For normative beliefs, participants are asked of groups or individuals that would approve or disapprove of them planning (Ajzen & Fishbein, 1980). The objective is to determine the salient referent groups. For this purpose, the following questions were asked: 'Please list the individual or groups who would approve or think you should be planning financially for retirement, for the next 12 months', 'Please list the individual or groups who would disapprove or think you should not be planning financially, for retirement, for the next 12 months', 'Please list any other individuals or groups who come to mind when you think about planning financially for retirement, for the next 12 months'.

For control beliefs, participants are asked regarding the factors or conditions that would ease or make it difficult for them to perform the behaviour (Ajzen & Driver, 1992). The following questions were asked: 'Please list any factors or circumstances that would make it easy or enable you to plan financially for retirement, for the next 12 months', 'Please list any factors or circumstances that would make it difficult or prevent you from planning financially for retirement, for the next 12 months' and 'Please list any other factors or circumstances that come to mind when you think about the ease or difficulty in planning financially for retirement, for the next 12 months'.

Each interview took about half an hour to two hours. Some of the interviews were recorded, with the consent of the interviewees. Upon completion, the interviews were transcribed and prepared for subsequent analysis. Responses are categorised into modal beliefs that are frequently cited by respondents. The responses are analysed together with a panel of three reviewers.

The responses identified represent the accessible beliefs adopted by the population of interest. These interviews identify and reaffirm the various variables and dimensions in the literature review which further explains the theoretical framework developed and also on the relationship of the constructs. The sets of modal beliefs obtained are used to guide the development of the belief portion of the questionnaire in the survey.

(a) Interview participants

Details of the preliminary interview participants' profile is available in Appendix L. Twenty-one participants agreed to be interviewed. The participants are selected based on several criteria. They must be Malaysians who are working as public or private university staff and have conducted or thought about their retirement. Sixteen public universities staff participated in this interview, representing 76% of the participants, while the remaining are from private universities (24%).

The participants are mainly aged 30-34 (28.6%) and 35-39 (33.3%). This age group represents about 60% of the participants. The remainder are aged 40 to above 60. Three of the participants range from 40-44 (14.3%), while one participant is from the range of 45-49 (4.8%) and another from 50-54 (4.8%). For the 55-59 age group, only one participant

has agreed to be interviewed (4.8%). The above 60 category has two participants, representing 9.4%.

As for the educational background, 76.2% have a Master degree while only 9.4% have PhD qualifications. For marital status, about 80.9% of the participants are married, 14.3% single and 4.8% divorced. The main ethnic groups in Malaysia are well represented, with more than 60% Malays, while about 30% are Chinese. The remainder are Indians (9.5%). The participants consist of four males (19%) and 17 females (81%). For religion, more than 60% are Muslim, 20% are Buddhist while the rests are Hindu and Christian.

As for the income range, participants from income range of less than RM30,000 and RM30,000-50,000 consist of one participant each (4.8%). Most of the participants (about 28.5%) have incomes between RM50,001-70,000. 19% are from the income range of RM70,001-90,000 and another 14.3% are from income range RM90,001-110,000. Those with an income range above RM110,000 comprise 19% (RM110,001-130,000) and 4.8% (RM130,001-150,000) respectively. One respondent would like to keep the income information confidential.

(b) Behavioural beliefs

Behavioural beliefs link intentions to plan financially for retirement to expected outcomes. During the interviews, participants spoke of how planning brings about a sense of awareness. Firstly, they are more aware of their future, in particular their financial situation. This, in turn, helps them to set directions and be more alert of possible financial benefits from the actions they take. Planning also helps them to visualise their future, guide their direction and take advantage of opportunities. Ultimately, they have more flexibility and choices in life. Three beliefs were identified from this part of the interview, that is "to be able to foresee the future", "to be able to set my own directions" and "to be aware of the environmental opportunities". Another aspect which also features predominantly in the interviews is that planning enables one to achieve independence and freedom. In particular, the participants emphasised that the goals set during planning gives them financial freedom. They emphasise that it ensures that they are self-sufficient in their time of need. Financial freedom allows them to dip into their financial resources, and reduces reliance on charity or welfare from others. This identify another important belief, that is "to be independent and self-sufficient".

These four beliefs are linked to an outcome, that is, to maintain autonomy in a person's life. In total, 71% of participants mentioned autonomy. Having autonomy is about being able to have choices. It gives a person freedom in life and have sufficient room to act and make decisions (Weinstein, Przybylski & Ryan, 2012).

Another set of beliefs highlighted that planning helps them to cope with uncertainties, problems and the emotional challenges in life. It gives them a sense of confidence and preparedness, especially when faced with healthcare problems. This provides them with choices, which enables them to make contingency plans to confront any eventualities in life. Two modal beliefs are identified, that is, "to be able to handle uncertainty confidently" and "to be prepared to handle problems". Participants further mentioned their concerns during retirement. Financial support is critical during this period as healthcare problems may arise. What is worrisome during retirement is the loss or reduction of income coupled with the possible increase in financial obligation and commitments. This may bear a negative impact on their quality of life during retirement. From these responses, two additional beliefs were identified, that is, "to be able to have adequate financial support during the remaining lifespan in retirement" and "to be able to maintain the quality of life during retirement".

These four responses are linked to how planning for retirement helps the participants to cope emotionally and financially with problems that may arise in life. 76% of the participants believe that planning financially for retirement will help them cope with life challenges. It will enable them to undertake activities to deal with or minimise the demands resulting from life stressors (Lazarus & Folkman, 1984).

The next outcome of planning financially for retirement involved self-control. About half of the participants highlighted that planning help them to overcome temptation. It highlights the area where they may be overspending, which disciplined them, such as providing them indicators when they exceed their budget. This reflects the advantages of planning financially for retirement.

At the same time, participants also highlighted the disadvantages of planning for retirement. They reflect on the inner conflicts they face, where at times they deem planning as burdensome as they have to exercise restraint in their urges to spend. Frequently, self-control is required to suppress emotions or behaviour. Baumeister, Vohs and Tice (2007) defined self-control as the capacity to alter one's responses to be in line with standards or social expectations and to support the pursuit of long-term goals. Three additional modal beliefs are identified, that is, "to be able to resist temptations", "to be able to gain benefits of savings and investing despite current limitations in money" and "planning is a burden".

In summary, the participants outlined their behavioural beliefs which relate to outcomes they hope to achieve when planning financially for retirement. In total, 11 questions were developed to measure behavioural beliefs of the participants; four related to outcomes in autonomy, another four related to coping and the final three related to self-control. There are many advantages of planning for retirement (Delpachitra & Beal, 2002; Ameriks et al., 2003; Hershey & Jacobs-Lawson, 2009; Street & Desai, 2011; Lusardi & Mitchell, 2011a), but it seems the most accessible ones for this group of participants are concerns with outcomes related to autonomy, coping and self-control. Using the modal beliefs generated, items addressing the behavioural belief are developed. Details are provided in Appendix L. These beliefs will be subjected to validity and reliability tests.

(c) Normative beliefs

The next set of questions intend to inquire into the social environment and its influences on the participants. Results reveal that most people closest to the participants influence their retirement and financial planning activities. Foremost are family members such as spouse, children, parents, siblings and relatives. Friends are also a rich source of social support. Some participants also mentioned that they had been approached by individuals or groups from the financial industry regarding their financial and retirement plans. However, many have trust issues with them.

During the interviews, the participants elaborated on the influences of these important referent individuals or groups. They provided information on financial matters and social support when required. Married participants talked about working together with spouses on critical financial matters, such as buying insurance or properties. Others mentioned that some spouses provided an alternative source of income, when necessary. Many also share financial expenses at home. Some participants report that they are the active financial planner in the household, while others left the role to their spouses.

Influences from immediate family members and relatives are also quite dominant. Parents constantly provide information and advice for the future. For example, they emphasise the importance of having a stable job and a steady stream of income to cater for life challenges. Similarly, many participants purchase insurance and make provisions for their children in order to reduce the risk of possible financial distress later in life. Some participants also report that siblings would share financial information such as investments and savings. Other participants also reported the role of relatives in providing information
(such as possible good and reputable investments) and educating them on financial management skills. Nonetheless, participants also lament that sometimes the different opinions and viewpoint confuses them and challenge their decision-making skills. Nonetheless, many admit that relatives are a rich resource of advice and support.

Similarly, friends provide support to the participants. They provide information on various types of investments that they have done (especially when the returns from the investments are lucrative). Some also share information on their concerns for the family. They purchase insurance plans or make specific savings for the future.

In order to develop the items, important individuals or groups are categorised as family members and non-family members. For family members, the spouse is the most frequently cited important person, followed by children, parents, siblings and relatives. Hence, the groups are divided into spouse, members of immediate family (which comprise of siblings, children and parents). As for family members other than spouse, parents, children and siblings, they are categorised as relatives. Non-family members are categorised as friends and acquaintances.

Most participants have not encountered parties that would disapprove of them planning financially for retirement. For those who have encounter disapproval, they are acquaintances, which have not initiated any planning activities for their future. They do not have an adequate understanding of finances or planning, or are afraid of the risk of losses should they make any investments.

The items developed are based on the individuals or groups mentioned by the participants. However, in reviewing the items developed, the panel reviewers are also of the opinion that an additional scale item "not applicable" should be added in the actual survey. This is to cater for the different social environments. For example, for unmarried

people, normative influences from a spouse would not be applicable. This is also in line with recommendations from Fishbein and Ajzen (2010).

The various categories of individuals and groups are based on the different social environment of the preliminary interview participants. As noted by the panel reviewers, it may not be all-encompassing. In a wider survey, respondents may be influenced by other individuals or groups not within the category identified. In order to address these other important referent persons or groups, a section will be included for respondents to complete, should there be these group of people or entity.

Normative Beliefs	Participants	Items developed
Proximal individuals/groups that are		Categories of important individual /
related to participants:		groups
Family	10	
Spouse	11	• My spouse
Children	9	• My immediate family
Parents	9	(example: siblings, parents, children)
Siblings	2	• My relatives
Relatives	3	(family members other that spouse,
		parents, children, siblings)
Proximal individuals/groups (but not		
related to participants):		
Friends	8	• My friends
Acquaintances		My acquaintances

 Table 3.6: Normative beliefs identified from preliminary interviews

From the interviews, participants also highlighted that they are influenced by what these important referents do. These are similar to descriptive normative beliefs. Descriptive normative beliefs refer to one's perception of what important others do (Cialdini et al., 2006; Cialdini, 2007). Participants highlighted that some of the important referents in their lives are also actively planning for their retirement. These referents have initiated strategies to accumulate wealth. Some of their family members and friends have saved their excess income and are leading comfortable lifestyles. Participants also reported that their siblings had grown their wealth by investing in good and reputable investment. They also receive a steady stream of income from these investments. On the non-financial side, some participants also noted that they admire referents which have continuously taken care of their health to reduce the risk of any serious illnesses.

Given their experience and expertise, many participants have sought help from these referent persons when they initiate their planning activities. Participants highlighted that their referents had helped them to build the competence and confidence to explore and build their planning strategies independently.

In conclusion, these interviews have identified important referents in the participant lives. They are not only concerned about whether what they do would be approve by these referents (injunctive normative beliefs), but also by what the referents are doing themselves to plan for their retirement future (descriptive normative beliefs).

(d) Control beliefs

Control beliefs are defined as the perceived presence of factors that ease and impede performance. Most participants reported challenges that impede their retirement planning and financial activities. The first area of concern pertains the reliability of the channels of information on financial matters. Participants reported receiving information from formal and informal channels. For formal channels, this includes mass media information such as newspaper, magazines, internet news, television and radio. Informal channel comes from family, relatives, friends and financial services providers. Some of the information is conflicting, and confusing. Participants also highlighted that they sometimes doubt the authenticity of the information provided. Given the variety of information, viewpoints and products received, participants find making planning, financial and retirement decisions stressful. They have to make the decisions in the current environment, but they will only know the effect or impact of their decisions many years into the future. It is difficult to envision what will happen then.

Also, many doubt their ability to plan. Participants acknowledge their lack of skills to understand and use financial information and opportunities that are available. Some also highlighted that time management is required to make their plans a success. Four modal beliefs are identified from the interviews, namely difficulties "to think about my future plans as am not sure if the information received is reliable", "to make decisions in view of the differing viewpoints and products received", "to learn the skills of financial management" and "to be able to have skills to manage time required for planning".

The four beliefs identified are linked to competence. Competence refers to the tools, equipment, training and supports a person have in order to perform to a certain standard. It encompasses a sense of feeling able and confident from a job that is satisfactorily completed (Deci & Ryan, 2000). In the case of financial planning for retirement, financial competence is required to plan for retirement. In total, 76% of the participants have reservations in their competence to plan financially for retirement.

Another area of concern is the adequacy of their income and financial resources to meet daily needs. Managing daily expenses, handling unforeseen increases in expenditure is hard, in view that resources are limited. Some cite that inadequate income not only makes it difficult for them to plan; there is also no excess income to save for long-term requirements, such as retirement. They realise that a steady stream of income is important. However, they have concerns in generating the income streams require, especially when faced with limited career opportunities and stagnated increments. Inadequate income also raises concerns when managing expenses. Increasing expenses and commitment are grave concerns. Some are worried that they would be unable to support the financial requirements of their dependents, let alone plan for their retirement requirements. Many are aware that serious lifestyle changes are required, but even this is a challenge. Participants also find it difficult to take advantage of all the environmental opportunities such as financial and retirement benefits that can help them in their planning effort. Five modal control beliefs are identified from the interviews, that is "to manage expenses and commitments when planning financially for retirement", "to have the excess income to plan financially for retirement", "to have the stable source of income which is important when planning financially for retirement", "to be aware of all the financial and retirement benefits that is important to planning efforts" and "to be flexible with the lifestyle changes required in financial plans for retirement". All these beliefs concern the area of managing resources. It is also a problem cited by 90% of the participants. In summary, grave concerns on the inadequacy of their income were highlighted. A steady stream of income is greatly desired, when planning. Unfortunately, the situation is made worse by increasing expenses and rising cost of living.

Another aspect that is of concern relates to the areas of risk. Two areas feature prominently in the discussion. One is the financial risk borne when participants make investments choices. Another concerns a form of social risk, that is, life course risk. Life course risk implies that social risk is distributed unequally across the life course of a person. This is especially for those in childhood and old age. Retired and ageing individuals face higher life course risk such as poverty, social isolation, loss of independence and autonomy as their earnings and health decline during the lifespan. About 40% of the participants are concern about these risks.

With regard to financial risks, participants are unsure whether they have made good investment decisions and worry whether it will generate the desired income streams in the future. Participants also acknowledges the risk of aging, with its possible deteriorating physical and mental conditions. Such risks are hard to manage, and it is inevitable. To counter these risks, many have bought insurance policies and adopt healthy lifestyles. Additional three model control beliefs are identified; "to evaluate areas which may bring financial risk to planning effort", "to be confident of what will happen to investments in the future" and "to be certain of the future situation". Details are provided in Appendix L.

3.1.3 Additional Information

Several additional questions are included in the survey instrument to provide a better understanding of the respondents' background. This information will be used in the descriptive analysis. A summary of the codes and categories developed is available in Appendix B.

(a) Demographic Information

Demographic information such as age, education level, nationality, gender, ethnicity, religion, marital status and details of occupation were asked. These questions are available in Section E of the survey instrument developed. For data entry and analysis purposes, each demographic item is coded.

For the demographic variable age, ten categories were identified, ranging from below 20 to above 60. This coded as AGE. As for the education background, six categories are identified. They are SPM or Certificate Level, Diploma, Bachelor Degree, Master, PhD and Others. Education background is coded as EDU1. Gender consists of two categories, Male or Female, and is coded as GENDER1. Four categories are identified for race or

ethnicity, which are Malay, Chinese, Indian and Others. This variable is coded as ETH1. The major religions in Malaysia are identified, which are Muslim, Buddhist, Hindu and Christian, with an additional 'Others' category. Religion is coded as RELIGION1. Marital status comprises four categories, namely Never Married, Married, Widowed and Divorced/ Separated, and coded as MARITAL1. To ensure that only the targeted population complete the survey, the variable nationality is identified. Nationality consists of two categories, Malaysian or Others, and is coded as NATION1. Only surveys completed by Malaysian university employees will be analysed.

University employees are wage earners, and a significant proportion of their income comes from their occupation. Further information is also obtained to better understand the individual's occupation and position in their organisation. Information on participants' occupation is available in Section E of the survey instrument developed.

In this section, respondents are to provide information on their occupation. It is in a free format which allows respondents to specify their position. This item is coded as OCC1_TYPE. The responses were analysed, and four categories are identified. They are academics, which comprise of lecturers, researchers and teaching staff, and non-academics. Non-academic are further classified into top management, management & professional and support staff.

Further inquiries were made concerning the status of their position, that is, whether they are permanent, contractual or have other arrangement with the organisation. This is an important aspect to determine the stability of their careers. This item is coded as OCC2. Other information collected pertains their organisation, whether it a public sector organisation funded by the government or in the private sector (relies on corporate investment, alumni, and/or student funding). An 'others' category is also developed for respondents in organisation that are neither public or private funded. This item is coded as

OCC3. Further information regarding the retirement schemes that respondents have contributed to were also obtained. In this regards, three categories were identified, that is, government pension, contribution to Employee Provident Fund (EPF) and other contribution schemes they are involved in. This item was coded as OCC4.

(b) Information on financial matters

In order to plan, monetary and income resources are important. Information on financial matters are asked in Section B and D of the survey instrument developed. These questions would provide an overview of the respondents' ability and resources to plan. The first question seeks to inquire on the estimated annual personal income. A range from less than RM30,000 to above RM150,000 was provided. Eight categories are identified, that is, less than RM30,000, RM30,000-50,000, RM50,001-70,000, RM70,001-90,000, RM90,001-110,000, RM110,001-130,000, RM130,001-150,000 and above RM150,000. This item is coded as IC1.

Other questions to capture respondents' monthly financial commitments and debt level were also collected. Respondents were asked about the monthly household debt repayments to financial institutions. A range from no debt to above RM10,000 was provided for respondents to select. Eight categories are identified, that is, no debt, less than RM1,000, RM1,001-3,000, RM3,001-5,000, RM5,001-7,000, RM7,001-9,000, RM9,001-10,000 and above RM10,000. This item is coded as IC2.

Questions to gauge the satisfaction level of the respondents with their current financial situation were also asked. The categories developed are not at all satisfied, slightly satisfied, moderately satisfied and very satisfied.

At the same time, for respondents with children, additional information on their dependents were solicited. Information on children is asked in Section E of the survey. This

includes information such as the number of dependent children (CHILD1), their ages (CHILD2) and the number of children they have which are still financially dependent on them (CHILD3). All these questions are in a free format. For number of dependent children, categories ranging from 0-6, 8 and 9 are identified. Six ages of children can be identified, children below 7, 7-12, 13-17, 18-21, 22-25 and above 25 years. As for the number of children who are still financially dependent on the respondent, seven categories are identified ranging from 0-6 and 9.

3.2 Goodness of Measures

Several methodological considerations are taken to ensure the validity and reliability of the survey instrument developed. These are important considerations before actual data collection activities begun.

A valid survey instrument would mean that the findings are really about what they appear to be, while reliability would reflect that the instrument would yield consistent findings (Saunders, Lewis & Thornhill, 2009). Sekaran (2003); Sekaran and Bougie (2010) explain this as assessing the 'goodness' of the measures developed. The instruments used must measure the variables as they are supposed to, and in an accurate manner.

3.2.1 Expert Validation

To ensure the validity of the survey, a team of experts is recruited. This team would be assessing each item in the survey instrument (Lynn, 1986). Details results of the evaluation are available in Appendix C to Appendix I.

For this study, six experts in fields related to this study have agreed to assess the items developed (Davis, 1992; Grant & Davis, 1997; Sekaran & Bougie, 2010). These experts are all PhD. holders, with a proven track record, experience and publications in refereed journals from international and well-known institutions (Grant & Davis, 1997). They are recruited on the basis that they have the knowledge and experience in areas relating to the theory (TPB) and are experts in the field of financial planning, retirement and aging.

Invitations were sent to the various experts (Waltz, Strickland & Lenz, 1991). Details of the research were provided and background information, such as objectives, population and context of the study that are deemed important in judging the elements of the questionnaire, are included. Details regarding the conceptual basis for the instrument, such as its underpinnings, research questions and the theoretical model, are also provided (Polit & Beck, 2006).

Each item in the constructs is presented in a tabular form. The experts are to evaluate each item in the questionnaire based on the consistency, representativeness, relevance and clarity using a 4-point rating scale (1 being 'not relevant or representative'; 2 is 'somewhat relevant or representative'; 3 is 'quite relevant or representative'; 4 is 'highly relevant or representative').

Consistency pertains consistency of the items with the conceptual definitions. Representativeness, in turn, reflects the suitability to the domain of interest. Meanwhile, relevance refers to relevance to interpretations. For clarity, it denotes a comprehensive understanding of each item. The experts were also free to make suggestions or modifications for the adequacy of each item, as well as for the addition or exclusion of items. Upon completion, the discussion is initiated with the panel members to explore and clarify any instrument issues. The experts' ratings are further analysed using two measures. The first is the kappa coefficient, which measures the level of interrater agreement (Waltz et al., 1991; Polit & Beck, 2006). The threshold of acceptable levels for interrater agreement is between 0.70 to 1.00 (Davis, 1992; Polit & Beck, 2006). The second measure used is the Content Validity Index (CVI) (Davis, 1992; Polit & Beck, 2006; Natalio, Faria, Teixeira-Salmela & Michaelsen, 2014). CVI measure the proportions of experts who have scored the items as relevant, that is either a 3 or 4. Therefore, for this study, as there are six experts, at least five of the experts must give a score of 3 or 4 in order to achieve a minimum CVI value of 0.833. This would mean that the acceptable CVI value should be between 0.833 to 1.00 (Davis, 1992; Polit & Beck, 2006; Natalio et al., 2014).

Items below the acceptable rate for kappa and CVI are investigated further. Discussions is initiated with the experts and issues clarified. This iterative process assures the consistent use of definitions and scaling options, and also the consistency, representativeness, relevance and clarity of the items in the survey instrument developed.

3.2.2 Validation for Behavioural Beliefs

From the preliminary interviews, a set of four items related to autonomy were identified. Of these four items, only three were within acceptable levels and coded. For the first item "If I plan financially for retirement, I will be able to envision my financial situation in the future", four of the experts mentioned in their comments that the item could be accepted if rephrase for better clarity. The item is promptly rephrased and approved by the experts. Only one item ('If I plan financially for retirement, I would be more aware of the financial benefits from the action I take') is removed as the kappa and CVI scores were below the threshold set for consistency, representation, relevance and clarity. The remaining items are 'If I plan financially for retirement, I will be able to envision my financial situation in the future' coded as BA1; 'If I plan financially for retirement, I will be able to set my directions (e.g. have more flexibility and choices in life)' coded as BA2 and 'By planning for my financial needs, I will be independent and self-sufficient' coded as BA3.

The second set of items deals with the belief that planning helps to cope with life's stressors. Of the four items outlined, three achieved the acceptable kappa and CVI levels. The item 'Planning financially for retirement will help me to be prepared in handling potential problems, i.e. financial problems' had achieved low kappa and CVI scores for the relevance criteria. Four experts further commented that this item is similar to the item 'Planning will help me face uncertainties more confidently' (BC1) and recommend that BC1 be retained with examples added. After the amendments, the remaining items are 'Planning will help me face uncertainties more confidently (example of uncertainties: healthcare problems, sudden loss in income, increase in financial commitments)', coded as BC1, 'By planning, I will be able to have adequate financial support during the remaining years of my life', coded as BC2 and 'If I plan, I will be able to maintain my quality during retirement', coded as BC3.

Three items are developed based on the beliefs related to self-control. These items achieved acceptable kappa and CVI scores. The items are validated and coded as BSC1 ('Planning will help disciplined me to resist unnecessary spending'), BSC2 ('Even though saving for retirement limits the money I have currently, the money I have saved and invested will benefit me after I leave the workforce') and BSC3 ('Planning financially for retirement is a burden - it prevents me from enjoying my life now') respectively.

In summary, of the 11 items developed, two were removed. Details of the items are available in Appendix C. Items developed are included in Section B of the final survey.

3.2.3 Validation for Normative Beliefs

The preliminary interviews identified several important persons who are categorised as spouse, immediate family members (such as siblings, parents, children), friends; relatives (such as family members other than spouse, parents, children, siblings) and acquaintances. These groups play a role in influencing individuals directly, either through their expectations (injunctive normative beliefs) or their actions (descriptive normative beliefs).

A total of five items were developed for these normative beliefs. The first four categories developed is within acceptable levels. For injunctive normative beliefs, these items are coded as INB1 (spouse), INB2 (immediate family), INB3 (friends) and INB4 (relatives) whereas for descriptive normative beliefs these items are coded as DNB1 (spouse), DNB2 (immediate family), DNB3 (friends) and DNB4 (relatives). Only one category is removed (acquaintances) as it received low ratings in the representativeness and relevance criteria. Details of the items are available in Appendix B. Items developed are included in Section C of the final survey.

3.2.4 Validation for Control Beliefs

From the preliminary interviews, several factors were identified to influence control beliefs. One of them is in reference to competence. Four items are developed to measure control beliefs. Of this, only three achieved the acceptable kappa and CVI levels. The item "It is difficult to have the skills to manage time, which is required when planning financially for retirement" did not fulfil the consistency and representative criteria of the experts and was removed. The remaining items 'It is difficult to think about my plans as I am not sure if the information I received is reliable' were coded as CC1, 'The differing viewpoints and financial products I receive makes it hard for me to make planning

decisions' were coded as CC2 and 'It takes a lot of effort to learn the skills of financial management' were coded as CC3.

As for managing resources, all five items developed have achieved acceptable kappa and CVI levels. The items 'I find it difficult to manage my expenses and commitments when planning' is coded as CR1, 'It is hard to find the extra income to put aside for retirement purposes' are coded as CR2, 'It is difficult to have a stable source of income required in a planning effort' are coded as CR3, 'It is difficult to be aware of all the financial and retirement benefits that are important to my planning efforts' are coded as CR4 and 'It is hard to be flexible with the lifestyle changes required in financial retirement plans' are coded as CR5.

Three items are developed in reference to risk concerns. All items are within acceptable kappa and CVI levels. The item 'It is hard to evaluate areas which may bring financial risk to my planning effort' are coded as CRS1, 'When I am planning financially for retirement, I am confident of what will happen to my investments in the future' are coded as CRS2 and 'I find planning for retirement very difficult as I am uncertain of my future health situation' are coded as CRS3.

In summary, a total of 12 items were developed. Of that, only one item was removed. Details of the items are available in Appendix E. Items developed are included in Section B of the final survey.

3.2.5 Validation for Items Adapted

Four constructs from the theoretical framework had adapted items from the various literature. These items were also provided to the experts for content validation. The first set of items measure attitudes. These items are measured using semantic differential scales

of 'good-bad', 'enjoyable-uninteresting', 'useful-worthless', 'beneficial-harmful' and 'pleasant-unpleasant'. All the items are within acceptable kappa and CVI levels. The item 'good-bad' is coded as DA1, 'enjoyable-uninteresting' coded as DA2, 'useful-worthless' coded as DA3, 'beneficial-harmful' coded as DA4 and 'pleasant-unpleasant' coded as DA5. Details of the results of the validation are available in Appendix F. Items adapted are included in Section A of the final survey.

To measure injunctive norms, two items are adapted. These items are 'I feel the social pressure to plan financially for retirement' and 'Most people who are important to me think I should plan financially for retirement'. Both items are within acceptable kappa and CVI levels and coded as IN1 and IN2 respectively. Details of the results of the validation are available in Appendix G. A measure was also adapted to capture descriptive norms that are, "Most people who are important to me plan financially for retirement" and is coded as IN1. These items are included in Section C of the final survey.

Four items are adapted to perceived behavioural control. The items, 'I am confident that I can effectively plan financially for retirement' (DPBC1) and "For me to plan financially for retirement is easy" (DPBC2) is adapted to reflect the confidence or self-efficacy of a person. Subsequently, the items, "The decision to plan financially for retirement is beyond my control" (DPBC3) and "Whether I plan financially for retirement or not is entirely up to me" (DPBC4) is adapted to measure controllability. All the items are within acceptable CVI and kappa levels. Details of the results of the validation are available in Appendix H. Items adapted are included in Section B of the final survey.

Intentions of the respondents are measured using three items. The first item is 'I *want* to plan financially for retirement in the next 12 months' (coded as DINT1), while the second item 'I intend to plan financially for retirement in the next 12 months' (DINT2) and the third item is 'I will try to plan'(DINT3). All the items are within acceptable kappa and

CVI levels. Details of the results of the validation are available in Appendix I. Items adapted are included in Section D of the final survey.

In summary, a total of 48 items were developed and adapted for the survey instrument. They comprise of 33 items developed from interviews with participants and 15 items adapted from various literature. Upon completion of the expert validation process, five of the items developed from interviews are removed. A total of 43 items remains, in which 28 are developed, and 15 are adapted. The following table is a summary of the results of the validation by experts.

Variable	Number of Items		Remove
	Initial	Final	
Items adapted		Þ.	
Attitude	5	5	-
Subjective norms	3	3	-
- Injunctive norms (2)			
- Descriptive norms (1)			
Perceived behavioural control	4	4	-
Intentions	3	3	-
Total items adapted	15	15	-
Items developed			
Autonomy	4	3	1
Coping	4	3	1
Self-control	3	3	-
Total Behavioural beliefs	11	9	2
Injunctive normative beliefs	5	4	1
Descriptive normative beliefs	5	4	1
Total Normative beliefs	10	8	2
Competence	4	3	1
Resources	5	5	-
Risks	3	3	-
Total Control beliefs	12	11	1
Total items developed	33	28	5
TOTAL (items adapted and developed)	48	43	5

 Table 3.7: Summary of Content Validation by experts

3.3 Pilot Studies

A pilot study is conducted before implementing the full study (Cooper & Schindler, 2011). A final check is made on the survey instrument before conducting the pilot test. This involve checking the words and sequence of questions. The words have to be easy to read, short and simple. The content of each item should not have two simultaneous questions. As for the sequence of questions, it should be clear and organised to minimise eyestrain, with clear instructions to the reader. All this will help to avoid missing values and unreliable answers.

The purpose of the pilot study is to test the research methodology, survey instruments, data collection techniques and other research techniques in preparation for a more comprehensive study (Hassan et al., 2006; In, 2017). The survey instrument has to be comprehensible and appropriate for the target population. It is important to ascertain that respondents are interpreting the question or statements as intended. Readability and clearness of words have to be established. The respondents' viewpoints are important in order to avoid possible misinterpretations, falsified answers and missing responses. Also, the length of time taken to complete the entire survey instrument is considered (Bowden, Fox-Rushby, Nyandieka & Wanjau, 2002).

The pilot test considers the various aspects of data collection. This would include entering and coding of survey items and appropriate statistical tests. This would further ascertain the goodness of the measures and to identify any potential problem areas and deficiencies that may arise.

Invitations are sent and follow-ups made via telephone calls to explain the various aspects of the pilot test and to solicit participation. Those who agree to participate are given the survey instrument to complete. Additional comments on the items, structure and interpretations of the questionnaire were sought. This is to learn whether they had difficulty

with any items and to receive other suggestions to improve the quality and readability of the instrument.

A total of 35 public and private universities employees responded to the pilot test. The respondents provided useful comments on the clarity of the items in the survey instruments. This includes the terms used, the style of language (ambiguity, readability, comprehensibility) and the flow of the statements used. Other aspects included suggested improvements to the cover letter, and the format of the instrument used (including the font and layout). Respondents also suggested that some examples be provided for better clarity.

The responses were entered into the Statistical Package for Social Sciences (SPSS) programme, using the codes prepared for each item. Preliminary descriptive analysis and reliability analysis were also performed.

The result demonstrated that the research is feasible. It did not appear to be disruptive to the respondents. Also, flaws identified in the survey instruments could be amended before implementing the full study. Data entry was not problematic.

3.3.1 Results of the Pilot Study

Descriptive analysis was conducted for this pilot study. The demographic information analysed are age, education background, gender, race or ethnicity, religion and marital status. The respondents must be Malaysian citizens currently working in a public or private university. Details of the results of the pilot study is available in Appendix K.

The respondents are from four categories of age groups, that is, 25-29 (17.1%), 30-34 (22.9%), 35-39 (31.4%), 40-44 (17.1%) and 45-49 (11.4%). A majority are of 35 to 39 years old while those from ages 45 to 49 had the least number of respondents.

As for the educational background, the respondents are ordered into three categories; Bachelor degree (11.4%), Master (71.4%) and PhD (17.1%). Most of the respondents have a Masters qualification. As for gender, 77% of the respondents are female, while the rest are male.

For ethnicity, most of the major races are represented. About 66% are Malay, 26% are Chinese and 6% from others. Only one respondent (2%) did not respond to the question. For religion, it is somewhat reflective of the ethnic diversity of Malaysia, which is Muslim (69%), Buddhist (23%) and Christian (6%). Similarly, only one respondent (2%) did not respond to the question. As for marital status, about 94% of the respondents are married while the rest are either widowed, divorced or separated.

An analysis of occupational details of the pilot study respondents reveals that most of them are academics (80%), namely lecturers, teaching staff and researchers. 20% are from the non-academic section consisting of management and professional (17%) and support staff (3%). Most of the staff have permanent positions (80%), while only 20% are contractual. Meanwhile, for type of organisation, most of the respondents are from the public sector universities (80%). Only about 20% are from private universities. As for the retirement contribution schemes, 80% choose to receive the government pension.

On the financial information, respondents revealed that they have annual personal income that ranges from less than RM30,000 to RM150,000. Responses come from seven categories, that is, less than RM30,000 (17.1%), RM30,000-50,000 (8.6%), RM50,001-70,000 (28.6%), RM70,001-90,000 (22.9%), RM90,001-110,000 (11.4%), RM110,001-130,000 (8.6%) and RM130,001-150,000 (2.9%). A major proportion are from the income group of RM50,001-70,000. Only one respondent is from the income range RM130,001-150,000.

110

As for the estimated total monthly household debt repayments made to financial institutions, seven categories are identified. They are from categories of less than RM30,000 (17.1%), RM30,000-50,000 (8.6%), RM50,001-70,000 (28.6%), RM70,001-90,000 (22.9%), RM90,001-110,000 (11.4%), RM110,001-130,000 (8.6%) and RM130,001-150,000 (2.9%). Most of the respondents pay debts of about RM1,001-3,000 per month. Only one person reported paying RM9,001-10,000 while another reported paying above RM10,000.

The respondents were also asked of their satisfaction with their current financial situation. Four categories of respondents were identified, not at all satisfied (11.4%), slightly satisfied (42.9%), moderately satisfied (31.4%) and very satisfied (14.3%). Most respondents are only slightly satisfied with their current financial situation (15 respondents). Four respondents are not at all satisfied with their situation.

For respondents with children, further information was solicited about their dependents. Of the respondents, only 22 (63%) have children. 29% have one child, 2.9% have two children, 22.9% have three children, 5.7% have five children and 2.9% have six children. Most respondents have at least one child. One person reported having six children (3%).

Further analysis revealed that the children consists of three categories, namely below 7 years, primary school going children (7-12 years), secondary school going children (13-17 years) and pre-university (18-21 years). For those with children below 7 years, 28.6% report having one child, 11.4% report having two children and 5.7% report having three children. For children between 7-12 years, 17.1% report having one child, 14.3% report having two children and 2.9% report having three children. As for children between 13-17 years, 2.9% reported having one child while 8.6% have three children of the age range. Only one respondent (2.9%) reported having two children between 18-21 years. From the analysis, most of the respondents have children below 12 years old.

111

Most of the respondents have children who are still financially dependent on them. This is a burden to the financial resources available for planning for the respondents. From the table below, most of the respondents (49%) have at least 1-3 dependent children.

In addition to the demographic analysis, a reliability analysis was conducted for this pilot study (details available in Appendix K). Several statistics are reviewed. Firstly, the correlation matrix is checked for negative values. All values should be positive, as this would indicate that the items are measuring the same underlying characteristic.

Subsequently the Cronbach's alpha value are reviewed. Cronbach's alpha measures internal consistency which indicates the relationship of the items in the construct (Sekaran & Bougie, 2010). Measures of Cronbach's alpha are between 0 to 1. Values closer to one indicates a higher internal consistency reliability. Generally, reliability of less than 0.60 are poor, those within 0.60 to 0.80 are acceptable while those above 0.80 are deem good (Nunnally, 1978; Pallant, 2016).

Next, the Item-Total Statistics table are reviewed. The Corrected Item-Total Correlation values indicate the degree in which each item correlates with the total score. Items measuring the same construct should be above 0.30. In addition, the Cronbach's Alpha if Item Deleted should be below the final Cronbach's Alpha value.

For the construct behavioural beliefs, all the items in the correlation matrix have positive values. The Corrected Item-Total Correlation values are above 0.30, except for BSC3_r. Nonetheless, the Cronbach's alpha is above 0.80 ($\alpha = 0.902$) which is deemed good and hence not items are required to be removed.

The scales which measure normative beliefs have a 'not applicable' option. These responses are treated as missing values (using listwise deletion). Hence, only about 69% are valid cases. Most of the items in the correlation matrix have positive values and

'Corrected Item-Total Correlation' above 0.30, except DNB1. However, the Cronbach's alpha is above 0.80 ($\alpha = 0.826$) which is deemed good. As the Cronbach's alpha is good, no items are required for removal.

For the construct control beliefs, all the items in the correlation matrix have positive values. In addition, the values in the 'Corrected Item-Total Correlation' are above 0.30, except for CRS2. However, the Cronbach's alpha is above 0.80 ($\alpha = 0.920$) which is deemed good; hence no items are removed.

The items in the construct attitudes did not have reliability ($\alpha = -1.332$). Items DA3 and DA5 have negative values in the correlation matrix and the Item-Total Statistics table. This is a reason for the negative Cronbach's Alpha value.

These two items were removed from the scales and reliability analysis was conducted again. The Cronbach's alpha improved to above 0.80 ($\alpha = 0.900$), which is deemed good. All the items in the correlation matrix have positive values, with values in the Corrected Item-Total Correlation above 0.30.

For the construct subjective norms, all the items in the correlation matrix have positive values. The Cronbach's alpha is below 0.60 ($\alpha = 0.515$) which is deemed low. In addition, IN1 has 'Corrected Item-Total Correlation' below 0.30. Removing this item would improve the Cronbach's Alpha value to acceptable level ($\alpha = 0.669$).

For the construct intentions, all the items in the Correlation Matrix have positive values. The Cronbach's alpha is above 0.80 ($\alpha = 0.979$), which is good. All 'Corrected Item-Total Correlation' are above 0.30. No items are removed.

A summary of the results of the reliability analysis are presented in the following table. Details are available in Appendix K.

Variables	No. of	Cronbach	Items	No. of items	Cronbach alpha
	items	alpha	Recommended	(after removal)	(after removal)
			for removal		
Attitudes	5	-1.332	DA3, DA5	3	0.900
Subjective norms	3	0.515	IN1	2	0.669
PBC	4	0.529	DPBC3_r	3	0.629
Intention	3	0.979	-	3	0.979
Behavioural beliefs	9	0.902	-	9	0.902
Normative beliefs	8	0.826	-	8	0.826
Control beliefs	11	0.920	-	11	0.920
Total:	43		Total:	39	

 Table 3.8: Results of reliability analysis (pilot study)

From the reliability analysis, all the belief constructs (behavioural beliefs, normative beliefs, control beliefs) have Cronbach's alpha above 0.80, which is deemed good. Control beliefs have the highest level of reliability ($\alpha = 0.920$), follow by behavioural beliefs ($\alpha = 0.902$) and Normative beliefs ($\alpha = 0.826$). Hence, no items are recommended for removal.

However, three constructs, attitudes ($\alpha = -1.332$), subjective norms ($\alpha = 0.515$) and PBC ($\alpha = 0.529$) have low reliability levels. Four items are recommended for removal to improve the reliability. This include two items from the construct attitudes (DA3, DA5), one item from subjective norms (IN1) and PBC (DPBC3_r), respectively. With the removal of these items, the Cronbach's alpha improved.

Generally, the results indicate that reliability is acceptable for most of the constructs used in the study (with recommendations for removal of selected items). However, this is based on a sample of 35 respondents. Also, a high or acceptable rate for Cronbach's alpha does not indicate dimensionality. As such, further analysis is conducted before removing any items.

3.4 Ethics Approval

Before conducting data collection for the main survey, approval was sought from the Ethics Committee, University of Malaya. In line with the ethical requirements, the purpose, procedures and nature of the project were reviewed. All aspects of the study conform to the requirements, and there were no potential risks.

The researcher undertook steps to ensure the confidentiality of the universities selected in the sampling procedure. Also, personal information was not revealed. The data collected is also not used for any other purpose.

An invitation letter, the participant information and instructions sheet were enclosed with the survey instrument to explain and obtain permission to conduct the survey. The information describes the purpose of the study and the identity of those who are conducting the research. Participation is voluntary, and confidentiality is assured. Individuals are also informed that the right to abstain from participating in the research would be respected. The documents also clearly specify the contact details of the researcher. All the information is in order, and the University's Ethics Committee duly approves the project.

3.5 Population and Sampling

Sekaran and Bougie (2010) emphasised that selecting a representative sample and having sufficient responses are important. For this purpose, the target population and sampling method must be clear. In this study, the targeted population are Malaysians employees, currently working in universities.

A sampling frame of all the universities in Malaysia is compiled (Malaysian Qualifying Agency, 2009; StudyMalaysia.com, 2015). From the list, two groups are identified. The first group consists of public universities which are funded by the government. There are

two forms of retirement schemes offered to this group, namely the pension and EPF schemes. The second group consists of private universities where the respective institutions are mandated to contribute to EPF. Details of both groups are available in Appendix P.

As for sampling strategy, this study adopts a simple random sampling process. The respondents are selected based on the list of universities. To do this, all the names and details of the universities are entered into SPSS. Using the random number generator option, five universities are selected from each group. Invitations, together with information regarding the study, were sent to these universities for permission to conduct data collection. Universities that declined to participate were then replaced by other universities, also selected randomly from the respective groups. Upon approval by the respective universities, the survey form and related information were sent through the staff e-mails via the university portal. This will ensure that all university employees have an equal chance of participating in the survey.

In deciding the appropriate sample size, several considerations were made. The total workforce in the higher education industry in Malaysia is estimated to be around 100,000 (Ministry of Education Malaysia, 2014a; Ministry of Education Malaysia, 2014b; Department of Statistics Malaysia, 2016a). From the literature on sample size, several researchers have recommended a sample size of 500 (Roscoe, 1975; Sekaran & Bougie, 2010). Similarly, Krejcie and Morgan (1970); Cohen (1988) suggested that a sample size of 384 is appropriate for a population of 1,000,000 and above. Roscoe (1975) rule of thumb advises that the minimum sample should be at least ten times the number of variables (Sekaran & Bougie, 2010). After considering all this, a sample size of about 500 is collected as it was large enough to accommodate issues such as measurement error (with a confidence level of about 95%). Nevertheless, attempts will be made to collect more to ensure that the rigour for this study is ascertained.

3.6 Data collection

Unfortunately, the data collection strategy discussed in the previous section generated a very low response rate. Only 30 completed surveys were received from the various universities. Additional measures via telephone and emails were initiated to increase the responses. These measures were unsuccessful in generating the additional responses. Discussions were subsequently conducted with the university administrators to increase the response rate. With their approval, hardcopies of the survey forms are distributed according to the terms and conditions specified (selected locations, within the premises of the universities, during working hours). Indirectly, these additional measures also ensure that only university employees complete the survey.

An additional 2,350 survey forms were distributed through these methods. With the additional efforts, a total of 677 survey forms were returned. Of this, 265 are from various public universities whilst the rest are from private universities.

The surveys received from the various respondents were checked to ensure completeness and readability before coding. A total of 177 respondents have to be excluded. This is because 28 of these respondents are not Malaysians. The remainder (149 respondents) of the survey forms were incomplete where they either had no comments and responses at all or did not complete major proportions of the sections (approximately 25% and more of the items in the questionnaire are incomplete). The excluded cases are removed and not keyed in at all (Sekaran, 2003; Hair et al., 2010).

Upon completion of all the required data checking, the total acceptable surveys are 500, in which 221 are from public universities, and a total of 279 are from private institutions. The following table summarises the responses received.

Details	Public universities	Private universities	Total
Questionnaires distributed	1200	1150	2350
Questionnaires returned	265	412	677
Response rate	22%	36%	29%
Excluded cases			
Non-Malaysians	26	2	28
Questionnaires incomplete	18	131	149
Total excluded cases	44	133	177
Usable questionnaires	221	279	500
Effective response rate	18%	24%	21%

Table 3.9: Summary of survey forms received

3.7 Preliminary Data Analysis

After data have been collected, preliminary analysis is conducted. This initial analysis will help ensure that the data are of assured quality. This involved data screening activities (checking and handling blank responses, coding, determining normality, identifying outliers and collinearity issues), exploratory factor analysis and checking the common method variances or bias. Detail results will be provided in the following chapter.

3.7.1 Data Screening

In surveys, there are possibilities that respondents do not answer all of the items in the survey. The respondents may not understand these questions, do not know the answer, were unwilling to answer or was simply indifferent to the need to respond to the entire questionnaire. Hence, there is a need to analyse the 'missingness' in the data.

Analysis of missing value helps address several concerns caused by incomplete data. If cases with missing values are systematically different from cases without missing values, the results can be misleading. Also, missing data may reduce the precision of calculated statistics because there is less information than originally planned. Another concern is that the assumptions behind many statistical procedures are based on complete cases, and missing values can complicate the theory required.

In order to identify the missing responses, two codes are used during the data entry process. The first code is '99'. This code caters for the blank or missing responses to the various items in the questionnaire. The second code is for the 'non-applicable' responses, which are coded as '999'. Questions, that have 'non-applicable' responses, are only relevant to selected segments of the sample. Using these codes, the pattern of missing data (such as the location of the missing values and how extensive they are) can be identified and analysed.

Once the patterns of missing data are analysed, decisions on how to handle this missing information can be made. Normally, missing values of less than 5% of the total number of cases can be dealt with using several methods. These methods would not have major impacts to the results. These methods are the conventional methods of listwise deletion (or complete case analysis), pairwise deletion and imputation methods. Listwise deletion would mean that the entire case with missing values is omitted. Although listwise deletion often results in a decrease in the sample size available for analysis, it does have advantages. If the data are missing completely at random, it leads to unbiased parameter estimates. Unfortunately, there is a loss in power using this approach (especially if it involves a large number of subjects).

As for pairwise deletion, only specific missing values are omitted (not the entire case). Under this approach each element of the intercorrelation matrix is estimated using all available data. The problem with this approach is that the parameters of the model will be based on different sets of data, with different sample sizes and different standard errors.

119

Other methods to handle missing data include replacing the missing values (imputation). Each missing value is substituted based on a reasonable guess. The analysis is then continued as if there were no missing values. Missing values can be imputed using single imputation methods such as means or linear regressions substitutions. Such actions could lead to an underestimation of standard errors and, thus, overestimation of test statistics.

Statistics have also developed a new approach to handling missing data. This method involves using multiple imputations. An analysis of patterns of missing data is performed with the objective of identifying multiple imputation of missing values. Multiple versions of the dataset are then produced, each containing its own set of imputed values. When statistical analyses are performed, the parameter estimates for all of the imputed datasets are pooled, providing estimates that are generally more accurate than they would be with only one imputation. This method is generally considered to be superior to single imputations. Multiple imputations adopts a simulation-based procedure, where its purpose is not to re-create the individual missing values but to handle missing data to achieve valid statistical inference (Schafer, 1997; Allison, 2001; Soley-Bori, 2013).

After reviewing the pro and cons of each method, the missing values are treated using the pairwise deletion option. This is because there are less than 5% missing values in the data set. Pairwise deletion would allow all available data to be used.

Another important data screening activity is coding. Coding involve activities such as defining and labelling each of the item in the survey and assigning numbers to the possible responses to each question. This also involved reversing and coding negatively worded questions so that all answers are in the same direction. Each item in the survey is labelled with abbreviated names that will be eventually used in the analysis. These names are unique so that the information can be clearly identified. Caution is exercise to keep each name and abbreviation as concise as possible to ensure easy readability. Coding these items and

responses will help in identifying items and sources of error in the data set. A list of the codes and categories developed are available in Appendix B.

The data is also subjected to normality tests. Normality tests are performed to determine if the data set has a normal distribution. For this purpose, the data is viewed graphically and numerically. Graphs such as histogram and Normal Q-Q plots were viewed. In addition, the Kolmogorov-Smirnov and Shapiro-Wilk test of normality were also performed.

Assessment of outliers is important in data screening. Outliers are observations or measures that are distant from other observations. They are extreme case scores which may affect the results of the analysis (Hair et al., 2010). To detect the observations outside the expected range, Mahalanobis distance (D_2) are checked.

The data is also checked for any collinearity issues. Multicollinearity may exist for situation when two (or more) variable are highly related which makes determining the specific contribution of each independent variable in predicting the dependent variable problematic. It is important that collinearity issues are identified before testing the hypothesis of the model (Hair et al., 2010; Sekaran & Bougie, 2010). The indicator used to assess collinearity is the Variance Inflation Factor (VIF) and it indicates the degree to which the standard error has been increased due to the presence of collinearity. VIF value of 5 and higher indicate a potential collinearity problem.

3.7.2 Exploratory Factor Analysis

In developing the survey instrument, a number of items were adapted and developed to measure intentions to plan. Steps such as expert validation and pilot test is taken to assessing validity and reliability of these items. In addition to these steps, an Exploratory Factor Analysis (EFA) is also conducted. EFA is a statistical technique applied to a set of items to uncover coherent subsets that are relatively independent of one another. Subsets of items that are correlated are then combined into factors. These factors reflect underlying processes that have created the correlations among the items. It reflects patterns of correlations, and reduces a large number of items to a smaller number of factors.

An important issue of consideration in EFA is the suitability of the data for factor analysis. It is important that the sample size is large enough for the correlations to be reliably estimated. Tabachnick and Fidell (2013) recommends that the sample size should be at least 300-500 cases. Hair et al. (2010) states that it should be above 100. Many researchers also proposed a minimum of at least ten times as many observations as the number of variables to be analysed (ratio of 10:1) (Nunnally, 1978; Hair et al., 2010; Tabachnick & Fidell, 2013). Undoubtfully, the highest cases-per-variable ratio would mean a more meaningful interpretation of the data. In addition, the correlation matrix should also be reviewed. There should also be a substantial number of correlations greater than 0.30 (Tabachnick & Fidell, 2013). For cases with missing data, the missing values must be dealt with accordingly (either using imputation, listwise or pairwise deletion).

Two additional statistical measures are also used to assess the factorability of the data. The first is Bartlett's test of sphericity (Bartlett, 1954). This measure should be significant (p < 0.05) for the factor analysis to be considered appropriate. The second is the Kaiser-Meyer-Olkin (KMO) measure of sampling adequacy (Kaiser, 1974). KMO has an index that ranges from 0 to 1, with 0.6 suggested as the minimum value for a good factor analysis (Tabachnick & Fidell, 2013).

Another issue to be considered is the number of factors that best describes the underlying relationship among the variables. Adequacy of extraction of factors is related to the number of factors. The more the factors are extracted, the better the fit and the greater the

percentage of variance in the data is explained by the factor solution. However, there is also a trade-off. More factors extracted would also mean less parsimony. Hence, there is a need to find a solution that explain as much of possible the variance in the original data set with as few factors as possible.

A number of techniques can be used to assist in the decision concerning the number of factors to retain. The first is Kaiser's criterion or the eigenvalue rule. Using this rule, only factors with an eigenvalue of 1.0 or more are retained for further investigation. The eigenvalue of a factor represents the amount of the total variance explained by that factor. Another method is the scree test (Cattell, 1966). The scree test plots each of the eigenvalues against the factors. The scree plot is inspected to find a point at which the shape of the curve changes direction and becomes horizontal. It is recommended to retain all factors above the elbow, or break in the plot, as these factors contribute the most to the explanation of the variance in the data set. The final technique is Horn's parallel analysis (Horn, 1965). It involves comparing the size of the eigenvalues with those obtained from a randomly generated data set of the same size. Only those eigenvalues that exceed the corresponding values from the random data set are retained.

In addition, to facilitate factor extraction, numerous procedures are available in SPSS. They are principal components analysis, principal axis factoring (PAF), image factoring, maximum likelihood factoring, alpha factoring, unweighted least squares and generalised least squares. Of these, principal components analysis and PAF are the most commonly used. All these methods have a common objective, that is to determine the smallest number of factors that can be used to best represent the interrelationships among the set of variables. For this study, principal axis factoring (PAF) is used. The following table summarises the goal of analysis for various extraction techniques.

Table 3.10: Summary of Extraction Procedures

Extraction Technique	Goal of Analysis
Principal components	Maximize variance extracted by orthogonal components.
Principal factors	Maximize variance extracted by orthogonal factors.
Image factoring	Provides an empirical factor analysis.
Maximum likelihood factoring	Estimate factor loadings for population that maximize the
	likelihood of sampling the observed correlation matrix.
Alpha factoring	Maximize the generalizability of orthogonal factors.
Unweighted least squares	Minimize squared residual correlations.
Generalized least squares	Weights variables by shared variance before minimizing
	squared residual correlations

Source: Tabachnick and Fidell (2013)

The results of factor extraction must be rotated. Rotation is used to maximise high correlations between factors and items and minimize low ones. It makes the solution more interpretable without changing the underlying mathematical properties. Two types of rotation are available, orthogonal and oblique.

Orthogonal rotation assumes that all factors are independent. There are several orthogonal rotation methods such as varimax, quartimax, and equamax. Varimax is the most commonly used. Orthogonal rotation methods produce a loading matrix. The sizes of these loadings reflect the extent of the correlations between observed items and factors. As there is only one loading matrix to interpret, it makes the solutions easier to report; but the assumptions that the underlying constructs are not correlated seems unrealistic (Tabachnick & Fidell, 2013).

Oblique rotation assumes that the factors are correlated. Two oblique rotation techniques are available; direct oblimin and promax. Direct oblimin is the more commonly used technique. This type of rotation produces several matrices. The loading matrix (in an orthogonal rotation) is split into two matrices; a structure matrix of correlations between factors and items and a pattern matrix of unique relationships between each factor and each observed variable. The meaning of factors is interpreted from the pattern matrix and the

factor correlation matrix. This makes the solutions more difficult to interpret, describe and report (Tabachnick & Fidell, 2013). However, its assumptions are more reasonable, as factors are more likely to be correlated. The following table summarises the goal of analysis for various rotational techniques.

Rotational Technique	Goals of Analysis	Comments
<u>Orthogonal</u> Varimax	Minimize complexity of factors (simplify columns of loading matrix) by maximizing variance of loadings on each factor.	Most commonly used rotation; recommended as default option
Quartimax	Minimize complexity of variables (simplify rows of loading matrix) by maximizing variance of loadings on each variable.	First factor tends to be general, with others sub clusters of variables.
Equamax	Simplify both variables and factors (rows and columns); compromise between quartimax and varimax.	May behave erratically
<u>Oblique</u> Direct oblimin	Simplify factors by minimizing cross- products of loadings.	Continuous values of gamma, or delta, δ , available; allows wide range of factor intercorrelations
Promax	Orthogonal factors rotated to oblique positions.	Fast

Table 3.11: Summary of Rotational Techniques

Although there seems to be a large number of combinations of extraction and rotation techniques available, in reality, the differences among them are trivial (Velicer & Jackson, 1990; Fava & Velicer, 1992). Most of the approaches often result in very similar solutions, especially when there is clear pattern of correlations among the items and the data set is large. If there are any apparent differences after extraction, it often disappear after rotation (Tabachnick & Fidell, 2013).

To interpret a factor, one tries to understand the underlying dimension that unifies the group of items loading on it. In an orthogonal rotation, the information in a loading matrix is reviewed. For oblique rotation, the pattern matrix is used. Generally, as a rule of thumb, only items with loadings of 0.30 and above are interpreted. The greater the loading, the more the item is a pure measure of the factor (Tabachnick & Fidell, 2013). The size of loadings is also influenced by the homogeneity of scores and the sample size where a lower cutoff can be used for sample that produces similar scores on observed items or low sample size (for example below 300). As a guide, Comrey and Lee (1992) suggest that loadings in excess of 0.71 (50% overlapping variance) are considered excellent, 0.63 (40% overlapping variance) very good, 0.55 (30% overlapping variance) good, 0.45 (20% overlapping variance) fair and 0.32 (10% overlapping variance) poor.

Other relationships on the data set are also available. This includes information on the communality of the items and factor correlation coefficients. Communality measures the total amount of variance an original item shares with all other items. Although no statistical guidelines indicate exactly what is an acceptable level, practical considerations dictate that levels of above 0.30 is recommended (Pallant, 2016). As for factor correlations, these are matrices which produces regression-like equations to predict scores on factors from scores of observed items. For orthogonal rotation, a factor transformation matrix is produced whereas for oblique rotation, a factor correlation matrix is available. Generally, correlations should be below 0.30 so that the overlap in variance among factors is below 10% (Tabachnick & Fidell, 2013).

Factors are usually interpretable when the items load highly on them and not the rest. Ideally, each item should load on one factor only. Identifying possible patterns is an exploratory approach. The objective is to find a 'simple structure' with as few factors as possible that will explain as much of the variance in the data set as possible (Thurstone, 1947; Tabachnick & Fidell, 2013). Igbaria, Iivari and Maragahh (1995) provide a general the criteria by identifying and interpreting items that load 0.50 or greater to belong to one factor and 0.35 or lower on other factors.

3.7.3 Common Method Variance

For this study, the measurements were based on the responses of individuals. Hence, the existence of common method variances or bias was possible. Common method variance (CMV) is defined as variances that are attributable to measurement methods and not from the constructs. It is a systematic error variance among constructs using the same source or method, which may result in inflated or deflated relationships between the variables (Podsakoff, MacKenzie, Lee & Podsakoff, 2003; Podsakoff, MacKenzie & Podsakoff, 2012).

Several of the procedural remedies suggested by Podsakoff et al. (2003); Podsakoff et al. (2012) were adopted in the development of the survey instrument to reduce common method variances. This involve efforts to motivate respondents to provide accurate answers to the questions. Wordings and statements used in the instrument were checked and verified to ensure that it is within the respondents' ability. Care was taken to ensure that statements and questions were about processes in which respondents had done. Clear and concise language were used and items was presented in sections. Each section had instructions and explanations that gave the respondent a context for them to answer more consistently and to avoid ambiguity. All the items were also subjected to content validation by experts and a pilot test.

A cover letter was also provided to explain the objective of the study, along with the involvement and interest of various reputable institutions. The importance of respondents' opinion and feedback was stressed. Confidentiality of the responses were also guaranteed,
and respondents were informed that there were no right or wrong answers and that people had different opinions.

In addition, to identify CMV, the Harman's single-factor test was employed. This is a diagnostic technique where all of the variables are loaded into an exploratory factor analysis. The results of the unrotated factor solution is analysed. The basic assumption of this technique is that if a substantial amount of common method variance is present, the unrotated first item would be above 50% (Podsakoff & Organ, 1986).

3.8 Descriptive Analysis

Demographic information such as age, education background, gender, ethnicity, religion, marital status, financial situation, income and details of occupation and dependents are tabulated using frequency tables. This is useful to develop a profile of the overall sample. At the same time, the responses for each item of the variables in the study (beliefs, attitude, subjective norms, perceived behavioural control, intentions) are tabulated using frequency tables. These variables are also analysed using measures of central tendency and spread (Sekaran & Bougie, 2010).

The sample is further divided to groups. Two groups are identified based on the retirement schemes they are participating in. The first are respondents which have opted for a government pension and the second are those which are contributing to Employee Provident Fund (EPF). Generally, those which have opted for a government pension are working in the public universities, whereas those contributing to EPF are from the private universities. These demographic profile and responses to the items in the questionnaire of each of this group is analysed further.

3.9 Structure Equation Modelling

In this study, the hypotheses and collected data are subjected to further statistical analysis. A myriad of statistical methods can be applied, but for this study, multivariate data analysis methods is used. Multivariate analysis refers to statistical techniques that simultaneously analyse multiple measurements on individuals or objects under investigation. This makes it easier to comprehend complex relationships (Hair et al., 2010).

One increasingly popular and powerful multivariate statistical technique is structural equation modelling (SEM). SEM is able to assess latent variables at the observation level (outer or measurement model) and test relationships between latent variables on the theoretical level (inner or structural model) simultaneously (Chin & Newsted, 1999; Bollen, 2011). This enable researchers to test complete theories and concepts (Rigdon, 2016). It also facilitates accounting for measurement error in observed variables. As such, SEM can identify relationships that likely would not otherwise be found.

SEM techniques can be confirmatory or exploratory in nature, depending on the nature of the research study. They are confirmatory when the hypotheses of existing theories and concepts are tested and exploratory when they search for patterns in the data, especially when there is no or only little prior knowledge on how the variables are related.

3.9.1 Path models

In applying SEM, path models are drawn. Path models are diagrams based on theory that visually display the hypotheses and relationships that are examined. An example of a path model is shown in Figure 3.1.

Each latent variable or construct is represented by circles or ovals in the path model (Y_1 , Y_2 and Y_3). Measures for these constructs (indicators or items) are represented in path models as rectangles by (X_1 to X_7). The path coefficients, which measure the relationship between these constructs, are indicated by the single-headed arrows. These arrows are predictive relationships and, with strong theoretical support, can be interpreted as causal relationships.

A path model consists of two elements. The first element is a structural model (inner model) which represents the constructs. The structural model also displays the relationships (paths) between the constructs. The second element are the measurement models (outer models) of the constructs which display the relationships between the constructs and the indicators or items. There are two types of measurement models. The exogenous latent variables explain other constructs in the model and the endogenous latent variables are those constructs that are being explained in the model.





Source: Hair, Hult, Ringle and Sarstedt (2017)

These path models are developed based on theory. Theory is a set of systematically related hypotheses developed to explain and predict outcomes. Thus, theories are multiple hypotheses that are logically linked together and can be tested empirically.

The path models are based on two types of theory. The first theory is the measurement theory which specifies how each construct is measured. The second is the structural theory which outlines how the constructs are related to each other.

In the measurement theory, the direction of causality between a construct and its indicators are important. The direction can be viewed as either reflective or formative (Jarvis, Mackenzie & Podsakoff, 2003). Constructs using reflective indicators (Y_3) have an arrow direction that points from the variable to the indicator, indicating the assumption that the construct causes the measurement (or the covariation) of the indicator variables. These indicators are examined using loadings (Petter, Straub & Rai, 2007), which represent the correlation between the indicators and the component scores (Gefen, Straub & Boudreau, 2000). These indicators should be consistent internally as all items are measuring the same phenomenon. Changes of values in a measure will result in changes for all the other values in the same direction.

Conversely, constructs using formative indicators have arrow direction from the indicators to the construct (Y_1 , Y_2), indicating a causal (predictive) relationship. Measures are based on weights which provides information on the importance of each indicator in the component (Chin, 1998b). They represent different themes or dimensions. Removing a measure would affect the meaning of the construct (Chin, 1998a; Gefen et al., 2000; Jarvis et al., 2003). Hence, these measures need not covary with one another.



Figure 3.2: Reflective and Formative Constructs

Source: Hair et al. (2017)

On the other hand, structural theory shows how the latent variables are related. This is represented by the constructs and the path relationships between them in the structural model. The location and sequence of the constructs are based on theory.

When latent variables serve only as independent variables, they are called exogenous latent variables (Y_1 and Y_2). When latent variables serve only as dependent variables (Y_3) or as both independent and dependent variables, they are called endogenous latent variables. Any latent variable that has only single-headed arrows going out of it is an exogenous latent variable. In contrast, endogenous latent variables can have either single-headed arrows going both into and out of them or only going into them (Y_3).

3.9.2 Covariance based and Variance based Techniques

There are two main approaches to estimating the relationships in a structural equation model. One is a covariance-based technique (CB-SEM) and the other is a variance based partial least squares (PLS-SEM). Each is appropriate for a different research context. Hence, the characteristics and objectives that distinguish these two methods is investigated before making a decision on which approach to use (Hair, Sarstedt, Ringle & Mena, 2011).

A crucial conceptual difference between PLS-SEM and CB-SEM relates to the way each method treats the latent variables in the model. CB-SEM considers the constructs as common factors that explain the covariation between its associated indicators. It analyses how well a proposed theoretical model can estimate the covariance matrix for a sample data set, or how well the theoretical model fit into the observed data. This is done using the maximum likelihood function (Barclay, Thompson & Higgins, 1995). Hence, it is possible to estimate latent variable scores within a CB-SEM framework, which is useful when confirming (or rejecting) theories (Chin, 2010; Gotz, Liehr-Gobbers & Krafft, 2010).

However, these estimated scores are not unique as there are numerous sets of latent variable scores that can fit the model equally well. Consequently, the correlations between a common factor and any variable outside the factor model are indeterminate (depending on which set of factor scores is chosen). As a result, this limitation makes CB-SEM unsuitable for prediction.

PLS-SEM, on the other hand, uses proxies to represent the constructs of interest. These proxies are weighted composites of indicators of a particular construct. These weights also facilitate accounting for measurement error. Weights offer important insights into each item's importance and its relationships with other composites in the structural model. In addition, once weights are established, a single specific (determinate) score for each composite can be produced. The use of weights is superior, in comparison to other SEM methods such as multiple regression (that uses sum scores).

Using these proxies as input, PLS-SEM applies an iterative sequence of ordinary least squares (OLS) regression with the objective of minimising the error terms (and maximising the explained variance) of the endogenous latent variables (Henseler, Ringle & Sinkovics, 2009). This feature achieves the prediction objective of PLS-SEM. PLS-SEM is therefore the preferred method when the research objective is theory development and explanation of variance.

Several additional considerations are important when deciding on which approach to used. For CB-SEM results to be acceptable, the data must be normally distributed (Urbach & Ahlemann, 2010; Hair, Ringle & Sarstedt, 2011). It also works best with a large sample size, of 200 to 800 (Chin & Newsted, 1999; Chin, 2010; Hair et al., 2010). CB-SEM constructs are a composite of the indicators plus an error term model (indeterminate), and model misspecification are noted to have a significant impact on estimates (Chin, 2010). In some studies, CBSEM has difficulties in handling models with fifty or more items (Barclay et al., 1995; Chin & Newsted, 1999; Chin, 2010). As for the direction of relationship among the constructs, CBSEM allows recursive (unidirectional) and nonrecursive (bidirectional) relationships.

PLS has no assumptions on measurement, distribution or sample size. It has statistical properties that makes very robust model estimations with data, regardless of whether the distribution of data is normal or not. Nonetheless, influential outliers and collinearity would still influence the regressions, hence the data and results must be evaluated for these issues (Hair et al., 2010). Although PLS-SEM has been reported to work well for small sample sizes (below 100 cases), the sample size must have sufficient statistical power and be representative of data (Chin & Newsted, 1999; Chin, 2010). Hence, it is important to consider the sample size against the background of the model and data characteristics. Generally, in PLS-SEM, model misspecification has no influence on the overall model. It

can also handle very complex models. However, the specified statistical model should comply with the conceptual model and the technical requirements (Barclay et al., 1995; Chin & Newsted, 1999; Chin, 2010). PLS-SEM model can contain feedback loops or take account of endogeneity if an adequate estimator is used for the structural model. A sufficient number of exogenous variables must also be available.

The use of CB-SEM is limited to research models that use reflective constructs. Although previous studies have used formative measures within the structural model, they usually lead to problems (Henseler et al., 2009). For instance, the use of formative constructs within CB-SEM would create a situation where the explanation of the covariance of all indicators is not possible. Further, the use of CB-SEM in handling both reflective and formative constructs is relatively complicated (Urbach & Ahlemann, 2010).

On the other hand, PLS-SEM can be used to analyse a research model that consists of either reflective, formative or a combination of both reflective and formative constructs at the same time (Chin, 1998b). It can also handle single-item constructs, with no identification problems. In comparison to CB-SEM, PLS-SEM has greater statistical power where it is more likely to render a specific relationship significant when it is in fact significant in the population. It can therefore be applied in a wide variety of research situations.

Table 3.12: Comparison between PLS-SEM and CB-SEM

Criterion	PLS-SEM	CB-SEM
Objective / Approach	Prediction oriented	Parameter oriented
	Variance based	Covariance based
Assumptions	Nonparametric	Parametric
Latent variables scores	Determinate-Explicitly estimated	Indeterminate
Relationship between latent variables and its indicators	Can be modelled in either formative or reflective	Typically only with reflective indicators
Model complexity	Large complexity (e.g., 100 constructs and 1000 indicators)	Small to modest complexity (e.g., less than 100 indicators)
Sample size requirements	20 to 100 cases	200 to 800

Source: Chin and Newsted (1999)

3.9.3 Justification for Using PLS-SEM

This study uses the Structural Equation Modelling (SEM), in particular, the Partial Least Squares (PLS) approach (Hair et al., 2010). From the various criterion discussed, there are advantages in employing PLS-SEM for this study.

PLS-SEM is suitable for exploratory research and theory development (Hair et al., 2011; Hair, Risher, Sarstedt & Ringle, 2019) which is in line with the objective of this study. From the previous chapter, the aim of this study is to investigate the determinants of a behaviour which is intentions to plan financially for retirement. Theory of Planned Behaviour (TPB) is used as the underpinning theory. In TPB, behaviours are assumed to be the result of reasoned decisions which can be predicted and explained using a set of determinants (beliefs, attitudes, subjective norms, perceived behavioural control) (Fishbein & Ajzen, 1975; Ajzen & Fishbein, 1977; Ajzen & Fishbein, 1980). However, whether these set of determinants can be adapted into a financial setting to explain retirement planning behaviours have yet been explored. These variables appear random, yet they are interrelated in many ways. Hence, their different effects have to be analysed. PLS-SEM can assist in understanding and exploring the complex relationships in retirement planning behaviours. The use of such understanding can extent the use of TPB and its body of knowledge in a financial setting.

Several of PLS-SEM capabilities also offer added benefits. Firstly, it has fewer rigid expectations in terms of the distribution of variables and error terms which is suitable for non-normal data. The data collected from this survey is also not normally distributed (Hair et al., 2010). There are several limitations to the distribution of the questionnaires (subjected to each university's procedural requirement, voluntary, random selection of universities).

Next, PLS-SEM can handle model misspecifications, where misspecifications does not have a significant influence on other parts of the model (Henseler et al., 2009). This is particularly important as beliefs are different in each population, and hence can only be obtained from qualitative interviews. It cannot be adapted from previously validated surveys. As for the adapted item, it has not been used extensively in a financial setting. Nevertheless, additional statistical tests (pilot test, EFA) have been conducted to ensure the validity and reliability of the items and constructs.

PLS is suitable for models with large complexity. For this study, there are many items and constructs developed and adapted. In summary, there is a total 9 constructs used (with a total of 43 items) to measure intentions to plan financially for retirement.

3.9.4 Evaluation Using PLS-SEM

Data can be analysed using PLS-SEM in two stages. The first stage involves the analysis of the measurement or outer model. The measurement model links the latent variables to the indicators. In this stage, the reliability and validity attributes are assessed, in particular, the internal consistency reliability, convergence and discriminant validity (Hair et al., 2017). The objective is to ascertain that the indicators correctly measure the constructs (Gefen et al., 2000). This approach assures that good constructs and measures are represented in a valid structural model (Gefen et al., 2000; Barroso, Carrion & Roldan, 2010; Hair et al., 2017).

The second stage involved the analysis of the structural or inner model. The structural model links the latent variables. It subsequently assesses the path and the significance of the hypothesised relationships (Barroso et al., 2010; Hair et al., 2010). Stage two can only proceed if the results in stage one are satisfactory.

Table 3.13: Evaluation of reflective models using PLS-SEM

Source: Hair et al. (2017)

<u>Step 1.</u>	<u>Step 2.</u>
Evaluation of measurement model	Evaluation of structural model
• Internal consistency reliability	• Coefficients of determination (<i>R</i> ²)
Convergent validity	• Predictive relevance (Q^2)
Discriminant validity	• Size and significance of path coefficients
	• f^2 effect sizes

3.9.5 Evaluating the Measurement Model

(a) Internal consistency reliability

The traditional criterion to assess internal consistency is Cronbach's alpha. It provides an estimate of the reliability, using the intercorrelations of the indicators. High Cronbach's alpha is an indication that the construct has the same meaning and range (Cronbach, 1971). However, Cronbach's alpha has limitations. It is sensitive to the number of items in a scale and tends to underestimate the internal consistency reliability. It assumes that all indicators are equally weighted and that there is no tau equivalent among the measures (Werts, Linn & Joreskog, 1974).

As such, with such limitations, composite reliability (CR) is also used to indicate internal consistency reliability. Composite reliability prioritises the indicators according to their loading (Chin, 1998b).

CR values vary between 0 and 1. Higher values indicate a higher level of reliability. Generally, values below 0.6 indicate a lack of reliability. Values between 0.6 to 0.70 are acceptable for exploratory research while values above 0.70 are regarded as satisfactory (Bagozzi & Yi, 1988; Gotz et al., 2010; Urbach & Ahlemann, 2010; Hair et al., 2017).

(b) Convergent validity

Convergent validity measures the extent that each indicator correlates positively with the alternative indicators of the construct. For this purpose, the indicator reliability and Average Variance Extracted (AVE) is evaluated.

To evaluate the indicator reliability, the size of the outer loadings is reviewed. High outer loadings would indicate that the associated indicators have much in common. A general rule would be to accept items with loadings of 0.707 or more. This is because loadings valued at 0.707 would be able to explain 50% of the indicators variance (Barroso et al., 2010; Chin, 2010). Indicators with outer loadings between 0.40 to 0.70 should be removed only if it results in an increase of composite reliability or AVE beyond the suggested threshold value. Outer loadings below 0.40 are recommended for removal.

Chin (1998b); Chin (2010) also caution that any removal of items must be done carefully. This is because removing some of the indicators may influence content validity. In PLS-SEM, poor indicators may have useful information that will help create a better construct score. Also, PLS-SEM have determinate constructs, where low indicators are factored by lower loadings. Hence, indicators with low loadings may still have predictiveness as long as other reliable indicators exist (Chin, 2010).

The second measure for convergent validity is AVE (Fornell & Larcker, 1981). It measures the amount of variance captured by a variable from its indicators, relative to the amount due to measurement error (Chin, 1998b). AVE is more conservative than CR. Ideally, AVE values should be greater than 0.50, as this would mean that more than 50% of the variance is accounted for (Bagozzi & Yi, 1988; Chin, 1998b; Barroso et al., 2010).

(c) Discriminant Validity

Discriminant validity refers to the degree to which the constructs are distinct from the other constructs (Barroso et al., 2010). This would mean that the indicators do not correlate well with other constructs, except for the one that it is theoretically associated with (Gefen et al., 2000). Discriminant validity can be evaluated using cross-loading (Chin, 1998a), Fornell-Larcker criterion (Fornell & Larcker, 1981) and heterotrait-monotrait ratio (HTMT) (Hair et al., 2017).

The first approach to evaluate discriminant validity involves assessing the indicators cross-loadings. Cross-loadings are derived by correlating each item with its construct and to all other constructs (Chin, 1998b). The correlation between the constructs and items would show a pattern, where indicators of its theoretically assigned construct will load higher in comparison to other items (Gefen et al., 2000). Indicators that load higher with constructs other than the one it is intended to should be considered for removal.

Another consideration for discriminant validity involves using the Fornell-Larcker criterion (Fornell & Larcker, 1981). This criterion involves comparing the square root of the AVE values with the construct correlations. The square root of each construct's AVE should be greater than its highest correlation with another construct, as it shares more variance with its associated indicators in comparison to others (Fornell & Larcker, 1981; Gefen et al., 2000).

Henseler, Ringle and Sarstedt (2015) found weaknesses when cross-loadings and the Fornell-Larcker criterion is used for discriminant validity. Cross-loading is unable to detect discriminant validity for perfectly correlated constructs, while Fornell-Larcker is ineffective for indicator loadings that have small differences (Voorhees, Brady, Calantone & Ramirez, 2016).

To counter such problems, the heterotrait-monotrait ratio (HTMT) method is proposed (Henseler et al., 2015). HTMT is the ratio of the between-trait correlations to the within-trait correlations. A threshold value of 0.85 and 0.90 has been recommended to ascertain discriminant validity. The guidelines for assessing reflective measurement model can be summarised in the following table.

Validity Type	Criterion	Guidelines
Internal consistency	Composite Reliability (CR)	CR > 0.7 (for exploratory study)
reliability		CR > 0.8 (advance research)
		CR < 0.6 (lack of reliability)
Convergent validity	Outer loadings	Indicator loading > 0.707
	Average Variance Extracted (AVE)	AVE > 0.50
Discriminant validity	Cross loading	Item's loading of each indicator is
		highest for its designated construct.
	Fornell and Larcker	The square root of the AVE of a
		construct should be greater than the
		correlations between the construct
		and other constructs in the mode
	Heterotrait - monotrait ratio	A threshold value of 0.85 and 0.90
	(HTMT)	

 Table 3.14: Guidelines for Assessing Reflective Measurement Model

3.9.6 Evaluating the Structural Model

Once the evaluation of the measurement model is validated, the second stage, which is the analysis of the structural model can begin. In this stage, the model's predictive power and the relationship between the constructs are important. The evaluation of the structural model involves several steps, as shown in the following figure.



Figure 3.3: Structural Model evaluation procedures

Source: Hair et al. (2017)

(a) Assess the structural model for collinearity issues

When two constructs are highly correlated, the issue of collinearity arises. This is possible when the set of independent variables could predict some other variables as well. As a result, minor variations in the sample may have a significant effect on the loadings. As such, each set of predictor constructs need to be assessed. Indicators used to assess collinearity is the Variance Inflation Factor (VIF). According to Hair et al. (2017), any VIF value above 5 would indicate potential collinearity problems.

(b) Assess the significance and relevance of the structural model relationship

Before assessing the significance of the structural model relationship, a bootstrapping routine is applied. This is a nonparametric technique that is used to estimate the accuracy of the estimates in PLS-SEM. This technique creates sample sets in order to obtain estimates for each parameter in the model. Each sample set is obtained via replacements to the original data set. The bootstrapping routine would yield the standard error, t-values, p-values and values for all the structural paths. The recommended bootstrapping techniques have suggested the use of 5,000 resamples (Hair et al., 2017).

Upon completion of the bootstrapping routine, the path coefficient should be examined. The path coefficient's algebraic sign, magnitude and significance must be checked. Path coefficient values can range from -1 to +1. The coefficient that is closer to +1 represents strong positive relationships while those closer to -1 indicates a strong negative relationship. Hence, those paths with signs that are contrary to the assumed theoretical relationship do not support the hypotheses.

The path coefficients magnitude indicates the strength of the relationship between the constructs. The higher the coefficient, the stronger its effect on the endogenous latent variable. Also, path coefficients should be statistically significant. In assessing the path coefficient, Sellin and Keeves (1994); Urbach and Ahlemann (2010) have stressed that the path coefficient should exceed 0.10 to have an impact on the model. Also, it should be significant at the 0.05 level.

(c) Assess the coefficient of determination (\mathbf{R}^2)

The coefficient of determination (R^2) measures the model's predictive power. This coefficient is calculated as the squared correlation between a specific endogenous construct's actual and predicted values. It represents the amount of variance in an endogenous construct that can be explained by all the related exogenous constructs. R^2

values range from 0 to 1, where higher values indicate a higher level of predictive accuracy (Barclay et al., 1995; Hair et al., 2017). In the context of this study, the coefficient of determination (R^2) would assess the extent that the variance in attitudes, subjective norms and perceived behavioural control is explained by beliefs (behavioural beliefs, normative beliefs, control beliefs). Subsequently, how well attitudes, subjective norms and perceived behavioural control explains intentions to plan financially for retirement is assessed.

Generally, R^2 values depend on the model complexity and the research discipline. Backhause, Erichson, Plinke and Weiber (2003); Gotz et al. (2010); Hair et al. (2017) argue that the acceptable threshold of R^2 depends on the study and the respondents. For example, the value of 0.20 is considered high in research involving consumer behaviours (Hair et al., 2017).

As a guide, R^2 values of 0.75, 0.50 or even 0.25 for endogenous latent variables can be described as substantial, moderate or weak (Hair et al., 2011). However, Chin (1998b) is of the opinion that the value of R^2 around 0.67 is substantial, 0.33 are average and values of 0.19 and lower are weak. At the same time, Cohen (1988) have described the R^2 levels of 0.26, 0.13 and 0.02, respectively, as substantial, moderate and weak. Similarly, some researchers also argue that the R^2 for endogenous variables or constructs should be greater than 0.1 to be deemed adequate (Falk & Miller, 1992; Santosa, Wei & Chan, 2005). For this study, Cohen (1988) values would be used. These values are widely accepted in the field of psychology and social sciences.

 R^2 is also influenced by the number of predictors constructs in a model. The more predictors a model have; the higher is the R^2 results. Hence, it is advisable to use the adjusted R^2 , which controls for model complexity (Hair et al., 2017). Adjusted R^2 is computed as follows:

$$R_{adj}^2 = 1 - (1 - R^2) \cdot \frac{n - 1}{n - k - 1}$$

(d) Assess the effect size (f^2)

The effect size (f^2) is used to assess the change in R^2 of a dependent construct when a specific predictor construct is used or omitted. The objective is to evaluate whether the omitted construct has a substantive impact on the dependent constructs. The effect size is calculated using the following formula:

$$f^{2} = \frac{R^{2} included - R^{2} excluded}{1 - R^{2} included}$$

where

- R^2 included: R^2 of the dependent variable when the independent variable is included
- R^2 excluded: R^2 of the dependent variable when the independent variable is excluded

The effect size indicates the influence of the predictor constructs on the dependent variable. The higher the effect size (f^2) value, the higher the influence. According to Cohen (1988), the effect size of 0.02, 0.15 and 0.35 indicates a small, medium or large influence, respectively. A small f^2 , however, does not necessarily imply a negligible effect (Wilson, 2010). In the context of this study, the effect size of the predictor variables (attitudes, subjective norms and PBC) is evaluated.

(e) Assess the predictive relevance (Q^2) (Blindfolding)

In addition to R^2 , Stone-Geisser's Q² value (Geisser, 1975) is also recommended to be used in examining the inner model's predictive relevance (Henseler, Hubona & Ray, 2016; Hair et al., 2017). When a model has predictive relevance, it means that the model can accurately predicts data not used in the model estimation. In the structural model, Q^2 values larger than zero for a specific reflective endogenous latent variable indicate the path model's predictive relevance for a particular dependent construct.

In PLS-SEM, the blindfolding procedure is used to obtain the predictive relevance (Q^2) value of the model. This procedure systematically eliminates data points. In order to do this, the omission distance (d) has to be specified. Using this technique, every dth data point is omitted in the endogenous construct's indicators as well as the parameters of the remaining data. The omitted data is considered as missing values and replaced with mean values. As such, the blindfolding technique is used as a general cross-validating metrics. PLS can accurately predict the data points of indicators in reflective measurement models of endogenous constructs and endogenous single item constructs. The omission distance should be between 5 and 12 is considered to be sufficient (Wold, 1985; Chin, 2010). Also, the sample size should produce a round number (not integer values) when divided by the omission distance.

Two different forms of Q^2 was generated from the blindfolding procedure (Chin, 1998b; Chin, 2010). The cross-validated redundancy measures the ability of the model to predict the endogenous manifest variables. It indicates the quality of the structural model (Hair et al., 2017). The cross-validated communality measures the ability of the path model to predict the manifest variables from their latent variable score and indicates the quality of the measurement model (Hair et al., 2017).

Chin (1998b); Chin (2010) suggests using the cross-validation redundancy measure to evaluate the predictive relevance of the structural model. Q^2 measures how well the model and its parameters reconstruct the observed values. Q^2 values should be more than zero to indicate that the model has predictive relevance. Values below zero would mean that the model has no predictive relevance (Hair et al., 2017). In this study, the endogenous

variables (attitudes, subjective norms, PBC and intentions) predictive relevance are assessed.

3.10 Summary

This chapter outline the research methodology adopted. A quantitative methodology is used in its research design. Primary data is collected via a survey. To address Hypotheses 1 to 3 (attitudes, injunctive subjective norms, descriptive norms, perceived behavioural control and intentions), scales are adapted from existing literature. For Hypotheses 4a, 4b and 4c, qualitative interviews are conducted to identify the accessible beliefs of the population of study. The beliefs identified are used to developed the behavioural beliefs, normative beliefs and control beliefs scales.

As survey is the main data collection method, care is taken to ascertain the goodness of the measures adapted and developed. A content validation by experts and a pilot test was used to checked the validity and reliability of the instrument developed.

Details was provided on the sampling procedures. The data collected goes through a preliminary data analysis which involve data screening and Exploratory Factor Analysis. Also, descriptive analysis is performed. Subsequently, SEM are discussed, along with using PLS-SEM statistical techniques and guidelines.

CHAPTER 4: RESULTS

For this study, the principle method used was quantitative, where a survey instrument is used to collected the data. Quantitative methods are also used in many behavioural studies (Zikmund, 2003; Neuman, 2006; Sekaran & Bougie, 2010). In financial behaviour, in particular the field of financial planning and retirement research, quantitative methods are recommended as it is not only able to elicit responses on unobservable behaviour, but is also considered the most economical and feasible method to elicit the views of a large number of participants.

4.1 **Results of Preliminary Analysis**

Before analysing the results using PLS, the data is screened. For this purpose, analysis of missing data, normality, outliers and checking for collinearity issues is conducted. The objective of this process is to avoid any potential problems at the later stage of the analysis.

4.1.1 Data Screening

During data screening, the 'missingness' in the data set is examined. Two forms of missingness can be identified from the data. These 'missingness' are coded using 99 and 999. The code '99' identifies items in the questionnaire that are unanswered or left blank by the respondents. From the codes, the missing values can be identified to three items from Section E-Personal details, that is IC2, IC3 and CHILD3. These are regarding financial matters related to the individual.

IC2 are information on the estimated total monthly household debt repayments a person makes to financial institutions. One person (0.2%) did not answer the question. As for IC3, the item measures the satisfaction level of the respondent to their current financial situation.

2% of the sample did not respond to the question. Meanwhile, CHILD3 documents information on number of financially dependent children. Two persons did not answer the question (0.6%). Overall, the missing value is below 5% of the total number of cases. Pairwise deletion would be used to deal with these missing values, where the case is only excluded for analysis for the items with missing data.

Table 4.1: Missing data

Variables	Code	Recorded	Missing values		Total
		responses	Frequency	Percent (%)	
Monthly household debt	IC2	499	1	0.2	500
Satisfied with current financial	IC3	490	10	2	500
situation					
Nos. of financially dependent	CHILD3	328	2	0.6	330
children					

Note: Missing data - code 99

The second form of missingness is pertaining the 'non-applicable' responses, which are coded as '999'. In total, these responses can be identified to nine items. The first item (CHILD1) is from Section E-Personal details. CHILD1 identifies the segments of the sample that have children. Only 66% have children. The remainder eight items are from Section C-The people around you and your environment. These are regarding normative beliefs. They include injunctive and descriptive normative beliefs identified from the codes INB1, INB2, INB3, INB4 and DNB1, DNB2, DNB3, DNB4, respectively. These variables will be used for further analysis in PLS-SEM.

Variables	Code	Recorded Responses Non-a		Non-applica	n-applicable responses		
		Count	Percent	Count	Percent		
Nos. of children	CHILD1	330	66.0	170	34.0	500	
Injunctive	INB1	375	75.0	125	25.0	500	
Normative Beliefs	INB2	489	97.8	11	2.2	500	
	INB3	470	94.0	30	6.0	500	
	INB4	454	90.8	46	9.2	500	
Descriptive	DNB1	368	73.6	132	26.4	500	
Normative Beliefs	DNB2	487	97.4	13	2.6	500	
	DNB3	471	94.2	29	5.8	500	
	DNB4	458	91.6	42	8.4	500	

 Table 4.2: Non applicable responses

Note: Non applicable responses - code 999

Most of the missing values coded 999 are also above 5% of the total number of cases. However, INB1, INB2, INB3, INB4 and DNB1, DNB2, DNB3, DNB4 are normative beliefs items. The presence of missing values will influence the results generated and have to be dealt with accordingly. Fishbein and Ajzen (2010) recommends that the average of each indicator is computed to derive an injunctive and descriptive normative belief composite. Hair et al. (2017) further supports this as a better approach to overcome missing data as it minimizes the decrease in variability of responses.

It is also important to test for normality. This study uses PLS in its analysis. PLS is suitable for studies involving non-normal data. To assess normality, the histogram and Normal Q-Q plots were viewed. The histogram does not show a normal distribution curve. For the Normal Q-Q plots, it also does not fit into the standard normal distribution line. In addition, the Kolmogorov-Smirnov and Shapiro-Wilk test of normality were significant (p<0.05), indicating that the distribution was not normal. The results of the normality tests confirm that the data is not normally distributed.

The data was further examined for outliers. Mahalanobis distances were generated. Examination of the D_2 values did not indicate outliers. Thus, all observations were retained for analysis.

To assess collinearity issues, the VIF is computed. As this study used reflective indicators, the inner VIF was assessed. VIF values should be below 5 to be of the acceptable range. From the following table, all the VIF figures are less than 5, thus indicating that there were no collinearity issues (Hair et al., 2017).

Constructs	Attitudes	Descriptive Norms	Injunctive Norms	Intentions	PBC	Subjective Norms
Attitudes				1.104		
Subjective Norms		1.000	1.000	1.012		
PBC				1.096		
Intentions						
Behavioural Beliefs	1.000					
Normative Beliefs						1.000
Control Beliefs					1.000	

 Table 4.3: Collinearity Statistics

4.1.2 Exploratory Factor Analysis

The objective of using Exploratory Factor Analysis (EFA) in this study is to uncover the underlying factor structure and interrelationships among the variables. It is a form of preliminary confirmation on the expected factors to be extracted which would support the used of Theory of Planned Behaviour. There is a total of 43 items developed and adapted using TPB guidelines. The list of these items is in Table 4.5.

For this study, the sample size used for EFA is 500 and missing data is dealt with using 'pairwise deletion'. This sample size is large enough for correlations to be reliably estimated (Nunnally, 1978; Hair et al., 2010; Tabachnick & Fidell, 2013). Oblique rotation (Direct Oblimin) is used, where all factors are assumed to be correlated. For factor extraction, principal axis factoring (PAF) is chosen, where the common variance is analysed and error variance removed.

An assessment of the suitability of the data for factor analysis is conducted. Inspection of the correlation matrix revealed the presence of coefficients of 0.30 and above, with Kaiser-Meyer-Olkin exceeding the recommended value of 0.6 (Kaiser, 1974) and Bartlett's Test of Sphericity shows that there is statistical significance (p<0.05) (Tabachnick & Fidell, 2013). However, attitudes, which was initially developed with 5 items (DA1_r, DA2_r, DA3, DA4_r, DA5), had separated into two factors. Items in the first factor (DA1_r, DA2_r, DA4_r) reflect all three underlying components of attitude (overall - DA1_r, experiential - DA2_r, instrumental - DA4_r) but items in the second factor only has two components (experiential - DA5, instrumental - DA3). Therefore, the first factor has stronger and a more comprehensive representation of attitudes. As for the second factor, DA3 (worthless-useful) and DA5 (unpleasant-pleasant) loaded later in EFA, and may not be suitable or well understood in the context of planning for retirement. It was decided to remove the two items and to run EFA again.

Using an exploratory iterative approach by analysing the eigenvalues, screeplot and parallel analysis, a final 8-factor structure is extracted. This solution is able to explain the variance in the data set with as few factors as possible. A summary of the items and variables were provided in Table 4.5.

Factor	Variable	Initial	theoretical framework	8-factor model		8-factor model For	
		Qtn	Item Codes	Qtn	Qtn Item Codes		
1	Behavioural	9	BA1, BA2, BA3	9	BA1, BA2, BA3,	DPBC4	
	beliefs		BC1, BC2, BC3		BC1, BC2, BC3,		
			BSC1, BSC2, BSC3_r		BSC1, BSC2,		
					DPBC4		
2	Control	11	CC1_r, CC2_r, CC3_r,	13	CC1_r, CC2_r,	BSC3_r	
	beliefs				CC3_r,		
			CR1_r, CR2_r, CR3_r,		CR1_r, CR2_r,	IN1	
					CR3_r,		
			CR4_r, CR5_r		CR4_r, CR5_r,	DPBC3_r	
			CRS1_r, CRS2,		CRS1_r, CRS3_r,		
			CRS3_r				
					IN1,		
					BSC3_r		
					DPBC3_r		
3	Descriptive	4	DNB1, DNB2, DNB3,	4	DNB1, DNB2,		
	Normative		DNB4		DNB3, DNB4		
	beliefs						
4	Intentions	3	DINT1, DINT2,	3	DINT1, DINT2,		
			DINT3		DINT3		
5	Attitudes	5	DA1_r, DA2_r, DA3,	5	$DA1_r, DA2_r,$	DA3,	
			DA4 r DA5		DA4_r DA3_DA5	DA5	
			Diri_i, Drie		Dite, Dite	Diffe	
6	Iniunctive	4	INB1, INB2,	2	INB3, INB4		
	Normative	\mathbf{D}^{+}	INB3 INB4				
	beliefs						
7	Perceived	4	DPBC1_DPBC2	3	DPBC1_DPBC2		
	Behavioural		DPBC3 r DPBC4	5			
	Control		DI DOS_I, DI DOI		CRS2		
8	Subjective	3	IN1 IN2	4	IN2		
Ŭ	norms		DN1		DN1		
			DIVI		INB1 INB2		
	Total	43				6	
	101111	75				0	

 Table 4.4: Initial theoretical framework and the 8-factor model

The first factor has a variance of 19.249% which comprise of 8 behavioural beliefs items (BA1, BA2, BA3, BC1, BC2, BC3, BSC1, BSC2) and one perceived behavioural control (DPBC4) item. Another item, BSC3_r converges into Factor 2-Control beliefs. Further analysis into DPBC4 ('Whether I plan financially for retirement or not is entirely up to me') reveal that it has loadings and communalities below 0.30. It is not loading as intended and the communality also indicates that it is not a substantive part of the factor solution. This item is removed. This factor is named Behavioural beliefs where all items have loadings and communalities above 0.30.

The second factor has a variance of 14.126%. It comprise of 10 items from control beliefs (CC1_r, CC2_r, CC3_r, CR1_r, CR2_r, CR3_r, CR4_r, CR5_r, CRS1_r, CRS3_r), one item each from behavioural beliefs (BSC3_r,), subjective norms (IN1) and perceived behavioural control (DPBC3_r). However, BSC3_r ('Planning financially for retirement is a burden - it prevents me from enjoying my life now'), IN1 ('I feel the pressure to plan financially for retirement') and DPBC3_r ('The decision to plan financially for retirement is beyond my control') was not designed to measure control beliefs. In addition, they have the lowest loading and communalities among all the items in this factor. All three items are removed. The remaining items are named Control beliefs. All the items have loadings and communalities above 0.30.

The third factor has a variance of 6.372%. It comprises of 4 descriptive normative beliefs items (DNB1, DNB2, DNB3, DNB4). The fourth factor has a variance of 4.725% and comprise of 3 items of intentions (DINT1, DINT2, DINT3). The fifth factor has a variance of 4.139%. It comprises of 3 items of attitudes (DA1_r, DA2_r, DA4_r). All items in Factor 3, 4 and 5 have loadings and communalities above 0.30, and is named accordingly.

The sixth factor has a variance of 3.764%. It initially comprises of 4 injunctive normative beliefs items (INB1, INB2, INB3, INB4). However, after running EFA, only two items are retained, representing 50% of the variables (INB3, INB4). Hence, factor 6 is recognized as injunctive normative beliefs. All items have loadings and communalities above 0.30. INB1 and INB2 converge into Factor 8 (subjective norms).

The seventh factor has a variance of 3.425%. It comprises of 4 perceived behavioural control items (DPBC1, DPBC2, DPBC3_r, DPBC4). However, only two items (DPBC1, DPBC2) are retained after running EFA and the factor was named perceived behavioural control. All items have loadings and communalities above 0.30. DPBC3_r and DPBC4 converge into Factor 2 (control beliefs) and Factor 1 (behavioural beliefs), respectively and has been removed. An additional variable (CRS2), initially developed to measure risks in the control belief variable, converge in to this factor. CRS2 ask the question 'I am confident of what will happen to my investments in the future'. It has similarity with the self-efficacy component in perceived behavioural control, which could be the reasons why it converges to this factor.

The eighth factor has a variance of 2.319%. It comprises of three subjective norms items (IN1, IN2, DN1). After running EFA, two items (IN2, DN1) are retained and was named subjective norms. These items have communalities above 0.30. IN1 converge with Factor 2-Control beliefs and has been removed. Two additional items (INB1, INB2), initially developed as injunctive normative belief variables has converge to Factor 8. These items have an 'non-applicable' option in the scales developed, hence not all respondents were able to answer the questions about certain referents. This could have an impact on its convergence. Nevertheless, it is still reflective of the impact of the social environment on the individual.

In summary, items DPBC4 (Factor 1), BSC3_r, IN1, DPBC3_r (Factor 2), DA3 and DA5 (Factor 5) are removed. The total variance extracted is at 58.119%, with the total number of items reduced to 37. The Kaiser-Meyer-Olkin value is at 0.83, with Bartlett's Test of Sphericity at significant levels (p<0.05). A number of correlation coefficients above 0.30 were identified. Loadings and communalities of all items are at acceptable levels. Details of the statistical results is available in Appendix Q.

This solution revealed the presence of a somewhat 'simple structure' (Thurstone, 1947), with items that form factors showing strong loadings, and each item loading substantially on only one factor. The factor correlation matrix further confirm that each variable is factorially distinct. These results support the use of the variables in Theory of Planned Behaviour. This structure is further confirmed in the measurement model assessment in PLS, where reliability, convergent and discriminant validities were analysed.

4.1.3 Common Method Variance

In the previous chapter, several procedural remedies were incorporated to reduce the common method variance in this study. An inspection of the correlation matrix of the variables did not reveal any high correlations between the constructs. Hence, there is no initial evidence of the possibility of common method variance (CMV) (Bagozzi, Yi & Phillips, 1991).

However, to test further, the Harman's single-factor test was employed (Podsakoff & Organ, 1986; Podsakoff et al., 2003). The results show that the unrotated factor analysis explains only 16.79% of the variance, which is below the threshold level of 50%. As a result, CMV is not view as a problem for this study. The following table is an extract of the results of Harman's single-factor test.

Factor	Initial Eigenvalues			Extrac	ared Loadings	
	Total	% of Variance	Cumulative %	Total	% of Variance	Cumulative %
1	7.888	18.344	18.344	7.222	16.795	16.795

Table 4.5: Results of Harman's single-factor test

4.2 **Respondents' Profile**

The unit of analysis for this study are Malaysians, currently employed in universities. In total, there are 500 respondents, where 44.2% are from public universities, largely funded by the government, while 55.8% are private universities which relies on corporate investments, alumni or student funding. 84% of the public universities' respondents have opted for the pension scheme.

In total, 185 respondents that has chosen the government pension scheme (37%) whilst 315 are not in the pension schemes (63%). The demographic details (age, education background, gender, race or ethnicity, religion and marital status) of the respondents are analysed. The two groups, those opting for the government pension scheme and those not in the pension scheme are available in Appendix M.

(a) Age

The age range of the respondents are from below 20 years to above 60 years. 45.4% of the respondents are between 25-29 (20.4%) and 30-34 (25%) years of age. Of this 32.9% are have opted for the pension schemes and 52.7% from the non-pension group. The remainder of the 55% of the sample respondents are below 20 and from ages 35 to above 60 (35-39 - 16.0%, 40-44 - 13.2%, 45-49 - 9.2%, 50-54 - 6.8%, above 60 - 3.2%, 20-24 - 3.0%, 55-59 - 3.0% and below 20 - 0.2%).

For the respondents who opted for the government pension scheme, 55.7% are from ages 30-44 (30-34 - 20.5%, 35-39 - 15.7% and 40-44 - 19.5%). This is followed by the other age groups, namely ages 35-39 (15.7%), 50-54 (12.4%), 25-29 (12.4%), 45-49 (8.1%), 55-59 (6.5%), above 60 (2.7%), 20-24 (2.2%) and below 20 (0.0%).

As for those that do not opt for the pension scheme, 52.7% are mainly from the ages 25 to 34, (30-34 - 27.6% and 25-29 - 25.1%). The remainder are from ages 35-39 (16.2%), 45-49 (9.8%), 40-44 (9.5%), 20-24 (3.5%), 50-54 (3.5%), above 60 (3.5%), 55-59 (1.0%) and below 20 (0.3%).



Figure 4.1: Respondents' age profile

(b) Educational background

As for educational background, options are provided (SPM / Certificate Level, Diploma, Bachelor Degree, Master, PhD, Others). Over 90% of the sample have been through tertiary education and have at least a Bachelor's degree, where those with postgraduate qualifications totals about 67%. Most of the respondents have Master qualifications (45.6%), followed by Bachelor Degree (23.6%), PhD (21.4%), Diploma (6.6%), SPM/Certified level (2.2%) and others (0.6%). Those with Diploma and SPM/Certified level are the minorities in this sample (8.8%).

In summary, for the respondents who opted for the government pension scheme, 92.5% have tertiary education (Master - 46.5%, Bachelor Degree - 15.7%, PhD - 30.3%). Meanwhile, for the non-pension respondents, 90.6% have tertiary education (Master - 45.1%, Bachelor Degree - 28.3%, PhD - 16.2%, Others - 1.0%). Further analysis within each of the education background categories also reveal that most of the tertiary levels are in the non-pension scheme.



Figure 4.2: Respondents' education background profile

(c) Gender

For gender, the sample consist mainly of females (68.4%) while males comprise of 31.6%. Similarly, those in the pension scheme (69.2%) and non-pension schemes (67.9%) are also female.



Figure 4.3: Respondents' gender profile

(d) Ethnicity

Meanwhile, the ethnic composition of the sample comprises mainly of Malays (52.80%) and Chinese (37.80%). The Indians (6.8%) and Others (2.6%) total less than 10%. The other category consists mainly of the indigenous races in East Malaysia (Kadazan, Bidayuh, Dusun), Eurasian and Punjabi. The ethnic distribution is reflective of Malaysia's population. Those that have opted for the government pension scheme, majority are Malay (76.2%), follow by Chinese (16.8%), others (3.8%) and Indians (3.2%). For those not in

the pension scheme, they comprise of Chinese (50.2%) follow by Malay (39%), Indian (8.9%) and Others (1.9%).

In summary, a major proportion of those that opted for the government pension scheme are Malays (76.2%). For those not in the pension scheme, they are mostly Chinese and Indian (59.1%).



Figure 4.4: Respondents' ethnicity profile

(e) Religion

As for religion, most of the respondents are Muslims (52.80%) and Buddhists (26.8%). The other religions are Christianity (13.4%), Hindus (4.8%) and others (atheists, Sikhs, Taoism) (2.2%), which total 20%. Similar to ethnicity, the Muslims dominate the sample, with the Buddhist coming in as second. As for those that have opted for the government pension scheme, majority are Muslims (77.3%), follow by Buddhists (11.4%), Christians (8.1%), Others (2.2%) and Hindus (1.1%). Similarly, for those not in the pension scheme, majority are also Muslims (38.4%) and Buddhists (35.9%), followed by the Christians (16.5%), Hindus (7%) and Others (2.2%).



Figure 4.5: Respondents' religion profile

(f) Marital status

On marital status, majority of respondents are either married (62.8%) or single (34.2%). 3% are from the widowed (0.6%) or divorced/separated (2.4%) category. For those in the government pension scheme, majority are married (74.6%), follow by the unmarried (19.5%). The other categories total only about 5.9% (widowed, divorced/separated). Similarly, for those not in the pension scheme, majority are also married (55.9%), with the unmarried (42.9%) very close behind. The other categories total only about 1.3% (widowed, divorced/separated).



Figure 4.6: Respondents' marital status profile

(g) Occupational details

Occupational details are also collected from the sample respondents. Most of the respondents (69.2%) are academics (lecturers, researchers and teaching staff) while the rest (30.8%) are non-academics (2% top management, 18.6% management and professionals and 10.2% are support staff). Similarly, those in the pension scheme (70.8%) and not pension schemes (68.3%) are also academics. Only 29.2% of the non-academics are in the pension scheme, in comparison to 31.7% in the non-pension scheme.


Figure 4.7: Respondents' occupation type profile

(h) Financial information

As for financial information, respondents' annual household income varies from less than RM30,000 (13.4%) to above RM150,000 (3.2%). 67.4% have an average annual household income of less than RM30,000 to 70,000, with majority from the RM30,000-50,000 range (30.8%) while the RM50,001-70,000 (23.3%) comes second and those with income of less than RM30,000 are about 13.40%. As for those with annual incomes above RM90,000, there are only about 32.6%.

As for those in the pension scheme, 54% of the respondents are from of the income range of RM30,000-90,000 (RM50,001-70,000 - 24.3%, RM30,000-50,000 - 16.2% and RM70,001-90,000 - 13.5%). The remainder are from the income range of less than RM30,000 (12.4%), RM90,001-110,000 (12.4%), RM110,001-130,000 (11.9%), above RM150,000 (4.9%) and RM130,001-150,000 (4.3%). As for the non-pension scheme,

75.9% are from the income range of less than RM30,000 to 70,000 (RM30,000-50,000 - 39.4%, RM50,001-70,000 - 22.5%, less than RM30,000 - 14%). The remainder are from the income range of RM70,001-90,000 (10.8%), RM90,001-110,000 (7.3%), RM110,001-130,000 (3.2%), above RM 150,000 (2.2%) and RM130,001-150,000 (0.6%).



Figure 4.8: Respondents' annual personal income profile

Additional information is also obtained on the monthly household debt repayment to financial institutions. Only 10.2% has no debt. Most respondents report a monthly debt commitment of RM1,001-3,000 (47.4%). This is followed by the debt range of RM3,001-5,000 (18%) and less than RM1,000 (14.4%). This is also similar for both the pension and non-pension scheme groups, where the debt range between less than RM1,000-5,000.



Figure 4.9: Respondents' monthly household debt repayment profile

Respondents were also asked if they were satisfied with the current financial situation. Ten persons did not respond to this question. 73.6% of the respondents were either slightly (24.0%) or moderately satisfied (49.6%). Only 12.4% were not at all satisfied, whereas those who were very satisfied or extremely satisfied totals 12%.

The results are also similar for the respondents in the pension scheme and non-pension scheme. 70.2% the respondents in the pension scheme and 75.5% from the non-pension scheme reported to be either slightly or moderately satisfied with their current financial situation. For those who were not at all satisfied, 9.7% were from the pension scheme and 14% from non-pension scheme. Meanwhile, for those who were very satisfied or extremely satisfied, 18.9% were from in the pension scheme and 7.9% from non-pension scheme.



Figure 4.10: Respondents' satisfaction with current financial situation

In summary, most of the respondents in the pension scheme are satisfied with their financial situation, where they are either slightly, moderately, very or extremely satisfied. Only a minority were not satisfied at all. Although most of the respondents in the non-pension scheme also reported almost similar results, those that are not at all satisfied are slightly higher (14.3%).

4.3 Descriptive Analysis

A preliminary analysis of the responses was initiated for all the variables in the theoretical framework. Preliminary analysis is an important feature in most empirical studies, as they provide a summary of the survey data. Tabachnick and Fidell (2013) argued that describing and making inferences about the data set are important for empirical research. It will help to explain each variable through its measures. This will provide researcher an overview of the responses of the survey. Details of the results are available in Appendix N.

The items in the survey are measured using semantic differential and Likert scales. Attitudes are measured using semantic differential scales. Beliefs, subjective norms, perceived behavioural control and intentions are measured by a set of Likert scales, ranging from 1-Strongly disagree, 2-Disagree, 3-Neutral, 4-Agree and 5-Strongly agree to reflect the level of agreement of the respondents (Vagias, 2006).

(a) Behavioural beliefs

Behavioural beliefs link a behaviour to the probability that it will produce a given outcome. The combination of all accessible beliefs will determine the attitude towards the behaviour (Ajzen, 2019). However, beliefs differ amongst populations and context, and it is recommended to identify them from pilot studies. For this purpose, preliminary interviews were conducted. Three outcomes related to autonomy (BA1, BA2, BA3), coping (BC1, BC2, BC3) and self-control (BSC1, BSC2, BSC3_r) were identified to be important behavioural beliefs for the respondents.

For the outcomes related to autonomy, most respondents (54.6%) agree with the first statement; that planning would help them imagine their future financial situation (BA1). Only less than 1% strongly disagree. In summary, those that agree with the statement (respondents who chose to agree or strongly agree) total 85% while those who chose to disagree (either disagree or strongly disagree) are less than 4%. Only 12.4% took a neutral stand. Similarly, 87% of the respondents that opted for pension scheme also agree or strongly agree with the statement, with only 4.3% disagreeing (or strongly disagreeing) with 8.6% remaining neutral. Although, most of the respondents in the non-pension scheme also agree (or strongly agree) with the statement (87%), a higher proportion chose to remain neutral (14.6%) whilst those that disagree (or strongly disagree) amounts to 2.6%.

As to whether planning will help them set directions and provide them with flexibility and choices in life (BA2), most respondents agree with the statement (56%). It recorded the lowest responses (0.8%) in the strongly disagree scale. In total, 87.4% believe that this is possible (agree or strongly agree), with less than 3% that disagree (or strongly disagree). Only 10% are neutral. Similarly, 88.7% of the respondents that opted for the pension scheme also agree (or strongly agree) with the statement, with only 3.8% disagreeing (or strongly disagree) and 7.6% remaining neutral. Although, those that are in the non-pension scheme also agree (or strongly agree) with the statement (86.6%), a higher proportion also chose to remain neutral (14.6%) whilst those that disagree (or strongly disagree) amounts to 1.9%.

As for whether planning will help respondents to be independent and self-sufficient (BA3), most respondents agree with the statement (53.2%). Only about 0.8% strongly disagree. In total, those who have chosen to agree (or strongly agree) totals 86% of the respondents, with only 11.4% remaining neutral. Only less than 3% disagree (or strongly disagree) that planning will help them achieve independence and be self-sufficient. Similarly, 88.1% of the respondents that opted for pension scheme also agree (or strongly agree) with the statement, with only 1.6% disagreeing (or strongly disagree) and 10.3% remaining neutral. Those that are in the non-pension scheme which agree (or strongly agree) with the statement are at 84.8%, 12.1% are neutral with those that disagree (or strongly disagree) at 3.2%.

The next three set of statements measures respondents' beliefs that planning will help them cope with life (BC1, BC2, BC3). For BC1, most respondents (49.6%) agree with the statement that planning will help them cope with life's uncertainties. Only about 0.8% strongly disagree. In total, 88.2% agree (or strongly agree), with about 8.6% remaining neutral. Only less than 4% disagree (or strongly disagree) with the statement. Similarly, 90.2% of the respondents that opted for pension scheme also agree (or strongly agree) with the statement, with only 3.7% disagreeing (or strongly disagree) and 5.9% remaining neutral. Although, those that are in the non-pension scheme also agree (or strongly agree) with the statement (87%), a higher proportion chose to remain neutral (10.2%) whilst those that disagree (or strongly disagree) amounts to 1%.

As to whether planning provides them with the financial support required for their remaining years (BC2), most of the respondents agree with the statement (52.6%). Only a small minority strongly disagree (0.4%). In total, 86.4% of the respondents agree (or strongly agree), while those who disagree (or strongly disagree) are less than 3%. 10.8% remain neutral. Similarly, 88.1% of the respondents that opted for pension scheme also agree (or strongly agree) with the statement, with only 2.2% disagreeing (or strongly disagree) and 9.7% remaining neutral. Although, those that are in the non-pension scheme also agree (or strongly agree) with the statement (85.4%), a higher proportion chose to remain neutral (11.4%) whilst those that disagree (or strongly disagree) amounts to 3.1%.

Most of the respondents (49.8%) agree that planning will help them maintain their quality of life during retirement (BC3). Only 0.4% strongly disagree with the statement. Similarly, in total, 84% agree (or strongly agree) with the statement. Only 13.2% are neutral, while less than 3% disagree (or strongly disagree) with the statement. Similarly, 86.4% of the respondents that opted for pension scheme also agree (or strongly agree) with the statement, with only 3.8% disagreeing and 9.7% remaining neutral. Although, many of the respondents from the non-pension scheme also agree (or strongly agree) with the statement (82.6%), a higher proportion chose to remain neutral (15.2%) whilst those that disagree and strongly disagree amounts to 2.2%.

Three statements are also developed to measure respondents' belief in self-control. 49.6% of the respondents agree with the statement that planning can help in resisting unnecessary spending while 0.8% strongly disagree (BSC1). In total, 85.4% agree (or strongly agree) that planning helps them maintain discipline when spending, while about 10.6% remain neutral. Only 4% disagree (or strongly disagree) with the statement. Similarly, 84.3% of the respondents that opted for pension scheme also agree (or strongly agree) with the statement, with only 4.3% disagreeing (or strongly disagree) and 11.4% remaining neutral. Meanwhile, those that are in the non-pension scheme also have similar opinions, where the agree (or strongly agree) are 86%, neutral are 10.2% and disagree (or strongly disagree) amounts to 3.8%.

As for the second statement on self-control 'Even though saving for retirement limits the money I have currently, the money I have saved and invested will benefit me after I leave the workforce' (BSC2), 48.2% agree with the statement while only about 0.6% strongly disagree. In total, those who agree (or strongly agree) were about 82%, with only 15% remaining neutral. Less than 3% disagree (or strongly disagree). Similarly, 83.7% of the respondents that opted for the pension scheme also agree (or strongly agree) with the statement, with only 2.2% disagreeing (or strongly disagree) and 14.1% remaining neutral. Meanwhile, those that are in the non-pension scheme are also in similar proportions, where the agree or strongly agree were 81.9%, neutral were 15.6% and disagree or strongly disagree amounts to 2.6%.

For the third statement 'Planning financially for retirement is a burden - it prevents me from enjoying my life now' (BSC3_r), a majority of 53.6% agree or strongly agree to it. Those that disagree or strongly disagree (24.8%) and neutral (21.6%) totals 46.4%. Similarly, 57.9% of the respondents that opted for pension scheme also agree or strongly agree with the statement, whilst those that disagree (24.3%) and neutral (17.8%) totals

42.1%. Meanwhile, those that are in the non-pension scheme are also in similar proportions, where the agree or strongly agree with the statement are 51.1%. The ones that are neutral (23.8%) and disagree or strongly disagree (25.1%) totals 48.9%. It can be seen that the proportion of those that agrees with the statements are almost similar to the ones that are neutral or disagree with the statement.

In summary, the overall mean of behavioural beliefs ranges from 3.43 to 4.23. The standard deviation ranges from 0.732 to 1.190. As for those who opted for pension scheme, the means ranges from 3.45 to 4.33 (standard deviation from 0.707 to 1.206) whilst those from the non-pension scheme, the mean ranges from 3.42 to 4.17 (standard deviation from 0.722 to 1.182). The results reflect that most respondents agree that planning will help them achieve autonomy and cope with life challenges. It will also help them in self-control and to foster discipline.

Behavioural beliefs	Pension		N	on-pension	Total		
items	Mean	Std. Deviation	Mean	Std. Deviation	Mean	Std. Deviation	
BA1	4.19	0.767	4.05	0.755	4.10	0.762	
BA2	4.22	0.744	4.11	0.722	4.15	0.732	
BA3	4.26	0.707	4.09	0.769	4.15	0.751	
BC1	4.33	0.776	4.17	0.757	4.23	0.768	
BC2	4.26	0.721	4.12	0.750	4.17	0.742	
BC3	4.23	0.777	4.10	0.754	4.15	0.765	
BSC1	4.19	0.842	4.15	0.774	4.16	0.799	
BSC2	4.19	0.755	4.11	0.787	4.14	0.776	
BSC3_r	3.45	1.206	3.42	1.182	3.43	1.190	

Table 4.6: Descriptive statistics for behavioural beliefs

(b) Normative beliefs

Normative beliefs refer to the perceived behavioural expectations of important referents to the population of study. The combination of a person's motivation to comply with these referents will determine subjective norms (Ajzen, 2019).

In this study, four groups of important referents are identified from the preliminary interviews. They are spouses, immediate family (siblings, parents, children), friends and relatives (family members other than spouse, parents, children, siblings). These groups play a role in influencing the respondents, either through their expectations (injunctive normative beliefs) or their actions (descriptive normative beliefs). However, for some individuals these important referents may not be applicable at all.

For injunctive normative beliefs of the referent spouses (INB1), it is applicable to those that are married. For this sample, 25% of the respondents are single (comprising of 8% from the pension group and 17% from the non-pension group). Most of the respondents strongly agree that these are important referents (33.6%), where only 1.8% disagree. Those who agree or strongly agree that their spouses think they should plan for retirement are 60.6%, while 10.2% are neutral. Only 4.2% disagree or strongly agree. For those in the pension scheme, 63.8% of the respondents agree or strongly agree with the statement, with only 7.6% disagreeing (or strongly disagree) and 5.9% remaining neutral. However, for those that are in the non-pension scheme only 58.7% agree (or strongly agree) with the statement, with a higher proportion choosing to remain neutral (12.7%) whilst those that disagree (or strongly disagree) amounts to 2.3%.

As for immediate family (siblings, parents, children) (INB2), this group is influential to a majority to the respondents (97.8%). Only 0.6% of the pension group and 1.6% of the non-pension group reported that this group is not applicable to them. About 39.4% agree these are important persons while only 2.2% strongly disagree. In total, those who agree or strongly agree that these are important referents to them are 72.8%, while only 18% remain neutral. 7% disagree or strongly disagree. For those in the pension scheme, 76.2% of the respondents agree or strongly agree with the statement, with only 9.7% disagreeing (or strongly disagree) and 12.4% remaining neutral. However, for those that are in the nonpension scheme only 70.8% agree or strongly agree with the statement, with a higher proportion choosing to remain neutral (21.3%) whilst those that disagree or strongly disagree amounts to 5.4%.

As for friends (INB3), this group is not applicable to only 6% of the respondents (or 1% of the pension group and 5% of the non-pension group). A majority of the respondents are neutral (31.4%), while 8% deemed them not influential. In total, those who agree or strongly agree that friends are important referents total 41.8% while 20.8% disagree or strongly disagree with the statement. For those in the pension scheme, only 47% of the respondents agree or strongly disagree and 30.3% remaining neutral. However, for those that are in the non-pension scheme, only 38.7% agree or strongly agree with the statement, with the neutral (38.7%) and those that disagree or strongly disagree (21.3%) amounts to 53.4%. Overall, most of the respondents are either neutral or disagree that this group influence their intentions to plan financially for retirement.

Relatives (family members other than spouse, parents, children, siblings) have lesser influence on the respondents (INB4). This group is not applicable to 9.2% of the respondents (or 2.4% of the pension group and 6.8% of the non-pension group). Most respondents are neutral about the statement (30.6%) while 9.2% strongly disagree. In total, 36.6% of the respondents reported that they agree or strongly agree that they are influence by what their relatives think they should do about retirement planning. About 23.6% disagree or strongly disagree with the statement. For those in the pension scheme, only 40.5% of the respondents agree and strongly agree with the statement, with only 22.1% disagreeing and 30.8% remaining neutral. However, for those that are in the non-pension scheme only 34.2% agree or strongly agree with the statement, with the neutral (30.5%) and those that disagree and strongly disagree (24.4%) totalling 54.9%. Although, most of

the respondents agree with the statement, and almost equal proportion are neutral or disagree.

Sections were also provided to respondents to list the important people other than the ones listed. Less than 14% responded with lists of other important people in their lives, which include financial services providers, reading sources, role models, the community they live in and people in their work environment.

In summary, the mean results are between 3.19 to 4.17 with standard deviation between 0.962 to 1.173. As for those who opted for pension scheme, the means ranges from 3.31 to 4.08 (standard deviation from 0.986 to 1.138) whilst those from the non-pension scheme the mean ranges from 3.57 to 3.95 (standard deviation from 0.943 to 1.025). Most respondents are agreeable that the important persons listed are influential in their lives. However, as not all the important persons listed are applicable to all respondents, this element of missingness has somewhat influence the results.

For descriptive normative beliefs of spouses (DNB1), this group is not applicable to 26.4% of the respondents (or 8.4% of the pension group and 18% of the non-pension group). From those that have responded, 28% agree that they are influence by what these important referents do, with only a minority that strongly disagree (2.4%). Those who agree or strongly agree are 51.6%, while 15.8% are neutral. Only 6.2% disagree or strongly disagree. For those in the pension scheme, 52.4% of the respondents agree or strongly agree with the statement, with only 7.1% disagreeing (or strongly disagree) and 17.8% remaining neutral. However, for those that are in the non-pension scheme only 51.1% agree or strongly agree with the statement, with 14.6% choosing to remain neutral whilst those that disagree or strongly disagree amounts to 5.7%.

As for immediate family (siblings, parents, children) (DNB2), this group is influential to a majority to the respondents (97.4%). Only 0.6% of the pension group and 2% of the non-pension group reported that this group is not applicable to them. About 41.6% agree these are important referents while only 3.2% strongly disagree. In total, those who agree or strongly agree that they are important referents in planning for retirement total 67.8%, while only 21.4% remain neutral. 8.2% disagree or strongly agree. For those in the pension scheme, 66.5% of the respondents agree or strongly agree with the statement, with only 10.8% disagreeing (or strongly disagree) and 21.1% remaining neutral. However, for those that are in the non-pension scheme only 68.6% agree and strongly agree with the statement, with 21.6% choosing to remain neutral whilst those that disagree or strongly disagree amounts to 6.7%.

As for friends (DNB3), this group is not influential to 5.8% of the respondents (or 1% of the pension group and 4.8% of the non-pension group). Most of the respondents agree with the statement (36.2%), while only 3% are not influence by these referents' actions. In total, those who agree or strongly agree that friends are important referents total 52.2% while 7.4% disagree or strongly disagree with the statement. 34.6% remain neutral. For those in the pension scheme, only 52.4% of the respondents agree or strongly agree with the statement, with only 8.6% disagreeing (or strongly disagree) and 36.2% remaining neutral. However, for those that are in the non-pension scheme only 52.1% agree or strongly agree with the statement, with the statement, with the neutral (33.7%) and those that disagree or strongly disagree (6.7%) amounts to 40.4%.

Relatives (family members other than spouse, parents, children, siblings) are also influential to the respondents (DNB4). This group is not applicable to 8.4% of the respondents (or 1.6% of the pension group and 6.8% of the non-pension group). Most respondents are neutral about the statement (38.2%) while 3% strongly disagree. In total,

45.8% of the respondents reported that they agree or strongly agree that they are influence by what their relatives do for retirement planning. About 7.6% disagree or strongly disagree with the statement. For those in the pension scheme, only 46.5% of the respondents agree or strongly agree with the statement, with only 8.1% disagreeing (or strongly disagree) and 41.1% remaining neutral. However, for those that are in the non-pension scheme only 45.4% agree and strongly agree with the statement, with the neutral (36.5%) and those that disagree and strongly disagree (7.3%) amounts to 43.8%.

In summary, the mean results for injunctive normative beliefs are between 3.19 to 4.17 with standard deviation between 0.962 to 1.173. As for those who opted for pension scheme, the means ranges from 3.31 to 4.08 (standard deviation from 0.986 to 1.138) whilst those from the non-pension scheme the mean ranges from 3.57 to 3.95 (standard deviation from 0.943 to 1.025). Meanwhile, the mean results descriptive normative beliefs are between 3.56 to 3.90 with standard deviation between 0.929 to 1.014. As for those who opted for pension scheme, the means ranges from 3.54 to 3.84 (standard deviation from 0.904 to 1.032) whilst those from the non-pension scheme the non-pension scheme the mean ranges from 3.57 to 3.95 (standard deviation from 0.943 to 1.025). From the results, most respondents are agreeable that they are influence by the expectations and actions of the important persons in their lives. However, as not all the important persons listed are applicable to all respondents, this have somewhat influenced the overall results.

Normative		Pensior	1	Ν	Non-Pens	ion		Total	
beliefs items	Missing Value	Mean	Std. Deviation	Missing Value	Mean	Std. Deviation	Missing Value	Mean	Std. Deviation
Injunctive no	ormative be	liefs							
INB1	42	4.08	1.095	90	3.95	1.025	125	4.17	0.971
INB2	3	4.02	0.986	10	3.89	0.955	11	3.99	0.962
INB3	5	3.36	1.102	24	3.65	0.944	30	3.28	1.128
INB4	12	3.31	1.138	34	3.57	0.943	46	3.19	1.173
Descriptive	normative	<u>beliefs</u>							
DNB1	42	3.84	0.998	90	3.95	1.025	132	3.90	1.014
DNB2	3	3.79	1.032	10	3.89	0.955	13	3.85	0.984
DNB3	5	3.56	0.904	24	3.65	0.944	29	3.61	0.929
DNB4	8	3.54	0.948	34	3.57	0.943	42	3.56	0.944

 Table 4.7: Descriptive statistics for Normative beliefs

(c) Control beliefs

Control beliefs reflects the beliefs of individuals on the perceived presence of factors that may facilitate or impede the performance of a behaviour. The combination of the perceived power of each control factor will determine the prevailing perceived behavioural control (Ajzen, 2019).

During the preliminary interviews, competence, resources and risk are identified as the control beliefs for this study. Three statements document the responses pertain impediments in regards to competence. For the first statement (CC1_r), a majority of 37.8% agree that that one of the impediments is the reliability of the information received. Only 3.8% strongly disagree. In total, 44% of the respondents agree or strongly agree with the statement, while 30.8% are neutral. Meanwhile, 25.2% disagree or strongly disagree with the statement and feels that the information received are reliable and useful for planning (CC1_r). Similarly, 43.8% of the respondents that opted for pension scheme also agree or strongly agree with the statement, with only 29.2% disagreeing and 27% remaining neutral. Although, those that are in the non-pension scheme also agree or strongly agree with the statement (44.1%), a higher proportion chose to remain neutral (33%) whilst those that disagree amounts to 22.9%.

As for whether the differing viewpoints and financial products received is an impediment to planning decisions (CC2_r), 41.4% agree with the statement, with only 2% that strongly disagree. In total, 49% of the respondents agree or strongly agree with the statement. 30.4% remain neutral, with 20.6% disagree or strongly disagree with the statement. Similarly, 43.7% of the respondents that opted for pension scheme also agree or strongly agree with the statement, with only 25.9% disagreeing (or strongly disagree) and 30.3% remaining neutral. Those that are in the non-pension scheme also agree or strongly agree with the statement (52.1%), with 30.5% choosing to remain neutral whilst those that disagree or strongly disagree amounts to 17.5%.

The third statement pertains effort to learn financial management skills (CC3_r). 49% agree that a lot of effort is required to learn these skills. Only 2% strongly disagree. In total, 64.2% agree or strongly agree that it takes a lot of effort to learn these skills, while 20% remain neutral. Less than 16% disagree or strongly disagree with the statement. Similarly, 63.2% of the respondents that opted for pension scheme also agree or strongly agree with the statement, with only 18.9% disagreeing (or strongly disagree) and 17.8% remaining neutral. Those that are in the non-pension scheme also agree or strongly agree with the statement (64.8%), with 21.3% choosing to remain neutral whilst those that disagree or strongly disagree amounts to 14%.

Five statements have been developed pertaining to challenges respondents faced while managing resources. The first statement is on the difficulty of managing expenses and commitments when planning (CR1_r). A majority of 31.6% agree that it is difficult to manage expenses and commitments when planning, while only 6% strongly disagree. In total, 39.2% agree or strongly agree that managing expenses and commitments is a challenge while 26.4% are neutral. About 34.4% disagree or strongly disagree and do not find it difficult to manage their expenses and commitments. For those in the pension

scheme, both the agree (or strongly agree) equals those that disagree (strongly disagree) (37.3%). The remainder 25.4% are neutral. As for those in the non-pension scheme, 40.3% agree or strongly agree with the statement, with 27% choosing to remain neutral whilst those that disagree or strongly disagree amounts to 32.7%.

Managing resources also involve setting aside extra income for retirement purposes (CR2_r). 40.8% find this difficult, with only 3.4% having no problem at all. In total, 54% agree and strongly agree that this is a difficult endeavour, with 25.4% remained neutral. Only 20.6% disagree or strongly disagree, and find it easy to set aside extra income for retirement purposes. Similarly, 50.2% of the respondents that opted for pension scheme also agree or strongly agree that it is difficult, with only 26.4% disagreeing (or strongly disagree) and 23.2% remaining neutral. Those that are in the non-pension scheme, 56.2% also agree or strongly agree that it is difficult, with 26.7% choosing to remain neutral whilst those that disagree and strongly disagree amounts to 17.2%.

When it comes to the stability of their income source (CR3_r), 31.8% are neutral, with 4.6% strongly disagree that the stability of their income is a challenge. In total, those who agree or strongly agree that the stability of their income source is a problem are 39% while 29% disagree or strongly disagree and do not face such difficulty. Similarly, 32.9% of the respondents that opted for pension scheme also agree and strongly agree that they face difficulties, with only 34% disagreeing and 33% remaining neutral. Those that are in the non-pension scheme, 42.6% also agree and strongly agree with the statement, with 31.1% choosing to remain neutral whilst those that disagree and strongly disagree amounts to 26.4%.

As to the challenge of being aware of all the financial and retirement benefits when planning, 39.8% agree that it is difficult (CR4_r). Only 2.8% strongly disagree. In total, 47.2% of the respondents agree or strongly agree that it is difficult to be aware of all the financial and retirement benefits that are important to any planning efforts while 29.2% remain neutral. Only 23.6% disagree or strongly disagree, and are of the opinion that they are aware of the opportunities that will aid them in planning. Similarly, 40.5% of the respondents that opted for pension scheme also agree or strongly agree with the statement, with only 32.4% disagreeing (or strongly disagree) and 27% remaining neutral. Those that are in the non-pension scheme, 51.1% also agree or strongly agree with the statement, with 30.5% choosing to remain neutral whilst those that disagree or strongly disagree amounts to 18.5%.

For the final statement on managing resources, 40.8% finds it difficult to be flexible with the lifestyle changes required in financial retirement plans, with about 3.2% strongly disagree (CR5_r). In total, 50.2% find it challenging to be flexible with the lifestyle changes required, while 25% are neutral. The remainder (24.8%) either disagree or strongly disagree that it is difficult to be flexible with the lifestyle changes required. Similarly, 48.6% of the respondents that opted for pension scheme also agree or strongly agree with the statement, with only 27.6% disagreeing (or strongly disagree) and 23.8% remaining neutral. Those that are in the non-pension scheme, 51.1% also agree or strongly agree with the statement, with 25.7% choosing to remain neutral whilst those that disagree or strongly disagree amounts to 23.2%.

Three statements are developed pertaining to the risks faced when planning. For the first statement, about 39.4% agree that they find it hard to gauge areas which may bring financial risk to the planning efforts made (CRS1_r). Only about 3.2% strongly disagree with the statement. In total, 46.4% agree or strongly agree that it is hard to gauge areas which may

bring financial risk, with 33% remaining neutral. 20.6% reported they disagree or strongly disagree. Similarly, 44.9% of the respondents that opted for pension scheme also agree or strongly agree with the statement, with only 24.3% disagreeing (or strongly disagree) and 30.8% remaining neutral. Those that are in the non-pension scheme, 47.3% also agree or strongly agree with the statement, with 34.3% choosing to remain neutral whilst those that disagree or strongly disagree amounts to 18.4%.

For the second statement, about 31.4% agree that they are confident of what will happen to their investments in the future (CRS2). Only about 4% strongly disagree with the statement. In total, 36.8% agree or strongly agree with the statement, with 40% remaining neutral. 36.8% reported they disagree or strongly disagree. Similarly, 40.5% of the respondents that opted for pension scheme also agree or strongly agree with the statement, with only 21.6% disagreeing (or strongly disagree) and 37.8% remaining neutral. Those that are in the non-pension scheme, 34.6% also agree or strongly agree with the statement, with 41.3% choosing to remain neutral whilst those that disagree or strongly disagree amounts to 24.1%.

Finally, 33.2% agree that planning is difficult because the future is uncertain (CRS3_r), with only 3.2% strongly disagree. In total, 43.8% agree or strongly agree that planning is difficult. However, 32% remain neutral with less than 24.2% disagree or strongly disagree with the statement. As for the respondents that opted for pension scheme, 38.4% agree or strongly agree with the statement, with only 28.1% disagreeing (or strongly disagree) and 33.5% remaining neutral. Those that are in the non-pension scheme, 47% also agree or strongly agree with the statement, with 31.1% choosing to remain neutral whilst those that disagree or strongly disagree amounts to 21.9%.

In summary, the mean results are between 2.71 to 3.62 with standard deviation between 0.928 to 1.069. As for those who opted for pension scheme, the means ranges from 2.45 to 3.23 (standard deviation from 0.906 to 0.842) whilst those from the non-pension scheme the mean ranges from 2.34 to 3.10 (standard deviation from 0.906 to 1.068).

Control beliefs	I	Pension		on-Pension		Total
items	Mean	Std. Deviation	Mean Std. Deviation		Mean	Std. Deviation
CC1_r	2.87	0.997	2.74	0.959	3.21	0.974
CC2_r	2.79	0.958	2.58	0.911	3.34	0.933
CC3_r	2.45	1.016	2.34	0.939	3.62	0.969
CR1_r	3.01	1.068	2.89	1.068	3.06	1.069
CR2_r	2.70	1.035	2.49	1.020	3.43	1.029
CR3_r	2.98	1.008	2.78	1.050	3.14	1.038
CR4_r	2.92	1.035	2.60	0.906	3.28	0.968
CR5_r	2.76	1.016	2.64	1.013	3.32	1.015
CRS1_r	2.79	0.990	2.66	0.926	2.71	0.951
CRS2	3.23	0.894	3.10	0.946	3.15	0.928
CRS3_r	2.85	0.994	2.66	1.017	2.73	1.012

Table 4.8: Descriptive statistics for control beliefs

(d) Attitudes

Attitudes reflects individual's response, where they evaluate a behaviour with some degree of favourableness or unfavourableness. In an overall evaluation of how respondents feel about making financial preparations for retirement (DA1_r), most respondents rate an overall 3 in the semantic differential scale. Only a minority (1.8%) rate it as bad (Scale 1). Those that rate it as good (Scale 4 or 5) totals 61.6% with 29.4% being neutral and 9% felt that it is bad (Scale 1 or 2). Similarly, for the pension group, 66.5% rate it as good (Scale 4 or 5), 29.4% neutral and 9% bad (Scale 1 or 2). As for the non-pension group, 58.7% rate it as good (Scale 4 or 5), 33% neutral and 8.3% bad (Scale 1 or 2).

In a more in-depth evaluation, respondents are also asked further on the experiential part of their evaluation, that is whether planning for retirement is an enjoyable or interesting experience (DA2_r). 39% have neutral feelings, rating it 3, while a minority (1.6%) rate it uninteresting (Scale 1). In total, most of the respondents (51%) rated it interesting and enjoyable (Scale 4 or 5), with 10% rating it as uninteresting (Scale 4 or 5). Similarly, this is reflected in both the pension and non-pension groups. For the pension group, 57.3% rate it as enjoyable (Scale 4 or 5), 31.9% neutral and 10.8% bad (Scale 1 or 2). As for the non-pension group, 47.3% rate it as good (Scale 4 or 5), 43.2% neutral and 9.6% bad (Scale 1 or 2).

Respondents were also asked another experiential question (DA5) that is whether the experience was pleasant or unpleasant. 34.6% have neutral feelings, rating it 3, while a minority (7.2%) rate it unpleasant (Scale 1). In total, most of the respondents (49.6%) rated it pleasant (Scale 4 or 5), with 15.8% rate it as unpleasant (Scale 4 or 5). Similarly, this is reflected in both the pension and non-pension groups. For the pension group, 51.9% rate it as pleasant (Scale 4 or 5), 26.5% neutral and 21.6% bad (Scale 1 or 2). As for the non-pension group, 48.2% rate it as good (Scale 4 or 5), 39.4% neutral and 12.4% bad (Scale 1 or 2).

As for the instrumental part of the evaluation (DA4_r), respondents were required to rate retirement planning in a scale, based on whether it is beneficial or harmful. Most of the respondents (42.6%) rate it as beneficial. Only a minority (2.2%) thinks it is harmful. In total, those who rate it as 1 or 2 totals 68.8%, while 21.2% have neutral feelings. 10% rate it 4 or 5, and find financial planning for retirement somewhat harmful. Similarly, this is reflected in both the pension and non-pension groups. For the pension group, 76.2% rate it as beneficial (Scale 4 or 5), 14.1% neutral and 9.7% harmful (Scale 1 or 2). As for the

non-pension group, 64.4% rate it as beneficial (Scale 4 or 5), 25.4% neutral and 10.2% harmful (Scale 1 or 2).

Another question was also asked on the instrumental part of the evaluation (DA3), that is whether retirement planning is useful or not. Most of the respondents (30%) rate it as useful. Only a minority (9.2%) thinks its worthless (Scale 2). In total, those who rate it as 4 or 5 totals 54.6%, while only 22.2% have neutral feelings. 23.2% rate it 1 or 2, where they find financial planning for retirement somewhat worthless. For the pension group, 54.1% rate it as useful (Scale 4 or 5), 16.2% neutral and 29.8% worthless (Scale 1 or 2). As for the non-pension group, 54.9% rate it as useful (Scale 4 or 5), 25.7% neutral and 19.4% worthless (Scale 1 or 2). A higher percentage of the pension group has deemed it worthless.

In summary, the mean results range from 3.47 to 3.99 with standard deviation between 0.982 to 1.37. As for those who opted for pension scheme, the means ranges from 3.34 to 4.16 (standard deviation from 1.062 to 1.535) whilst those from the non-pension scheme the mean ranges from 2.11 to 3.81 (standard deviation from 0.927 to 1.259). Most respondents are agreeable that financial planning for retirement is good, an interesting experience and beneficial to them.

Attitudes	ŀ	Pension		on-Pension	Total		
items	Mean	Std. Deviation	Mean	Mean Std. Deviation		Std. Deviation	
DA1_r	4.03	1.130	3.81	1.004	3.89	1.056	
DA2_r	3.74	1.062	3.56	0.927	3.63	0.982	
DA3	3.34	1.535	3.55	1.259	3.47	1.370	
DA4_r	4.16	1.081	2.11	1.056	3.99	1.072	
DA5	3.43	1.301	3.52	1.038	3.49	1.142	

 Table 4.9: Descriptive statistics for attitudes

(e) Subjective Norms

Subjective norms are the perceived social pressure to engage or not to engage in a behaviour. Social pressure comes from the social environment of a person. It has an injunctive and descriptive component. Injunctive norms are concerned with gaining approval from important groups or individuals whereas descriptive norms are related to one's perception of what important others do in relations to the behaviour of study (Ajzen, 2019).

Two statements are developed to measure injunctive norms. The first is 'I feel the pressure to plan financially for retirement' (IN1). 33% agree that they feel the social pressure to plan financially for retirement, while only about 8.6% strongly disagree. In total, those who agree and strongly agree total 42.6%, while 32% reported neutral feelings. 25.4% do not feel the pressure to plan. However, 38.3% of the respondents that opted for pension scheme agree and strongly agree with the statement, with another 31.9% disagreeing while 29.7% remaining neutral. As for those that are in the non-pension scheme, 45.1% agree and strongly agree with the statement, while 33.3% choose to remain neutral and those that disagree and strongly disagree amounts to 21.6%.

The second statement on injunctive norms is 'Most people who are important to me think I should plan financially for retirement' (IN2). 48.8% agrees with the statement, with 2.2% strongly disagree. In total, those who agree or strongly agree total 61.8% with about 26.8% being neutral. A minority (11.4%) disagree that important people in the lives thinks they should plan financially for retirement. Similarly, 60.6% of the respondents that opted for pension scheme also agree or strongly agree with the statement, with only 13.5% disagreeing (or strongly disagree) and 25.9% remaining neutral. Although, those that are in the non-pension scheme also agree or strongly agree with the statement (62.5%), a higher

proportion chose to remain neutral (27.3%) whilst those that disagree or strongly disagree amounts to 10.2%.

As for descriptive norms, for the statement 'Most people who are important to me plan financially for retirement' (DN1), 41.8% agree with the statement. Only 1.2% strongly disagree. In total, 52% agree or strongly agree that important people in their lives plan financially for retirement whereas 38.2% are neutral. Less than 10% disagree or strongly disagree with the statement. Although, 48.7% of the respondents that opted for pension scheme also agree or strongly agree with the statement, about 41.6% choose to remain neutral, with only 9.7% disagreeing. A wider majority of those in the non-pension scheme agree or strongly agree with the statement (54%), with 36.2% choose to remain neutral whilst those that disagree or strongly disagree amounts to 9.9%.

In summary, the mean results for injunctive norms is 3.18 and 3.61 with standard deviation of 1.093 and 0.903, respectively. For those who opted for pension scheme, the means is 3.05 and 3.54 (standard deviation 1.192 and 0.938, respectively) whilst those from the non-pension scheme the mean is 3.26 and 3.66 (standard deviation 1.025 and 0.88, respectively). Most respondents are agreeable that they feel the social pressure to plan financially for retirement.

As for descriptive norms, the mean result is 3.51 with a standard deviation of 0.836. Those who opted for pension scheme, the means is 3.49 and standard deviation 0.822 whilst those from the non-pension scheme the mean is 3.53 and standard deviation 0.846. Most respondents are agreeable that important people in their lives are also planning financially for retirement.

Subjective norms]	Pension	Non	-Pension	Total		
items	Mean	Std.	Mean	Std.	Mean	Std.	
		Deviation		Deviation		Deviation	
IN1	3.05	1.192	3.26	1.025	3.18	1.093	
IN2	3.54	0.938	3.66	0.880	3.61	0.903	
DN1	3.49	0.822	3.53	0.846	3.51	0.836	

Table 4.10: Descriptive statistics for subjective norms

(f) Perceived Behavioural Control

Perceived behavioural control refers to the extent to which a person believe that they are capable and have control over their ability to perform a given behaviour (Ajzen, 2019). It takes into account a person's perception of the availability of information, skills, opportunities and resources required to perform the behaviour as well as possible barriers or obstacles that may have to be overcome.

In terms of self-efficacy, a majority of 39.6% are confident that they can effectively plan financially for retirement (DPBC1). Only a minority of 2.8% strongly disagree with the statement. In total, those who agree or strongly confident of their planning abilities totals 45.8%. At the same time, 39.2% are neutral on their confidence level. 15% have a low level of confidence in their effectiveness to plan. Similarly, 51.3% of the respondents that opted for pension scheme also agree or strongly agree with the statement, with only 15.1% disagreeing (or strongly disagree) and 33.5% remaining neutral. Although, those that are in the non-pension scheme also agree or strongly agree with the statement (42.5%) and equivalent proportion also choose to remain neutral. Those that disagree or strongly disagree amounts to 14.9%.

As for whether it is easy for the respondents to plan financially for retirement or not (DPBC2), 37.8% of the respondents are neutral, with only a small proportion (4.2%) strongly agreeing with the statement. In total, 25.4% agree that it is easy, while 36.8% feel it is hard. Those that opted for pension scheme are almost of similar proportions with 31.9%

of the respondents agree or strongly agree that planning for retirement is easy, with only 31.9% disagreeing and 36.2% remaining neutral. For those that are in the non-pension scheme, those that agree or strongly agree with the statement are lower (21.6%), with 38.7% remaining neutral whilst those that disagree or strongly disagree amounts to 39.6%. A higher proportion seems to deem planning for retirement hard.

As for whether the decision to plan financially for retirement is beyond control (DPBC3_r), most respondents are neutral (34.8%), with a minimal of 4.2% that feels it is beyond their control. Overall, about 43% feels that retirement planning is still within their control, whilst 22.2% feel that the decision to plan is beyond their control. Most of those that opted for pension scheme disagree or strongly disagree with the statement, with 38.4% remaining neutral and 17.8% agreeing that retirement planning is beyond their control. For those that are in the non-pension scheme, those that disagree or strongly disagree with the statement are 42.6%, with 32.7% remaining neutral whilst those that agree and strongly agree amounts to 24.8%.

As for whether it is up to the individual to plan financially for retirement or not (DPBC4), 49.4% agree to the statement with only a small proportion (1.6%) strongly disagreeing with the statement. In total, 68.2% agree that they are in control, while 21.4% are neutral and 10.4% thinks retirement planning is not up to them. Most of the respondents that opted for pension scheme agree or strongly agree with the statement (70.8%), with only 12.4% disagreeing (or strongly disagree) and 16.8% remaining neutral. For those that are in the non-pension scheme, 66.7% of the respondents agree or strongly agree with the statement, with 24.1% remaining neutral whilst those that disagree or strongly disagree amounts to 9.3%.

In summary, the mean results for perceived behavioural control ranges from 2.85 and 3.75 with standard deviation of 0.873 to 0.99. As for those who opted for pension scheme, the means ranges from 3.02 to 3.76 (standard deviation from 0.869 to 0.973) whilst those from the non-pension scheme the mean ranges from 2.76 to 3.75 (standard deviation from 0.874 to 1.035). Most respondents are neutral or agreeable to this statement.

PBC	Pension		Ν	on-Pension	Total		
items	Mean	lean Std. Deviation		Std. Deviation	Mean	Std. Deviation	
DPBC1	3.42	0.869	3.3	0.874	3.34	0.873	
DPBC2	3.02	0.958	2.76	0.978	2.85	0.978	
DPBC3_r	3.29	0.910	3.23	1.035	3.25	0.990	
DPBC4	3.76	0.973	3.75	0.881	3.75	0.915	

 Table 4.11: Descriptive statistics for Perceived behavioural control

(g) Intentions

Intention is an indication of a person's readiness to perform a given behaviour, and it is considered to be the immediate antecedent of behaviour (Ajzen, 2019). Three items are used to measure intentions. The first item measures the respondents desire to act (DINT1). Most of the respondents agree that they want to plan (47.4%). 2.2% strongly disagree to the statement. In total, 77% agree or strongly agree that they want to plan financially for retirement. 17% are neutral while 5.8% disagree or strongly disagree with the statement. Similarly, 76.7% of the respondents that opted for pension scheme also agree or strongly agree with the statement, with only 6.5% disagreeing (or strongly disagree) and 16.8% remaining neutral. Those that are in the non-pension scheme also agree or strongly agree with the statement (77.2%), with 17.5% remain neutral whilst those that disagree or strongly disagree amounts to 5.4%.

For the second measure of intentions, 47.2% agree that they have intentions to plan, with only 2.8% strongly disagreeing with the statement (DINT2). In total, 72.8% agree or strongly agree that they have intentions to plan while 7.4% reported that they disagree or strongly disagree with the statement. 19.8% are neutral about their intentions. Similarly, 74.1% of the respondents that opted for pension scheme also agree or strongly agree with the statement, with only 7.5% disagreeing (or strongly disagree) and 18.4% remaining neutral. Those that are in the non-pension scheme also agree or strongly agree with the statement (72.1%), with 20.6% remaining neutral whilst those that disagree or strongly disagree amounts to 7.3%.

As for whether they will try to plan or not (DINT3), 44.8% agree that they will try to plan, with 4.2% strongly disagree. In total, 69% agree or strongly agree that they will try to plan, while 21.2% are neutral. 9.8% are not willing to try. Similarly, 70.2% of the respondents that opted for pension scheme also agree or strongly agree with the statement, with only 7.6% disagreeing (or strongly disagree) and 22.2% remaining neutral. 68.3% of the respondents in the non-pension scheme also agree or strongly agree with the statement, with 20.6% choosing to remain neutral whilst those that disagree or strongly disagree amounts to 11.1%.

In summary, the mean results for intentions ranges from 3.79 to 3.99. The standard deviation is from 0.901 to 1.007. For those who opted for pension scheme, the means ranges from 3.86 to 3.98 (standard deviation from 0.878 to 0.951) whilst those from the non-pension scheme the mean ranges from 3.75 to 3.99 (standard deviation from 0.915 to 1.039). Most respondents have intentions and will try to plan financially for retirement.

Intentions	Pe	nsion	Non-pen	ision	Total		
items	Mean	Std.	Mean	Std.	Mean	Std.	
		Deviation		Deviation		Deviation	
DINT1	3.98	0.878	3.99	0.915	3.99	0.901	
DINT2	3.91	0.911	3.86	0.953	3.88	0.937	
DINT3	3.86	0.951	3.75	1.039	3.79	1.007	

Table 4.12: Descriptive statistics for intentions

4.4 Partial Least Square (PLS) SEM Assessment

The data collected is further analysed using PLS. A two-step process will be used. The first step would involve assessing the measurement model. Once the data satisfy the requirement of the measurement model, the structural model is evaluated (Hair et al., 2017).

4.5 Measurement Model Assessment

In the previous section, EFA was conducted. The results reveal an 8-factor structure and recommended the removal of 6 items (DA3, DA5, DPBC3_r, DPBC4, IN1, BSC3_r,). In this section, the initial theoretical framework is tested using PLS-SEM (SmartPLS 3.0). The objective is to assess the psychometric properties of the measurement model and estimates the parameters of the structural model. The research model comprise of reflective indicators, hence the testing of the measurement model would encompass assessing its convergent validity, internal consistency reliability and discriminant validity (Gefen et al., 2000). The results would provide a level of assurance of the validity and reliability of the constructs in the theoretical framework.

4.5.1 Convergent Validity

To determine convergent validity, two measures are examined. Firstly, outer loadings are evaluated to ascertain indicator reliability. The threshold level is to accept the indicators with loadings above 0.707. The results of the analysis, available in the table below, shows that a major proportion of the constructs display outer loadings above 0.707.

However, there are a few indicators that are between 0.40 to 0.707. Chin (1998b) have advised that indicators such as these are to be removed only if it results in improvements in composite reliability (CR) and average variance extracted (AVE), as they may still have content validity. These improvements should result in CR level to be at least 0.60 while the acceptable threshold level for AVE is 0.50.

Based on these guidelines, all constructs have CR above 0.60; with three constructs having AVE below 0.50 (Control Beliefs, Attitudes, PBC). From these constructs, a total of three items are identified for removal, one each from Control Beliefs (CRS2), attitudes (DA3) and PBC (DPBC4). These items have both low loadings (below 0.707) and AVE (below 0.50) (Bagozzi et al., 1991; Hair et al., 2011). Another indicator from behavioural beliefs (BSC3_r) have loadings below 0.40 and is also identified for removal.

Upon removal of these items, all items load well and are within acceptable threshold levels. Results of the indicators (after running the bootstrapping procedure) are available in Appendix O. Indicator loadings are significant at 0.05, indicating that the overall model is preserved.

Constructs	Items		Initial model		Revised model		
	codes	Outer	Composite	AVE	Outer	Composite	AVE
		loading	Reliability		loading	Reliability	
Attitudes	DA1_r	0.779	0.812	0.473	0.832	0.842	0.578
	DA2_r	0.795			0.844		
	DA3	0.428					
	DA4_r	0.755			0.798		
	DA5	0.612			0.522		
Subjective	DN1	0.835	0.772	0.547	0.835	0.772	0.547
Norms	IN2	0.859			0.859		
	IN1	0.454			0.454		
Injunctive	IN1	0.627	0.753	0.612	0.627	0.753	0.612
Norms	IN2	0.911			0.911		
Descriptive	DN1	1.000	1.000	1.000	1.000	1.000	1.000
PRC	DPRC1	0 795	0.685	0 395	0.747	0.753	0.505
FBC	DIDCI	0.795	0.005	0.595	0.747	0.755	0.505
	DFDC2	0.778			0.737		
	DIDC3_I	0.304			0.044		
Intentions	DI BC4	0.155	0.024	0.802	0.864	0.024	0.802
intentions	DINT2	0.005	0.924	0.002	0.004	0.724	0.002
	DINT2	0.952			0.952		
Behavioural	BA1	0.807	0.921	0.578	0.800	0.937	0.651
Beliefs		0.845	0.921	0.570	0.849	0.957	0.001
Deneis	BA3	0.815			0.824		
	BC1	0.842			0.848		
	BC2	0.851			0.860		
	BC3	0.835			0.845		
	BSC1	0.663			0.664		
	BSC2	0.732			0.731		
	BSC3 r	0.255			01701		
Normative	DNB A	0.696	0.764	0.621	0.696	0.764	0.621
Beliefs	INB A	0.871			0.871		
Control	CC1 r	0.634	0.900	0.458	0.641	0.911	0.508
Beliefs	CC2 r	0.636			0.658		
	CC3 r	0.575			0.583		
	CR1 r	0.677			0.707		
	CR2 r	0.740			0.744		
	CR3 r	0.762			0.782		
	CR4 r	0.750			0.765		
	CR5 r	0.750			0.765		
	CRS1 r	0.743			0.745		
	CRS2	0.358					
	CRS3 r	0.712			0.711		
	_						

Table 4.13: Convergent validity analysis

4.5.2 Internal Consistency Reliability

Cronbach's alpha is traditionally the indicator used for the evaluation of reliability. Measures of reliability are important as it indicates the extent that a variable is consistent with what is measured. However, Cronbach's alpha is a more conservative measure, as it tends to underestimate reliability. In PLS-SEM, composite reliability (CR) is used. Values can range from 0 (completely unreliable) to 1 (perfectly reliable) (Straub, Boudreau & Gefen, 2004).

The threshold value for composite reliability is above 0.70 (Bagozzi & Yi, 1988; Gotz et al., 2010; Urbach & Ahlemann, 2010). The following table shows the CR for each construct used in this study. The Cronbach's alpha ranges from 0.40 to 1.00, while the CR is between 0.753 to 1.00. The rho A scores are between 0.439 to 1.00. The CR scores are above 0.70, which indicates that the constructs have internal consistency reliability.

Constructs	Cronbach's	rho_A	Composite
	Alpha		Reliability
Attitudes	0.739	0.741	0.842
Subjective Norms	0.568	0.663	0.772
Injunctive Norms	0.400	0.502	0.753
Descriptive Norms	1.000	1.000	1.000
PBC	0.519	0.484	0.753
Intentions	0.876	0.888	0.924
Behavioural Beliefs	0.922	0.926	0.937
Normative Beliefs	0.404	0.439	0.764
Control Beliefs	0.891	0.898	0.911

Table 4.14: Internal consistency reliability

4.5.3 Discriminant Validity

The first criterion to evaluate discriminant validity involves assessing the cross-loadings of the indicators. To establish discriminant validity, the loadings must load highly on its respective construct (Barclay et al., 1995; Chin, 1998b; Chin, 2010; Urbach & Ahlemann, 2010). The cross-loadings were examined across each row in the table. Using the crossloading table, each indicator has loaded higher on its respective constructs. This indicates that the indicators have discriminant validity.

				Cor	structs ¹				
Items	BB	СВ	ATT	INT	DN	SN	PBC	IN	NB
BA1	0.811	0.014	0.309	0.301	0.160	0.135	0.216	0.077	0.314
BA2	0.849	0.002	0.302	0.303	0.195	0.178	0.228	0.115	0.317
BA3	0.824	0.026	0.320	0.264	0.209	0.186	0.254	0.115	0.285
BC1	0.848	-0.010	0.292	0.257	0.217	0.190	0.229	0.114	0.352
BC2	0.860	0.000	0.328	0.239	0.260	0.219	0.268	0.124	0.305
BC3	0.845	-0.066	0.296	0.201	0.237	0.187	0.236	0.092	0.278
BSC1	0.664	-0.035	0.240	0.259	0.165	0.188	0.175	0.157	0.208
BSC2	0.731	0.045	0.286	0.292	0.217	0.181	0.231	0.101	0.245
CC1_r	0.041	0.641	0.134	0.037	-0.063	-0.100	0.278	-0.112	0.001
CC2_r	-0.075	0.658	0.074	-0.093	-0.110	-0.154	0.314	-0.157	-0.056
CC3_r	-0.121	0.583	0.013	-0.127	-0.082	-0.167	0.311	-0.201	-0.115
CR1_r	0.024	0.707	0.224	0.007	-0.017	-0.121	0.365	-0.184	-0.013
CR2_r	0.003	0.744	0.179	-0.076	-0.061	-0.192	0.371	-0.260	-0.080
CR3_r	0.035	0.782	0.166	-0.057	-0.058	-0.170	0.427	-0.228	0.004
CR4_r	-0.013	0.765	0.148	-0.067	-0.052	-0.128	0.347	-0.166	-0.017
CR5_r	0.006	0.765	0.164	-0.043	0.000	-0.127	0.380	-0.207	0.024
CRS1_r	-0.027	0.745	0.117	-0.059	-0.079	-0.155	0.324	-0.188	-0.038
CRS3_r	0.062	0.711	0.227	-0.012	-0.052	-0.159	0.468	-0.215	0.005
DA1_r	0.260	0.181	0.832	0.120	0.068	0.031	0.255	-0.011	0.100
DA2_r	0.269	0.216	0.844	0.194	0.087	0.087	0.245	0.059	0.134
DA4_r	0.316	0.127	0.798	0.117	0.007	0.035	0.207	0.046	0.180
DA5	0.259	0.110	0.522	0.171	0.133	0.116	0.165	0.070	0.134
DINT1	0.298	0.013	0.257	0.864	0.260	0.291	0.147	0.242	0.239
DINT2	0.316	-0.095	0.173	0.952	0.318	0.375	-0.029	0.327	0.319
DINT3	0.260	-0.100	0.107	0.868	0.253	0.319	-0.044	0.293	0.268
DN1	0.258	-0.078	0.097	0.311	1.000	0.835	0.139	0.482	0.346
DN1	0.258	-0.078	0.097	0.311	1.000	0.835	0.139	0.482	0.346
IN1	-0.089	-0.429	-0.116	0.128	0.137	0.454	-0.312	0.627	-0.015
IN2	0.217	-0.113	0.132	0.332	0.526	0.859	-0.054	0.911	0.279
DPBC1	0.284	0.262	0.214	0.069	0.232	0.080	0.747	-0.074	0.119
DPBC2	0.098	0.379	0.163	0.005	0.179	0.038	0.737	-0.096	0.021
DPBC3_r	0.244	0.408	0.236	0.001	-0.065	-0.148	0.644	-0.183	0.043
IN1	-0.089	-0.429	-0.116	0.128	0.137	0.454	-0.312	0.627	-0.015
IN2	0.217	-0.113	0.132	0.332	0.526	0.859	-0.054	0.911	0.279
INB_A	0.327	-0.104	0.154	0.257	0.276	0.297	0.021	0.235	0.871
DNB_A	0.229	0.081	0.139	0.233	0.278	0.203	0.122	0.086	0.696

Table 4.15: Cross loadings output

Note:

¹ List of abbreviations used

ATT – Attitude, SN - Subjective Norms, IN - Injunctive Norms, DN - Descriptive Norms, PBC - Perceived Behavioural Control, INT - Intentions, BB - Behavioural beliefs, NB - Normative beliefs, CB -Control beliefs

The second approach to assessing discriminant validity involves using the Fornell-Larcker criterion. The square root of the AVE of each construct is compared to the correlations of all other constructs. The correlation tables reveal that the diagonal elements are larger than the off-diagonal elements. These results confirm that each indicator have discriminate validity, and it correlates only with its own measures. Using the Fornell-Larcker criterion, the result confirms that the measures have discriminant validity.

Constructs ¹	ATT	BB	СВ	DN	IN	INT	NB	PBC	SN
ATT	0.760								
BB	0.369	0.807				5			
CB	0.211	-0.003	0.713						
DN	0.097	0.258	-0.078	1.000					
IN	0.057	0.137	-0.274	0.482	0.782				
INT	0.202	0.327	-0.067	0.311	0.321	0.896			
NB	0.185	0.359	-0.036	0.346	0.219	0.309	0.788		
PBC	0.290	0.286	0.513	0.139	-0.176	0.028	0.077	0.711	
SN	0.090	0.227	-0.208	0.835	0.884	0.368	0.323	-0.034	0.740
Note:							•	•	•

Table 4.16: Fornell-Larker's criteria

¹List of abbreviations used

ATT – Attitude, SN - Subjective Norms, IN - Injunctive Norms, DN - Descriptive Norms, PBC - Perceived Behavioural Control, INT - Intentions, BB - Behavioural beliefs, NB - Normative beliefs, CB -Control beliefs

The final approach to assessing discriminant validity involved the use of HTMT as a criterion. The HTMT values are compared with a predefined threshold level. For this study the threshold level is 0.85 or HTMT.⁸⁵. The HTMT values in the following table ranges from 0.086 to 0.518. All the values are below the threshold value and the results further support that satisfactory discriminant validity is achieved.

Constructs ¹	ATT	BB	СВ	DN	IN	INT	NB	PBC	SN
ATT									
BB	0.446								
СВ	0.256	0.088							
DN	0.115	0.269	0.086						
IN	0.286	0.318	0.568	0.663					
INT	0.248	0.365	0.107	0.330	0.490				
NB	0.335	0.574	0.205	0.550	0.471	0.518			
PBC	0.464	0.421	0.708	0.309	0.526	0.140	0.228		
SN	0.242	0.354	0.395	1.005	1.910	0.496	0.617	0.504	

Table 4.17: HTMT values

Note:

¹List of abbreviations used

ATT – Attitude, SN - Subjective Norms, IN - Injunctive Norms, DN - Descriptive Norms, PBC - Perceived Behavioural Control, INT - Intentions, BB - Behavioural beliefs, NB - Normative beliefs, CB -Control beliefs

In summary, the validity and reliability tests conducted has indicated that the measurement model has internal consistency reliability, convergent and discriminant validity. This model is valid and can be used for further testing in the structural model.

4.6 Structural Model Assessment

Upon validating the measurement models, the structural models were evaluated. This would provide support for the theoretical model developed. The structural model illustrates the relationships between constructs that were hypothesized. As the primary objective of the PLS SEM model is prediction, the strength of each structural path and the combined predictiveness (R^2) is evaluated (Chin, 1998b). Bootstrapping re-sampling procedures were also applied to examine the stability of the estimates. The bootstrap re-sampling procedure using 5,000 sub-samples was run in PLS SEM. The results were used to assess the significance levels of the PLS estimates.

4.6.1 Path Coefficients

Before assessing the path coefficients, a bootstrapping routine of 5,000 re-samples is used. The significance level is set at 0.05, which is in line with many social science studies (Hair et al., 2017). Subsequently, the path coefficients' direction, magnitude and significance are evaluated to assess the relationships between the independent and dependent variables.

Most of the path coefficients displayed the expected positive sign and is significant (p<0.05). Only PBC displayed a negative sign (-0.011) and did not have significant influence on intentions (p=0.842). Hence, this indicates that most of the theoretical assumptions about the model are supported and significant, except for the path from PBC to intentions.

The magnitude of the path coefficient is also evaluated. Each significant relationship has a path coefficient of more than 0.1 and therefore cannot be neglected (Sellin & Keeves, 1994; Urbach & Ahlemann, 2010). Two constructs are hypothesised to influence intentions. They are attitudes and subjective norms. Of these two constructs, subjective norm has demonstrated the highest path coefficient of $\beta = 0.352$, p<0.05, followed by attitudes ($\beta = 0.173$, p<0.05).

As for beliefs (behavioural, normative and control beliefs), all three demonstrated influence on the respective construct of attitudes, subjective norms and PBC. Of the three constructs, Control Beliefs appear to exert the most significant influence on PBC. Control Beliefs displayed a positive influence on PBC ($\beta = 0.513$, p<0.05). This is followed by positive influences of behavioural beliefs on attitudes ($\beta = 0.369$, p<0.05) and positive normative influences on subjective norms ($\beta = 0.323$, p<0.05). The following table summarises the results of the bootstrapping routine.
Path	Original	Sample	Standard	t value	р
	Sample	Mean	Deviation		Values
	(β)	(M)			
Influences on intentions					
Attitudes -> Intentions	0.173	0.175	0.050	3.441	0.001
Subjective norm -> Intentions	0.352	0.354	0.044	7.971	0.000
PBC -> Intention	-0.011	-0.011	0.053	0.200	0.842
Influence of beliefs					
Behavioural Beliefs -> Attitudes	0.369	0.375	0.044	8.336	0.000
Normative Beliefs -> Subjective norms	0.323	0.327	0.046	7.038	0.000
Control Beliefs -> PBC	0.513	0.520	0.032	16.140	0.000
					5

Table 4.18: Path Coefficients, t-Statistics and p-value hypothesized path

Note: Critical value 1.96 (Significant Levels 5%) two tailed

4.6.2 Effect Size (f^2)

The effect size (f^2) is examined to determine the significance of the constructs on the model. Using the guidelines given by Cohen (1988), the effect size (f^2) are evaluated. Values of 0.02, 0.15 and 0.35 reflect small, medium and large effects, respectively.

All the values are above 0.02 except PBC, which has no effect on intentions. The two remainder independent constructs, attitudes (0.032) and subjective norms (0.147) have medium effects on intentions. Similarly, normative and behavioural beliefs also have medium effects on subjective norms (0.117) and attitudes (0.158), respectively. Only control beliefs have large effects on perceived behavioural control (0.357).

Table 4.19: Effect size

Constructs	Attitudes	PBC	Subjective	Intentions
			Norms	
Attitudes				0.032
Subjective Norms				0.147
PBC				0.000
Behavioural Beliefs	0.158			
Control Beliefs		0.357		
Normative Beliefs			0.117	

4.6.3 Coefficient of Determination (R^2)

Coefficient of Determination (R^2) measures the predictive accuracy of the model. It represents the combine influence of the exogenous variables on the endogenous variable(s), with higher levels indicating higher levels of predictive accuracy (Barclay et al., 1995; Hair et al., 2017).

Using the rule of thumb by Cohen (1988), the adjusted R^2 levels of 0.26, 0.13 and 0.02, is deemed substantial, moderate and weak, respectively. For this study, the adjusted R^2 of the dependent variable, intentions, is 0.159, which is deemed moderate. It indicates that the combined direct and indirect effects (beliefs, attitudes, subjective norms and PBC) accounts for 15.9% of the variance in the intention construct. Similarly, Behavioural Beliefs explains 13.5% of the variance in attitudes (moderate). As for control beliefs, it explains 26.2% of the variance in PBC, which is substantial. Finally, the variance in subjective norms is the weakest, where only 10.3% is explained by normative beliefs.

In summary, most of the adjusted R^2 is greater than 0.10, which is the minimal requirement stipulated by some researchers (Falk & Miller, 1992; Santosa et al., 2005). However, some researchers also opines that there is no generalisable value for R^2 , as it depends on the study and the respondents (Backhause et al., 2003; Gotz et al., 2010). In reviewing the literature on other financial behaviour studies, most also exhibited similar results. Hence, for this study, the adjusted R^2 of 0.159 is considered acceptable.

Constructs	R ²	Adjusted R ²
Attitudes	0.136	0.135
Subjective Norms	0.105	0.103
PBC	0.263	0.262
Intentions	0.164	0.159

Table 4.20: Coefficient of Determination (R^2)

4.6.4 Predictive Relevance (Q^2)

In addition to adjusted R^2 , the predictive relevance (Q^2) of the theoretical framework is also ascertain using the blindfolding procedure. Q^2 can be calculated using the crossvalidated redundancy and cross-validated communality approach. Hair et al. (2017) recommends using the cross-validated redundancy approach, as it indicates the quality of the structural model.

For this study, the omission distance chosen is 7. In order to have predictive relevance, Q^2 values must be larger than 0. From the following table, all values are larger than 0, hence confirming that the model has predictive relevance.

Constructs	Cross-validated Redundancy	Cross-validated Communality
Attitudes	0.073	0.302
Subjective Norms	0.051	0.165
Injunctive Norms	0.457	-0.011
Descriptive Norms	0.680	1.000
Intentions	0.127	0.582
PBC	0.113	0.012
Behavioural Beliefs		0.548
Normative Beliefs	S	-0.030
Control Beliefs		0.395

Table 4.21: Predictive relevance

4.7 Hypotheses Results

In the preceding sections, the measurement and structural models were evaluated and the found to be adequate. The path coefficients, t-statistics and p-value are also use in evaluating the hypothesis. Each path that connects the variables, represents a hypothesis, is presented in Table 4.29. Hypothesis 1, states that attitudes have a positive influence on intentions to plan financially for retirement among University employees. This hypothesis was fully supported, with the paths linking attitudes to intentions being positive and significant ($\beta = 0.173$, t = 3.441, p < 0.05). The results are consistent with the TPB, which posits that attitudes are the immediate antecedent of intentions.

Hypothesis 2, which evaluated the effects between subjective norms and intentions, displayed a significant positive effect on intention ($\beta = 0.352$, t = 7.971, p < 0.05), and is supported. This is as predicted by TPB, which states that positive subjective norms have a positive influence on intentions to plan financially for retirement among University employees.

Hypothesis 3 addressed the effects between PBC and intentions to plan financially for retirement. Perceived behavioural control is hypothesized to have a positive influence on the intentions to plan financially for retirement among University employees. The hypothesis was not supported ($\beta = -0.011$, t = 0.200, p > 0.05). TPB suggests that an individual's perceptions of control will influence intentions to plan financially for retirement. Unfortunately, the outcome of this study suggests that these factors do not affect intentions.

Hypotheses 4a, 4b and 4c addressed the influence of beliefs. Hypothesis 4a, which stated that behavioural beliefs have a positive influence on attitudes, is fully supported by the model. A positive and significant influence is displayed ($\beta = 0.369$, t = 8.336, p < 0.05). Similarly, Hypothesis 4b is also supported. This hypothesis states that normative beliefs have a positive influence on subjective norms ($\beta = 0.323$, t = 7.038, p < 0.05). Hypothesis 4c which states that control beliefs have a positive influence on perceived behavioural control also displayed a positive and significant influence ($\beta = 0.513$, t = 16.140, p < 0.05). Overall, the results were shows that beliefs have a significant and positive influence over

attitudes, subjective norms and perceived behavioural control, which supports the TPB model.

In summary, the assessment of the path coefficient shows that all proposed hypotheses are supported, except for hypothesis H3. All supported hypotheses are significant at the level of 0.05 and have expected sign directions (i.e. positive). The following table summarises the hypotheses established for this study.

No.	Hypothesis statement	Result
1	Attitudes have a positive influence on intentions to plan financially	Supported
	for retirement among University employees.	
2	Subjective norms have a positive influence on intentions to plan	Supported
	financially for retirement among University employees.	
3	Perceived behavioural control has a positive influence on intentions	Not
	to plan financially for retirement among University employees.	Supported
4a	Behavioural Beliefs has a positive influence on attitudes among	Supported
	University employees.	
4b	Normative Beliefs has a positive influence on subjective norms	Supported
	among University employees.	
4c	Control Beliefs has a positive influence on perceived behavioural	Supported
	control among University employees.	

Table 4.22: Results of Hypothesis Testing

The path model is also reflected in Figure 4.1. The circles represent the latent variables or constructs. The arrows between the latent variables indicate influences between the constructs. The effects of PBC on intentions (in italics) is not supported.



Figure 4.11: PLS Results of the Path Analysis

4.8 Summary

This chapter reports the results of the study. A preliminary analysis was performed. Information on missing values, normality, outliers and collinearity are reported. Details on the Exploratory Factor Analysis conducted were discussed. Subsequently, the results of the statistical analysis for CMV is reported. There is no common method variance for this study. Subsequently, the respondents' profile was analysed. This include demographic information such as age, education background, gender, race or ethnicity, religion and marital status. Descriptive analysis was also conducted on the items in the constructs. Both the respondents' profile and descriptive analysis also compare those opting for the government pension scheme and those not in the pension scheme.

The model also undergoes measurement and structural model assessments using PLS-SEM. Results of the measurement model assessment reveal that the model has internal consistency reliability, convergent and discriminant validity. In the structural model assessment, path coefficients' direction, magnitude and significance were evaluated to assess the influence between the independent and dependent variables. Subsequently, the coefficient of determination (R^2), effect size (f^2) and predictive relevance (Q^2) was assessed.

The final section reported the hypothesis results. Five hypotheses were developed. All the hypothesis was supported except for Hypothesis 3, where perceived behavioural control was discovered to have no influence on intentions to plan financially for retirement among University employees.

CHAPTER 5: DISCUSSION

Retirement is recognised as a life stage where a person's ceases his or her working life (Denton & Spencer, 2009). The decision to retire may be voluntary or forced. Regardless of the circumstances surrounding one's retirement decisions, when a person retires, their regular income stream ceased and reliance on alternatives means of support begins. However, the alternatives available to an individual is greatly dependent the preparation and planning done in their earlier lives. These initiatives are voluntary in nature, and very much dependent on the efforts and capabilities of the individual and whether it is facilitated (or impeded) by developments in the environment (Morgan & Eckert, 2004).

With that in mind, this study aims to understand the various determinants and processes of retirement planning. The Theory of Planned Behaviour (TPB) is used as the underpinning theory. TPB assumes that behaviour is influence by a similar set of constructs. These constructs are beliefs, attitudes, subjective norms and perceived behavioural control. Beliefs about financials, planning and retirement lead to the formation of attitudes, subjective norms and perceived behavioural control, which in turn influence intentions to plan financially for retirement (Fishbein & Ajzen, 1975; Fishbein & Ajzen, 2010). However, the relative importance of these determinants varies from one population to another.

A quantitative method using survey is employed to investigate the determinants of financial planning for retirement amongst a sample of 500 public and private universities employees in Malaysia. The questionnaire developed used validated scales and qualitative interviews, with stringent validity and reliability checks. Procedural remedies were also incorporated to reduce common method variance (CMV). Data is cleaned and screened before final analysis. This chapter discuss in greater depth the result of the analysis of data.

5.1 Demographic Results

The sample respondents are analysed based on demographics such as age, education background, gender, marital status, ethnicity and religion. The demographic details of the respondents have been discussed in the previous chapter and its details available in Appendix M.

The age of the sample ranges from below 20 to above 60 years old. Age has always been an important factor in retirement planning. Generally, as one grows older, the level of preparation is expected to increase (Hershey et al., 2007; Hershey et al., 2007; Padawer et al., 2007).

Overall, more than 60% of the sample respondents are below 40 years old. However, those who have opted for the pension schemes are almost similar in proportions (below and above 40) whilst more than 70% of the non-pensions group are from the younger age groups.

The respondents hold various responsibilities in the universities. Most of them are academics, responsible for teaching, lecturing and research activities. The rest of the employees holding non-academic positions such as management, administrative, student related activities and support staff. These proportions are reflective of a university environment.

The survey respondents have various education background, ranging from basic entry level qualifications such as SPM / Certificate Level, Diploma to tertiary and postgraduate degrees. As the main respondents are academics, it is unsurprising that this sample comprise of educated individuals where most of them have at least been through tertiary education. Those that are more educated are reported to plan more as they have the means

and capacity to do so. Generally, as a majority of the respondents are form the non-pension group, they also dominate most of the educational categories. As for the pension group, most of the respondents are of the postgraduate categories.

Gender also has its influences, with men reported to be more likely to engage in financial planning. However, the respondents for this sample are mostly female. This is reported to be common for the education industries in Malaysia, where around 60% are females (Ministry of Education Malaysia, 2014b; Department of Statistics Malaysia, 2016b).

For ethnicity and religion, most of the respondents are Malay Muslims, followed by Chinese Buddhists and Indians Hindus. This is somewhat reflective of Malaysia's ethnic groups and, which is around 60% for Malay Muslims, 25% Chinese Buddhists and 15% Indians Hindus or others religions (Department of Statistics Malaysia, 2015).

The income level is another important determinant in financial planning for retirement (Ekerdt et al., 1996; Ekerdt, 2010; Ekerdt & Baker, 2014). Generally, Malaysia has classified its income levels into three categories. Those that earn less than RM4,360 per month is categorised as B40. They represent the lower-class earners, or the bottom 40% income earners of the country. The second category are the middle classes also known as M40, which earn between RM4,361 to RM9,619 per month and represent the middle 40% of the population, or M40. The final category earns above RM9,619 per month. They represent the upper class earners, and is known as the top 20% earners of the population, or T20 (Khazanah Research Institute, 2016).

Using the above guide, the respondents are classified accordingly. Those earning less than RM30,000 to RM50,000 annually are classified as B40. They represent 44.2% of the sample, where 28.6% are of the pension group and 53.4% are of the non-pension group. The middle classes or M40 are from the income range between RM50,001 to 130,000

annually. They represent 50.6% of the sample. 62.1% of the pension group and 43.8% of the non-pension group are of this income range. The remainder (5.2%), the T20, has an income range of RM130,001 to above RM150,000. 9.2% of the pension group and 2.8% of the non-pension group are of this income range. In summary, a major proportion of the sample respondents are from the middle classes (M40), follow by the B40 group. Most of the middle-income earners are from the pension group.

5.2 Data Analysis

The sample were tested using the Structural Equation Modelling (SEM) approach, in particular, Partial Least Squares (PLS). Results of the hypothesis testing reveal that five out of the six proposed paths within the structural model were supported. The adjusted R^2 reveals that most of the variables can explain intentions to plan financially for retirement. Each of the variables is analysed and compared with the literature and theory.

5.2.1 Influence on Intentions

This study aims to evaluate the suitability of TPB as a generic model of planning for financial retirement, contextualised among University employees in Malaysia. TPB is a social psychology model which postulate that behaviour is a reasoned action which can be predicted and explained using a set of determinants (attitudes, subjective norms, perceived behavioural control) (Fishbein & Ajzen, 1975; Ajzen & Fishbein, 1977; Ajzen & Fishbein, 1980). For this study, these determinants have predictive relevance ($Q^2 = 0.127$) with moderate influence on intentions to plan financially for retirement (adjusted $R^2 = 15.9\%$).

Findings for this study is mostly consistent with the prediction of the theory. Both attitudes and subjective norms can predict intentions to plan financially for retirement for

university employees in Malaysia, with subjective norms having relatively higher values. However, perceived behavioural control (PBC) did not show any influence on intentions. Although this contradicts with the predictions of the theory, it is consistent with the results of many empirical studies in TPB. The results of the study were discussed in greater detail with the research questions, literature, theory and context of the research.

(a) Research Question 1 - What is the influence of attitudes on the intention to plan financially for retirement among University employees?

The objective of this research question is to examine the influence of attitudes on intention to plan financially for retirement among University employees. In TPB, attitudes is expressed as the evaluation of a person on an object or behaviour (Zimmerman, Canale, Britt & Seay, 2015). Evaluation would encompass instrumental and experiential aspects of a behaviour. Instrumental aspects consider the rewards (or punishment) and experiential aspects refers to a person's feelings or emotions towards the behaviour. Generally, individuals are more likely to perform behaviours that are evaluated positively.

The results reveal that the attitudes of university employees in Malaysia toward retirement planning are generally positive, both instrumentally and experientially, where most of the factor loadings are above 0.80. Only one item is removed due to low convergent validity (DA3). University employees acknowledged that retirement planning is good for them and will benefit them in the long term. Planning provides direction and clarity in goals, which helps one navigate over life many challenges, making the overall activity enjoyable.

Attitudes also has statistically significant influence on intention to plan financially for retirement. However, both its path ($\beta = 0.173$, p<0.05) and effect size ($f^2=0.032$) is relatively lower when compared to subjective norms, indicating that the influence is

weaker. Respondents are also wary of the possibility that plans may fail if it were not implemented well, which will lead to unpleasant experiences. This would somewhat influence the motivation to plan, resulting in a lower effect size.

From the descriptive analysis of the items, the overall importance of financial planning for retirement is acknowledged. Instrumentally, retirement planning is beneficial and is even experientially enjoyable to some. However, there are mixed feeling about their experiences. There are also those who find the experience unpleasant. The pattern of evaluation is similar between the pension and non-pension group.

The influences of attitudes on intentions is consistent with the literature discussed in earlier chapters. Van Deventer et al. (2014) found attitudes to be important to influencing intentions towards financial planning for retirement among the student population in South Africa. Binswanger and Carman (2012); Brucker and Leppel (2013) illustrated that having clear and formal plans is essential for retirement planning.

Positive attitudes are also important in other forms of financial behaviours. Croy et al. (2010b) reported that attitudes have positive effects on the likelihood to save. This is in line with studies by Joo and Grable (2005) which discover attitudes, in particular, positive attitudes, play an important role in a successful savings plan. Savings have been frequently used as a proxy for financial planning for retirement and wealth accumulation effort (Hershey et al., 2007).

Attitudes play an important role in encouraging investments. Investments are important considerations in financial planning. Sivaramakrishnan et al. (2017) discovered that investors attitudes have an impact on the intention to invest in equity markets. Nosi et al. (2017) also discovered that attitudes influence the purchase of longevity annuity coverage

in a private pension plan. Attitudes also influence an individual to manage their debts better in a study done by Xiao, Tang, Serido and Shim (2011).

Vyvyan, Blue and Brimble (2014), in exploring the influence of attitudes on financial capabilities, identified important aspects of financial behaviour such as money management, managing risk, having a planning horizon and money consciousness. Croy et al. (2010b) highlighted that positive attitudes are important in influencing the contributions to superannuation funds and to actively manage their investment strategy.

(b) Research Question 2 - What is the influence of subjective norms on the intention to plan financially for retirement among University employees?

The objective of this research question is to explain the influence of subjective norms on the intention to plan financially for retirement among University employees. Subjective norms reflect social norms that are deemed acceptable. These norms exert social pressure. The stronger the social norms are, the more likely an intention to perform the behaviour will be formed (Ajzen & Fishbein, 2005; Ajzen, 2005a; Fishbein & Ajzen, 2010).

Subjective norms have injunctive and descriptive components (Cialdini et al., 1991; Cialdini et al., 2006; Cialdini, 2007). Injunctive norms refer to perceptions of important referents or groups concerning what should be done with respect to a behaviour, whereas descriptive norms refer to perceptions of what important others are doing in regards to the behaviour. Both injunctive and descriptive norms among university employees in Malaysia have high loadings (above 0.80), thus indicating that both of these components are influential.

The path from subjective norms to intentions for University employees in Malaysia is statistically stronger ($\beta = 0.352$, p<0.05), with higher effect size ($f^2=0.147$), in comparison to attitudes. Respondents are motivated to conform to the expectations of the important

referents in their lives. The activities of these important referents also serve as a guide for respondents to begin or continue their planning activities.

From the descriptive analysis of the items, the injunctive and descriptive norms have positive influences (mean above 3.5), except for IN1, which indicates the perception of social pressure to plan (M=3.18). The pattern is similar for the pension and non-pension group.

The positive influence of subjective norms is consistent with the results from numerous researchers (Xiao & Wu, 2008; Croy et al., 2010b; Yang & Devaney, 2012; Nosi, D'Agostino, Pagliuca & Pratesi, 2014; Vyvyan et al., 2014; Nosi et al., 2017). Social pressures from important referents are effective in influencing intentions. Referents from the immediate surroundings of the individual, such as parents, are often emulated by children and the future generation (Friedman & Scholnick, 1997a; Binswanger & Carman, 2012; Wang et al., 2014). Literature have also supported the subjective influences of important referents such as colleagues and friends (Taylor-Carter et al., 1997; Hershey et al., 1998; Jacobs-Lawson & Hershey, 2003; Bayer et al., 2009; Wang & Shultz, 2010; Croy et al., 2012; Zheng et al., 2012; Zhan et al., 2013). Spouses are also known to frequently coordinate retirement plans together (Borsari & Carey, 2003; Street & Desai, 2011). These important referents also play a role in influencing other forms of responsible financial behaviours besides financial planning such as savings and investment (Koposko & Hershey, 2014; Tang et al., 2015; Koposko & Hershey, 2016; Koposko et al., 2016; Kimiyaghalam et al., 2017).

Sources not immediately present to the individual environment such as various professionals and social groups also play a role in influencing financial planning for retirement (Taylor-Carter et al., 1997; Hershey et al., 1998; Hershey et al., 2003; Bayer et al., 2009; Croy et al., 2012). Professional advisers (such as financial planners, lawyers,

accountants and counsellors) dispense advice and influence the financial decision making process of their clients (Joo & Grable, 2005; Grable et al., 2011; Marsden et al., 2011)

(c) Research Question 3 - What is the influence of perceived behavioural control on the intention to plan financially for retirement among University employees?

The objective of this research question is to determine the influence of PBC on the intentions to plan financially for retirement among University employees. PBC measures an individual perception of their ability to perform a behaviour. Two aspects are important considerations in the measures of PBC. The first is self-efficacy, which is the degree of confidence a person has towards performing a behaviour. A person with high self-efficacy would be inclined to deem a task easy and are confident of being able to successfully execute a behaviour. The other aspect is perceived controllability where a person believes the ability and resources required for the behaviour is within their control (Ajzen, 2002b). The higher the self-efficacy and perceived controllability a person has, the higher would be the degree of PBC (Ajzen, 1991). High degrees of PBC are expected to form strong behavioural intentions (Ajzen & Fishbein, 2005; Ajzen, 2005a; Fishbein & Ajzen, 2010).

However, in this study, PBC do not to have any significant effect on intentions to plan of University employees. The factor loadings for self-efficacy is high (above 0.7), but perceived controllability is weak. Items DPBC3_r and DPBC4, which addressed controllability, was recommended for removal during EFA; and in the analysis using PLS-SEM, item DPBC4 was removed. Although item DPBC3_r was retained, it has the lowest loadings in comparison to the other items. From the descriptive analysis, the overall results reveal that most of the respondents are neutral (M=3.15). The pattern is similar between the pension (M=3.24) and non-pension (M=3.10) group. Self-efficacy aspects have overall mean of 3.10, with those of pension group (M=3.22) having higher means in comparison to the non-pension group (M=3.03). Perceive controllability has mean of 3.25, where the pension group has mean of 3.29 and the non-pension group has mean of 3.23.

University employees are confident in their ability to plan and manage their retirement future. However, they also believe that they are already preparing for old age via the existing pension and contribution schemes that they are in. For those in the pension schemes, they would be receiving regular income from the government in their retirement days. Those in the non-pensions scheme, they are mandated to make regular contributions to EPF, which they believe would give them adequate income in their old age. These schemes are mandated by the laws of Malaysia in which the respondents are required to comply to. Unfortunately, the adequacy of the retirement income generated from these schemes are questionable.

Most of the respondents in this sample are below 40 years old. Literature has documented that most individuals start preparing or thinking about retirement after 40. This is due to the fact that many of those below 40 are still struggling with the various commitments of life such as paying mortgage, household and dependents expenses (Harrison, 2005; Harrison et al., 2006; Davis & Hustvedt, 2012; Griffin, Loh & Hesketh, 2013). Hence, university employees may not have thought about preparing for retirement or even have enough resources to plan.

Although the results contradict with TPB, many empirical studies have similar results. It was reported to have weak results in the usage of the Earned Income Tax Credit (Zimmerman et al., 2015) and teenagers' willingness to volunteer with elderly persons (Reuveni & Werner, 2015). Study by Trivedi, Shehata and Mestelman (2005) on PBC and tax compliance intentions also did not find any significant influences in their student sample. A survey conducted by Bobek and Hatfield (2003) using three different scenarios also have mixed results. Two of the scenarios fail to find any effects between PBC and intentions.

5.2.2 Influence of beliefs

Attitudes, subjective norms and PBC can be analysed further to its antecedents, beliefs. Beliefs guide a person's intentions to perform (or not perform) the behaviour in question (Fishbein & Ajzen, 1975; Fishbein & Ajzen, 2010). Beliefs may be formed directly (from observation) or indirectly (from formal sources of information, interactions with family and friends or through self-generated inference processes).

(a) Research Question 4 - What is the influence of beliefs on attitudes, subjective norms and perceived behavioural control among University employees?

The objective of this research question is to determine the influence of beliefs on attitudes, subjective norms and PBC among University employees. According to TPB, individuals have many beliefs, but only the accessible ones are influential. Different population also hold different types of beliefs. Hence, it is recommended to conduct preliminary pilot work to elicit these beliefs (Fishbein, 1963; Fishbein & Ajzen, 1975; Ajzen, 2011). For this study, interviews were conducted amongst a representative sample of university employees.

Three types of beliefs are identified in TPB. They are behavioural beliefs, normative beliefs and control beliefs. Behavioural beliefs influence attitudes; normative beliefs influence subjective norms while control beliefs will help in understanding PBC. From the results, the beliefs elicited have predictive relevance among University employees. Control belief have a substantial influence and large effect size on PBC (adjusted R^2 =0.262, f^2 =0.357, Q^2 =0.113). This is followed by behavioural beliefs with moderate influence and

effect size on attitudes (adjusted $R^2=0.135$, $f^2=0.158$, $Q^2=0.073$) and normative beliefs with weak influence but moderate effect size on subjective norms (adjusted $R^2=0.103$, $f^2=0.117$, $Q^2=0.051$). The results are discussed in greater detail in the following sections.

(b) Influence of Behavioural Beliefs on Attitudes

Behavioural beliefs link intentions to plan financially for retirement to expected outcomes. Generally, if individuals believe that performance of the behaviour will result in positive outcomes, positive attitudes toward a behaviour will be formed (Ajzen & Fishbein, 2005; Ajzen, 2015).

Nine behavioural beliefs items were identified during preliminary pilot work and its influence on attitudes tested. These items are related to autonomy (BA1, BA2, BA3), coping (BC1, BC2, BC3) and self-control (BSC1, BSC2, BSC3_r) outcomes.

Autonomy outcomes is related to the degree of freedom a person have in making choices in life. Every individual desire the freedom to act and make decisions for themselves (Weinstein et al., 2012). The three items linked to autonomy outcomes are highly valued and have high factor loadings (above 0.80). University employees believe that by planning they would be able to visualise their future, which would help them build goals (Stawski et al., 2007), set directions (Noone, Stephens & Alpass, 2009; Noone, Stephens & Alpass, 2010) and be independent and self-sufficient in old age (Harrison, 2005; Grable et al., 2011).

Planning also helps a person cope with uncertainties, problems and the emotional challenges in life (Atkinson et al., 2007). Items BC1, BC2 and BC3 measure the role of planning in coping. These items have factor loadings above 0.80. The foremost concerns during retirement is the loss of regular income (Croy et al., 2010b). University employees believe that by planning they would have the confidence to cope with this loss. Retirement

planning would help ensure that they have adequate resources to maintain their quality of life during retirement.

The third outcome, self-control, highlight some of the inner conflict faced when planning (Thaler & Benartzi, 2004; Benartzi & Thaler, 2007). Planning involved delayed gratification, where a person has to suppress emotions or behaviour in pursuit of long-term goals (Baumeister et al., 2007). Self-control outcomes are represented by three items (BSC1, BSC2, BSC3_r). However, these items have low to moderate loadings. One of the items, BSC3_r, is eventually removed due to low convergent validity. This item is also recommended for removal during EFA. The remaining items have factor loadings below 0.80. University employees acknowledged the importance of planning as a tool of self-discipline and to exercise restraint when spending (Hayes-Roth & Hayes-Roth, 1979; Friedman & Scholnick, 1997a). While this form of restraint may be deemed burdensome by some, the money saved or invested will be beneficial for the future.

Behavioural beliefs items have statistical influence on attitudes. The path coefficient is supported ($\beta = 0.369$, p<0.05). One of the items, BSC3_r, which has the lowest mean value and low convergent validity is eventually removed. With the removal, overall mean ranges from 4.10 to 4.23. Result are also similar for the pension (mean range: 4.19 to 4.33) and non-pension groups (mean range: 4.05 to 4.17). University employees believe that financial planning for retirement will help to achieve autonomy, coping and self-control outcomes with results showing moderate adjusted R^2 and effect size (adjusted $R^2=0.135$, $f^2=0.158$, $O^2=0.073$).

(c) Influence of Normative Beliefs on Subjective Norms

Normative beliefs reflect the perceived expectations and behaviours of important referent individuals or groups. These beliefs influence the perceived social pressure or subjective norms towards a behaviour. Generally, if individual value the expectation and

behaviours of these important referents, positive subjective norms are formed (Ajzen & Fishbein, 2005; Ajzen, 2015).

Normative beliefs have injunctive and descriptive components. Injunctive normative beliefs identifies the important referents' beliefs and expectations whereas descriptive normative beliefs identifies the important referents that are performing the behaviour in question (Fishbein & Ajzen, 1975). During the preliminary interviews, four categories of important persons are identified to have injunctive (INB1, INB2, INB3, INB4) and descriptive influences (DNB1, DNB2, DNB3, DNB4). They are spouse, immediate family members, friends and relatives. Factor loadings generated are high (around 0.70).

However, not all of the four categories of important people identified are applicable to university employees. As a result, normative beliefs have positive but weak predictive influence on subjective norms (adjusted $R^2=0.103$). Nevertheless, the path from normative beliefs to subjective norms is supported ($\beta=0.323$, p<0.05) and it has predictive value ($Q^2=0.051$) with moderate effect size ($f^2=0.117$).

From the descriptive analysis, spouse (M=4.17) and immediate family members (M=3.99) display stronger injunctive normative beliefs, with those from the pension schemes (mean range: 4.02-4.08) displaying higher means in comparison to the non-pension group (mean range: 3.89-3.95). The findings indicate that university employees are more likely to comply with the expectations of these two groups. As for friends and relatives, mean is at 3.28 and 3.19, respectively, where the pension group has a mean range of 3.31-3.36 and non-pension group at 3.57-3.65. This indicates that most of the respondents would either take a neutral stance and are less likely to comply to latter two groups.

Results for descriptive normative beliefs for the four groups of important referents have similar patterns as injunctive normative beliefs, but the differences are less pronounced. The overall mean for spouse (M=3.90) and immediate family members (M=3.85) have higher means compared to friends (M=3.61) and relatives (M=3.56). Similar results are obtained for the pension and non-pension groups.

Literature is supportive of the findings of the study. The first category, spouse, is influential to a majority of the married respondents. Wang et al. (2011) documented that spouses would frequently coordinate their financial and social plans together. Similarly, Street and Desai (2011) recorded that spouses plans also influence retirement transitions and decision making.

The second category are immediate family members. Parents' retirement plan have a significant effect on a person (Koposko & Hershey, 2014; Koposko & Hershey, 2016; Koposko et al., 2016). Kimiyaghalam et al. (2017) discover that children emulate efforts by parents. Tang et al. (2015) further reported that parents also influence responsible financial behaviour, serving as a reference for future behaviour. Also, Kim and Moen (2001) discussed that having dependent children are some reasons why retirees' continue to work. Conversely, some may retire early to take care of grandchildren or dependents with illness.

The third influential category are friends. Friends influence retirement planning and activities as people are more receptive and trusting towards individuals close to them. They may provide support, opportunities, and a desirable social environment (Wang & Shultz, 2010; Zheng et al., 2012; Zhan et al., 2013). Bayer et al. (2009) reported that individuals seem more likely to plan after being exposed to financial preparation seminars with colleagues in their workplaces. Similarly, many seems more willing to engage in financial preparation acquired during seminars with friends (Hershey et al., 1998; Hershey et al.,

2003). Croy et al. (2012) reported that peer effects have influence saving decisions and the formulation of saving plans in United States.

The final influential category are relatives. They represent the family network and are important providers of support and care for people in retirement in many Eastern countries (Yoo & Kim, 2010). However, such traditional family structures are slowly breaking, giving way to smaller families (Teh, Ng, Tey & Siti-Norlasiah Ismail, 2013).

(d) Influence of Control Beliefs on Perceived Behavioural Control

The third beliefs concern control beliefs. These beliefs refer to the perceived presence of factors that facilitate or impede intentions to plan financially for retirement. If more facilitating control beliefs are identified than inhibiting factors, perceived behavioural control should be high and vice versa. Eleven control beliefs items are identified to influence perceived behavioural control during preliminary pilot work, and they are related to competence (CC1, CC2, CC3), resource (CR1, CR2, CR3, CR4, CR5) and risk management (CRS1_r, CRS2 and CRS3_r).

Competence refers to the tools, equipment, training and support a person have in order to perform a behaviour (Deci & Ryan, 2000). In the case of financial planning for retirement, issues of financial literacy, preliminary interviews have uncovered that reliability of sources and choices of financial products has inhibit financial planning for retirement. Such challenges is also reported to be an issue to financial planning for retirement behaviour in similar studies (Mermin, Johnson & Murphy, 2007; Gerrans & Hershey, 2013; Hershey et al., 2013; Lusardi & Mitchell, 2017). Although University employees were influenced by issues of competencies, the factor loadings are moderate (0.58-0.65). The objective of financial planning is to ensure that there are adequate resources during the retirement phase (Balasuriya et al., 2014; Chou et al., 2015). However, the process involved in managing resources (income, expenses and commitments) are challenging for university employees. Similarly, this is a problem to many individuals worldwide (KRC Research, 2016; Employee Benefit Research Institute, 2018). The items linked to managing resources also have higher factor loadings in comparison to competencies (0.70-0.80). University employees voice concerns in sustaining their cost of living in view of the raising expenses and stagnated income. These challenges drained their financial resources, and even with lifestyle changes, it is difficult for them to plan for their retirement requirements.

The final area of concern relates to managing risk. Risk in financial planning for retirement comes in the form of financial risk and social risk. Financial risk is borne by the individual when they make investments choices in financial planning. An investment may or may not yield the returns desired. Though such risks is inevitable, it inhibits university employees intentions when planning financially for retirement (Ainslie, 2005; Farmer & Geanakoplos, 2009; Ainslie, 2010; Charlton et al., 2011; Teuscher & Mitchell, 2011; Choi & Kim, 2014).

Social risk comes in the form of life course risk, which is related to the retirement lifespan (Denton, French, Gafni, Joshi, Rosenthal & Webb, 2001; Denton, Kemp, French, Gafni, Joshi, Rosenthal & Davies, 2004; Denton & Spencer, 2009). During retirement, individuals face higher life course risk such as poverty, social isolation, loss of independence and autonomy as their earnings and health decline. University employees acknowledges that these are risk of aging, and they may face deteriorating physical and mental conditions during retirement. Such risks are hard to manage, and it is inevitable. Items CRS1_r (0.745) and CRS3_r (0.711) measure the risk management. These items

have similar moderate factor loadings as resource management. One of the items, CRS2, is eventually removed due to low convergent validity.

Control beliefs have predictive value on PBC ($Q^2=0.113$). The path coefficient also has high statistical influence ($\beta = 0.513$, p<0.05). Adjusted R^2 is substantial (0.262) with large effects size ($f^2=0.357$).

Descriptive analysis based on 10 items have an overall mean range of 2.71 to 3.62, with most of the items reporting means of above 3. Only items of risk management have means below 3. One of the items, CRS2, is eventually removed due to low convergent validity. However, both the pension and non-pension group have means below 3. Further analysis reveals that most of the respondents are either agreeable or neutral on issues of control beliefs. This is similar for both the pension and non-pension groups. Although university employees agree that issues related to competency, resource and risk management has an impact to inhibit financial planning for retirement, they also feel it is difficult to mitigate its effects.

5.3 Summary

Chapter 5 discussed the results reported in the previous chapter. An overview of demographic analysis results is discussed. This include analysis of university employees' age, education background, gender, marital status, ethnicity, religion and income level.

Results from PLS-SEM were also discussed in details. Five out of the six hypothesis were supported. The adjusted R^2 reveals that most of the variables can explain intentions to plan financially for retirement. The influence of attitudes and subjective norms on intentions are supported, which is in line with previous literature. However, the influence of perceived behavioural control (PBC) on intentions is not supported. Although this

contradicts with the predictions of TPB, it is consistent with the results of many empirical studies in TPB. The results of the study were discussed in greater detail with the research questions, literature, theory and context of the research. As for the influence of beliefs, the effects of behavioural beliefs, normative beliefs and control beliefs on attitudes, subjective norms and PBC, respectively, is well supported.

university

CHAPTER 6: CONCLUSION

The objective of this study is to investigate the determinants of intentions to plan financially for retirement within a selected sample in Malaysia, that is university employees. Retirement planning is deemed as a critical activity worldwide, yet research in this area is fragmented. Investigations has focus on a variety of financial products such as savings, investments, tax or risk management. But financial planning behaviour is not limited to selected products. Focus on only one financial area at the expense of another, would render the findings incomplete. In addition, initial studies have focus on demographics of the individual. Demographic indicators are not explanatory variables and have limited use in theory development (Hershey et al., 2013). Nevertheless, such variables are useful as background factors that indirectly influence intentions and behaviours (Fishbein & Ajzen, 2010; Ajzen & Klobas, 2013).

In view of such limitations, this study attempts to make a contribution to the existing literature by using TPB as a generic model of planning for financial retirement, which can be contextualized to different social contexts and populations. The use of TPB in financial and retirement behaviour studies are limited, but existing studies had shown promising results (Bobek & Hatfield, 2003; Xiao, Sorhaindo & Garman, 2006; Xiao & Wu, 2006; Xiao & Wu, 2008; Croy et al., 2010b; Kimiyaghalam et al., 2017; Nosi et al., 2017). The contributions and limitations of the study was discussed in greater detail in the subsequent sections with recommendations for future research.

6.1 Contributions

This study expanded the use of TPB in financial planning for retirement. Most of the constructs in TPB are able to predict intentions to plan financially for retirement. Several major contributions are addressed in this study.

6.1.1 Empirical Contributions

There is a consensus among researchers that there are a multitude of constructs to explain financial behaviour (Tan & Folk, 2011; Hershey et al., 2013). Using TPB, this study has uncovered that subjective norms and attitudes towards financial planning for retirement are important considerations. Subjective norms have a stronger influence on intentions. Although positive attitudes do influence intention to plan financially for retirement, its effects are weaker when compared to subjective norms. PBC, in turn, was not significant in the final model.

This indicates that University employees are more motivated by the expectations and actions of important referents in their social environment in comparison to attitudes towards financial planning for retirement. For many Eastern countries, the main source of retirement support lies with the family network which provides both formal and informal support.

University employees also have positive attitudes towards financial planning. They recognised the importance and benefits of planning. However, it does not play a major role in motivating behaviour. Instead, University employees tend to rely on the existing legal or formal mechanism to save for their retirement (pension, EPF). Such findings contributes to discoveries in recent studies by Fishbein and Ajzen (2010); Ajzen and Klobas (2013); Klobas and Ajzen (2015) which states that different antecedents are expected to have different effects on intentions. Hence, for this sample of university employees in Malaysia subjective norms and attitudes play an important role.

There are two categories of universities in Malaysia, that is public and private universities. Public universities employees have access to both of the retirement contribution schemes in the country. These would include the government pension schemes; which is a form of defined benefit plan, where a lifelong monthly payment is provided upon retirement. They can also opt for savings through the Employees Provident Fund (EPF), which are similar to defined contribution plans, where the contributions are based on a prescribed rate for employees and employers. Private universities only have access to EPF. The sample enable further analysis into the groups that opt for the pension and non-pension schemes, enabling demographic and descriptive comparisons between these groups.

6.1.2 Theoretical Contributions

According to Whetten (1989); Corley and Gioia (2011), a theoretical contribution is the extension of an existing theory that will fundamentally change any prior understanding of the relationships of the indicators. TPB, the underpinning theory, has been used in many studies such as health behaviour (Conner & Sparks, 2005), leisure choices (Ajzen & Driver, 1992), weight loss (Sparks et al., 1995), fertility intentions (Ajzen & Klobas, 2013) and information technology usage (Taylor & Todd, 1995). However, there is limited research on its use in financial and retirement behaviour studies.

Nonetheless, findings from this study has proven that TPB constructs can be used to explain the complexity of financial planning for retirement. It is able to integrate a small number of constructs (beliefs, attitudes, subjective norms and PBC) to explain planning for financial retirement among university employees, thus providing a valuable generic model to extend the body of knowledge on financial behaviour.

Most research in retirement planning and using TPB is done in developed, western countries, which offer a very different set of contextual variables. Firstly, developed countries has very different social protection initiatives, culture and income level. These countries are also ageing societies, where issues pertaining planning for retirement are critical.

However, for developing countries and non-western societies, there are limited studies on retirement planning. These countries have not reached the ageing population stage; hence issues of retirement planning had not been a major focus. Nonetheless, interest in this topic is increasing. Most countries are expected to have a demographic trend of an aging population eventually, in view of increasing lifespan and a drop in fertility rates among its citizens.

For Malaysia, it is expected to be an aging population in 2035. There is a growing interest on retirement and planning. Yet, most researchers have to rely on existing research in developed countries. Such studies have limitations, as they are based on the respective countries' legislative, administrative, social and cultural environment. It is not practical to adopt the findings of research from other countries.

Nonetheless, TPB have yielded significant findings from this study. Such findings proved that TPB is a useful model that can be contextualized into different population and retirement settings.

6.1.3 Methodological Contributions

Financial planning for retirement is a complex behaviour. To understand its' complexity, a variety of approaches, methodologies and data sources is required. Several methodological choices are made to ensure that the data collected is of quality and able to

address these complexities. These are novel methods that also contributes towards fortifying the validity and reliability of the study.

Firstly, in the development of the survey instrument, two separate methods are used to develop the items (Morgan, 1998; Tashakkori & Teddlie, 1998; Fishbein & Ajzen, 2010). Attitudes, subject norms and perceived behavioural control are adapted from existing validated scales from the literature. Beliefs (behavioural beliefs, normative beliefs and control beliefs), however, are developed from qualitative interviews. This is because beliefs are unique and should be elicited from the population of study (Fishbein, 1963; Ajzen & Fishbein, 1980). Subsequent tests using PLS-SEM reveal that the set of modal beliefs developed are significant indicators. Such steps provide a richer view of the complexity and uniqueness of financial human behaviour.

Secondly, to ensure the 'goodness' of the measures used, all items in the survey instrument are subjected to content validation by a team of experts. Each item is evaluated based on consistency, representativeness, relevance and clarity. These ratings are used to calculate Content Validity Index (CVI) and the Kappa Coefficient (Polit & Beck, 2006). A set of decision rules are used to eliminate items that do not meet with the criteria specified. Such steps ascertains the validity of the items used in the survey.

Many studies on human behaviour have non-normal data distribution. Similarly, this study also has such challenges, necessitating the use of nonparametric techniques. Steps are taken to choose a statistical program that can cater to data that have normality issues. PLS-SEM is chosen as it allows structural equation modelling to be used successfully with non-normal data. The findings are able to explain the effects between the various constructs and also provide information on the contribution of each item. Furthermore, the use of 5,000 re-samples in the bootstrapping technique and the robust validation process enhance the level of confidence on the findings.

230

6.2 Practical Implications

In view of the external environment challenges (such as an aging population), many governments are increasingly shifting responsibilities of retirement support to individuals. Yet, how prepared the individual is to take such responsibilities is questionable. Governments should not merely depend on the prudent behaviour of individuals to plan for retirement and hope that it will bring its just rewards. Policymakers can use TPB as a framework to understand planning behaviours and introduce policy frameworks that can help individual undertake their planning responsibilities more efficiently. Inadvertently, such efforts will increase the effectiveness of the policies developed and the well-being of individuals (Cohen, 2014).

This study is also beneficial to financial planning practitioners and the financial services industry in general. Financial planners are professionals who offer financial advice and advocate the importance of planning. These professionals charge fees or commission for the financial advisory services they provide to assist clients in planning. The financial planning industry is currently at its infancy stage, but has promising market potential in Malaysia. Unfortunately, the trust level towards these professionals are low, and many are sceptical about their commitment to solving their clients' financial problems.

For the industry to develop, the financial players (financial planners and the financial services sector) need to understand what motivates the mindsets of their clients. TPB offers a framework into the consumers psychology when planning for retirement. Understanding the motivation and needs of their clients will help increase their effectiveness in product development and marketing of appropriate financial products or services.

6.3 Limitations

In this study, these determinants of TPB (beliefs, attitudes, subjective norms and perceived behavioural control) accounts for about 16% of the variance in the intention construct. Although the overall effects are moderate, it also indicates that there are other possible variables that play a role in influencing intentions. However, the inclusion of all possible variables in one model is beyond the scope of this study and is an area for future research.

For this study, primary data is collected via a survey. Initial data collection had a very low response rate. Subsequent attempts were able to generate higher responses, but such steps were bounded by the policy and procedural requirements of the respective institutions (selected locations, within the premises of the universities, during working hours). Some of the survey forms collected were also unusable and had to be excluded.

The survey instrument comes in the form of a self-administered questionnaire. Although strict measures are used during the questionnaire development process and complete anonymity and confidentiality is guaranteed, there are possibilities of biasness and misinterpretations. Respondents may opt to give socially desirable responses, have poor memory or have self-presentation concerns. There are also respondents who perceived the questions as sensitive and personal, which deters them from participating.

This research is a cross-sectional study. It provides a snapshot of the respondents' views on intentions. But behaviours and intentions may change over time. However, many cross-sectional studies on TPB also have useful and insightful discoveries (Notani, 1998; Armitage & Conner, 2001; Schulze & Wittmann, 2003; Rivis & Sheeran, 2004; Rivis et al., 2009). In addition, Fishbein and Ajzen (1975); Ajzen (1991) have maintained that beliefs and attitudes tend to remain stable over time.

In summary, care is taken to minimize the limitations and to ensure quality in the overall study. However, the limitations discussed are unavoidable and are inherent in similar studies. Nonetheless, the findings are still relevant, and is proven significant through rigorous testing.

6.4 Future Research

This study has proven that the key determinants of TPB is able to explain intentions to plan financially for retirement. Yet, at the same time, it also acknowledges that there are other possible variables that play a role in influencing intentions. The sufficiency of TPB in addressing financial and retirement behaviour is an area for future research (Ajzen & Fishbein, 2005; Klobas & Ajzen, 2015). This, coupled with the interdisciplinary nature of planning offers a myriad of research opportunities. Efforts can be directed to gain better insights into the origins of beliefs underlying the theory's predictor variables or inclusion of additional variables to improve explained variance. The direct, indirect, moderator or mediation effects of the variables can also be explored further (Conner & Armitage, 1998; Armitage & Conner, 2001). Further exploration into the influence of PBC on intentions to plan may also yield interesting findings. In some studies, PBC has been reported to influence behaviour directly and even moderates the effects of subjective norms and attitudes (Ajzen, n.d.). Hence, there are many venues that can improve the explanation on financial and retirement planning.

The findings of this research reveal that subjective norms are an important determinant of intentions to plan financially for retirement. The role of families, adult children and the extended family network have traditionally been an important support for elderly people in many Eastern countries (Yoo & Kim, 2010). This support is a form of filial obligation and responsibility. It is also conceptualised as a societal attitude, which deems it as a duty of

adult children to meet the needs of their ageing parents (Walker, 2002). In many countries, it is incorporated as a fundamental religious and moral commandment.

However, with social changes such as the increase in life expectancy, recent changes in divorce, child-bearing, urbanisation and migration and women's employment, filial practices are slowly eroding. Traditional family structures, which not only involve the immediate families but also extended families such as relatives, are breaking, giving way to smaller families (Teh et al., 2013). As a result, there is a reduced number of children to share both social and financial responsibilities. This increased burden of responsibility takes a toll on the existing resources of children, where some may find it increasingly difficult to fulfil their filial obligations. In many Eastern countries, such situations have dire consequences to people in retirement, as the family network is an important provider (sometimes the only provider) of support and care. Currently, studies on filial obligations are limited and with the growing importance of retirement, this is an important area for future research (Lapinski et al., 2017).

The respondents in this study are university employees. They represent a segment of the formal sectors in Malaysia which are protected by the retirement contributions schemes of the country (pension, EPF). However, there are many sub-groups from the informal sector that are not protected by any contribution schemes such as farmers and entrepreneurs. How these individuals handle retirement planning has yet been fully explored. Investigations into this area will yield many intervention opportunities that can influence individuals' behaviour in a desired manner.

6.5 Concluding Comments

The growing importance of financial planning for retirement is undeniable. Effective financial planning is a key component to ensuring quality of life during retirement. Yet, many individuals have not been successful in planning. In general, individuals seem disinterested or unmotivated in managing their retirement affairs (Harrison et al., 2006). In fact, financial planning for retirement is a universal challenge for many individuals and policymakers. Nonetheless, the impact of not preparing for retirement are dire, as evidences of low savings rate and financial problems are already prevalent in many countries (Brady, 2010; Kennedy & Matwijiw, 2010; MacDonald et al., 2011; Croy et al., 2012; Ellen et al., 2012; Pfau & Kariastanto, 2012).

In response to the challenges of planning, this study has applied theories and methodological approaches that have been used successfully in behavioural research. Some of these approaches are not widely applied in financial planning for retirement research, but have proven to be a worthwhile contribution towards understanding this complex behaviour. The findings contributes toward the body of knowledge and has unravel the mysterious world of financial planning for retirement.
REFERENCES

- Adams, G. A. & Rau, B. L. (2011). Putting Off Tomorrow to Do What You Want Today: Planning for Retirement. *American Psychologist, 66*(3), 180-192.
- Agnew, J. R., Szykman, L. R., Utkus, S. P. & Young, J. A. (2012). Trust, plan knowledge and 401(k) savings behavior. *Journal of Pension Economics & Finance*, 11(1), 1-20. doi: <u>http://dx.doi.org/10.1017/S1474747211000230</u>
- Ainslie, G. (2005). Précis of Breakdown of Will. *Behavioral and Brain Sciences*, 28, 635-673.
- Ainslie, G. (2010). Procrastination, the basic impulse. In C. Andreou & M. White (Eds.), *The Thief of Time: Philosophical Essays on Procrastination* (pp. 11-27). New York: Oxford University Press.
- Ajzen, I. (1985). From intentions to actions: A theory of planned behavior. In J. Kuhl & J. Beckman (Eds.), Action-control: From cognition to behavior (pp. 11- 39). Berlin: Springer Verlag.
- Ajzen, I. (1991). The Theory of Planned Behavior. Organizational Behavior And Human Decision Processes, 50, 179-211.
- Ajzen, I. (2002a). Construction of a standard questionnaire for the theory of planned behavior. Retrieved 21 August, 2017, from <u>http://www-unix.oit.umass.edu</u>
- Ajzen, I. (2002b). Perceived Behavioral Control, Self-Efficacy, Locus of Control, and the Theory of Planned Behavior. *Journal of Applied Social Psychology*, 32(4), 665-683. doi: 10.1111/j.1559-1816.2002.tb00236.x
- Ajzen, I. (2005a). Attitudes, Personality And Behavior (2nd ed.). Berkshire: Open University Press.
- Ajzen, I. (2005b). Laws of human behavior: Symmetry, compatibility, and attitudebehavior correspondence. In B. B. A. Beauducel, M. Bosniak, W. Conrad, G. Schönberger, D. Wagener, (Ed.), *Multivariate research strategies* (pp. 3-19). Aachen: Shaker Verlag.
- Ajzen, I. (2006). Constructing a TPB Questionnaire. Retrieved 17 June, 2016, from http://people.umass.edu/~aizen/pdf/tpb.measurement.pdf
- Ajzen, I. (2011). The theory of planned behaviour: Reactions and reflections. *Psychology* & *Health*, 26(9), 1113-1127. doi: 10.1080/08870446.2011.613995
- Ajzen, I. (2012). Martin Fishbein's Legacy: The Reasoned Action Approach. *The Annals* of the American Academy of Political and Social Science, 640(1), 11-27.
- Ajzen, I. (2015). The theory of planned behaviour is alive and well, and not ready to retire: a commentary on Sniehotta, Presseau, and Araújo-Soares. *Health Psychology Review*, 9(2), 131-137. doi: 10.1080/17437199.2014.883474

- Ajzen, I. (2019). TPB Diagram. Retrieved 4 March, 2019, from http://people.umass.edu/aizen/tpb.diag.html#null-link
- Ajzen, I. (n.d.). Frequently Asked Questions. Retrieved 18 June, 2016, from <u>http://people.umass.edu/aizen/faq.html</u>
- Ajzen, I. & Albarracin, D. (2007). Predicting And Changing Behavior: A Reasoned Action Approach. In I. Ajzen, D. Albarracín & R. Hornik (Eds.), *Prediction And Change Of Health Behavior: Applying the Reasoned Action Approach* (pp. 3-21). Mahwah, New Jersey: Lawrence Erlbaum Associates.
- Ajzen, I. & Dasgupta, N. (2015). Explicit and implicit beliefs, attitudes, and intentions: The role of conscious and unconscious processes in human behavior. In P. Haggard & B. Eitam (Eds.), *The sense of agency* (pp. 115-144). Oxford: Oxford University Press.
- Ajzen, I. & Driver, B. L. (1991). Prediction of leisure participation from behavioral, normative, and control beliefs: An application of the theory of planned behavior. *Leisure Sciences*, 13(3), 185-204.
- Ajzen, I. & Driver, B. L. (1992). Application of the theory of planned behavior to leisure choice. *Journal of Leisure Research*, 24, 207-224.
- Ajzen, I. & Fishbein, M. (1969). The Prediction of Behavioral Intention in a choice situation. *Journal of Experimental Social Psychology*, *5*, 400-416.
- Ajzen, I. & Fishbein, M. (1977). Attitude-behavior relations: A theoretical analysis and review of empirical research. *Psychological Bulletin*, 84(5), 888-918.
- Ajzen, I. & Fishbein, M. (1980). Understanding Attitudes and Predicting Social Behaviour. New Jersey: Prentice-Hall, Inc.
- Ajzen, I. & Fishbein, M. (2000). Attitudes and the attitude-behavior relation: Reasoned and automatic processes. In W. Stroebe & M. Hewstone (Eds.), *European Review of Social Psychology* (pp. 1-33). New York: John Wiley & Sons.
- Ajzen, I. & Fishbein, M. (2005). The influence of attitudes on behavior. In D. Albarracín, B. T. Johnson & M. P. Zanna (Eds.), *The handbook of attitudes* (pp. 173-221). Mahwah, New Jersey: Lawrence Erlbaum Associates.
- Ajzen, I. & Klobas, J. (2013). Fertility intentions: An approach based on the theory of planned behavior. *Demographic Research*, 29, 203-232.
- Ajzen, I. & Sheikh, S. (2013). Action versus inaction: anticipated affect in the theory of planned behavior. *Journal of Applied Social Psychology*, 43(1), 155-162. doi: 10.1111/j.1559-1816.2012.00989.x
- Allison, P. (2001). *Missing data Quantitative applications in the social sciences* (Vol. 136). Thousand Oaks: Sage.
- Allport, G. W. (1935). Attitudes. In C. A. Murchison (Ed.), A handbook of social psychology (pp. 798 - 844). Worcester, MA: Clark University Press.

- Ameriks, J., Caplin, A. & Leahy, J. (2003). Wealth Accumulation And The Propensity To Plan. *Quarterly Journal of Economics*, 118(3), 1007-1047. doi: 10.1162/00335530360698487
- Ando, A. & Modigliani, F. (1963). The "Life Cycle" Hypothesis of Saving: Aggregate Implications and Tests. *The American Economic Review*, 53(1), 55-84. doi: 10.2307/1817129
- Antolín, P. & Stewart, F. (2009). Private Pensions and Policy Responses to the Financial and Economic Crisis. OECD Working Papers on Insurance and Private Pensions No. 36, OECD publishing. doi: 10.1787/224386871887
- Armantier, O., de Bruin, W. B., Topa, G., van der Klaauw, W. & Zafar, B. (2015). Inflation Expectations And Behavior: Do Survey Respondents Act On Their Beliefs? *International Economic Review*, 56(2), 505-536. doi: 10.1111/iere.12113
- Armitage, C. & Conner, M. (1999). The theory of planned behaviour: Assessment of predictive validity and perceived control. *British Journal of Social Psychology*, 38(1), 35-54.
- Armitage, C. J. & Conner, M. (2000). Social cognition models and health behavior: A structured review. *Psychology and Health*, *15*, 173-189.
- Armitage, C. J. & Conner, M. (2001). Efficacy of the Theory of Planned Behaviour: A meta-analytic review. *British Journal of Social Psychology*, 40(4), 471-500.
- Atkinson, A., McKay, S., Collard, S. & Kempson, E. (2007). Levels of Financial Capability in the UK. *Public Money and Management*, 27(1), 29-36.
- Atkinson, A. & Messy, F. (2011). Assessing financial literacy in 12 countries: An OECD/INFE international pilot exercise. *Journal of Pension Economics & Finance*, 10(4), 657-665.
- Atkinson, A. & Messy, F. (2012). Measuring Financial Literacy: Results of the OECD / International Network on Financial Education (INFE) Pilot Study (Organisation for Economic Co-operation and Development Working Papers on Finance, Insurance and Private Pensions, No. 15). Paris: OECD Publishing.
- Backhause, K., Erichson, B., Plinke, W. & Weiber, R. (2003). *Multivariate Analysemethoden: Eine Anwendungsorientierte Einfuhrung*. Berlin: Springer.
- Bagozzi, R. P. & Yi, Y. (1988). On the Evaluation of Structural Equation Models. *Journal* of the Academy of Marketing Science, 16(1), 74-94.
- Bagozzi, R. P., Yi, Y. & Phillips, L. W. (1991). Assessing Construct Validity in Organization Research. *Administrative Science Quarterly*, *36*(3), 421-458.
- Balasuriya, J., Gough, O. & Vasileva, K. (2014). Do optimists plan for retirement? A behavioural explanation for non-participation in pension schemes. *Economics Letters*, *125*(3), 396-399. doi: 10.1016/j.econlet.2014.10.012

- Bandura, A. (1995). *Self-efficacy in changing societies*. New York: Cambridge University Press.
- Bandura, A. & Pallak, M. S. (1982). Self-efficacy mechanism in human agency. *American Psychologist*, *37*(2), 122-147.
- Barclay, D., Thompson, R. & Higgins, C. (1995). The Partial Least Squares (PLS) Approach to Causal Modeling: Personal Computer Adoption and Use as an Illustration. *Technology Studies*, 2(2), 285-309.
- Barroso, C., Carrion, G. C. & Roldan, J. L. (2010). Applying Maximum Likelihood and PLS on Different Sample Sizes: Studies on SERVQUAL Model and Employee Behaviour Model. In V. E. Vinzi, W. W. Chin, J. Henseler & H. Wang (Eds.), *Handbook of Partial Least Squares: Concepts, Methods and Applications* (pp. 427-447). Heidelberg: Springer.
- Bartlett, M. S. (1954). A note on the multiplying factors for various chi square approximations. *Journal of the Royal Statistical Society, 16*(Series B), 296-298.
- Baumeister, R. F., Vohs, K. D. & Tice, D. M. (2007). The Strength Model of Self-Control. *Current Directions In Psychological Science, 16*(6), 351-355.
- Bayer, P. J., Bernheim, B. D. & Scholz, J. K. (2009). The Effects Of Financial Education In The Workplace: Evidence From A Survey Of Employers. *Economic Inquiry*, 47(4), 605-624.
- Beal, D. J. & Delpachitra, S. B. (2003). Financial Literacy among Australian University Students. *Economic Papers*, 22(1), 65-78.
- Beal, D. J. & McKeown, W. (2008). Personal Finance (4th Ed.). Milton, Queensland: John Wiley & Sons.
- Bearden, W. O., Netemeyer, R. G. & Haws, K. L. (2011). *Handbook of Marketing Scales: Multi-Item Measures for Marketing and Consumer Behavior Research*. (3rd ed.). Thousand Oaks, California: Sage Publications, Inc.
- Beck, L. & Ajzen, I. (1991). Predicting dishonest actions using the theory of planned behavior. *Journal of Research in Personality*, 25, 285-301.
- Benartzi, S. & Thaler, R. H. (2007). Heuristics and Biases in Retirement Savings Behavior. Journal of Economic Perspectives, 21(3), 81-104.
- Berger, E. D. & Denton, M. A. (2004). The interplay between women's life course work patterns and financial planning for later life. *Canadian Journal On Aging*, 23(suppl.), S81 - S95.
- Binswanger, J. & Carman, K. G. (2012). How Real People Make Long-Term Decisions: The Case Of Retirement Preparation. *Journal of Economic Behavior & Organization*, 81(1), 39-60. doi: 10.1016/j.jebo.2011.08.010

- Bobek, D. & Hatfield, R. C. (2003). An Investigation of the Theory of Planned Behavior and the Role of Moral Obligation in Tax Compliance. *Behavioral Research in Accounting*, 15, 13-38.
- Bollen, K. A. (2011). Evaluating Effect, Composite, And Causal Indicators In Structural Equation Models. *MIS Quarterly*, *35*(2), 359-372.
- Borsari, B. & Carey, K. B. (2003). Descriptive and injunctive norms in college drinking: A meta-analytic integration. *Journal of Studies on Alcohol and Drugs*, 64, 331-341.
- Börsch-Supan, A., Bucher-Koenen, T., Coppola, M. & Lamla, B. (2015). Savings in times of demographic change: Lessons from the german experience. *Journal of Economic Surveys, 29*(4), 807-829. doi: 10.1111/joes.12116
- Bowden, A., Fox-Rushby, J. A., Nyandieka, L. & Wanjau, J. (2002). Methods for pretesting and piloting survey questions: illustrations from the KENQOL survey of health-related quality of life. *Health Policy And Planning*, *17*(3), 322-330.
- Bowlby, G. (2007). Defining Retirement. Perspectives on Labour and Income, 8, 15-19.
- Brady, P. J. (2010). Measuring retirement resource adequacy. Journal of Pension Economics & Finance, 9(2), 235-262.
- Brucker, E. & Leppel, K. (2013). Retirement Plans: Planners and Nonplanners. *Educational Gerontology*, 39(1), 1-11.
- Bruner II, G. C. (2009). Marketing Scales Handbook: A Compilation of Multi-Item Measures for Consumer Behavior & Advertising Research (Vol. 5). Illinois: GCBII Productions.
- Calasanti, T. (2010). Gender Relations and Applied Research on Aging. *The Gerontologist*, 50(6), 720-734. doi: 10.1093/geront/gnq085
- Cattell, R. B. (1966). The scree test for the number of factors. *Multivariate Behavioral Research*, *1*, 245-276.
- Central Bank Of Malaysia. (2018). Deputy Governor's Speech at the Launch of the Financial Management for Retirement Module - "Shaping the Future Starts Now". Retrieved 29 July, 2018, from http://www.bnm.gov.my/index.php?ch=en_speech&pg=en_speech&ac=722
- Charlton, S. R., Gossett, B. D. & Charlton, V. A. (2011). Beyond The Shadow Of A Trait: Understanding Discounting Through Item-Level Analysis Of Personality Scales. *Psychological Record*, 61(4), 583-598.
- Chin, W. W. (1998a). Issues and opinion on structural equation modeling. *MIS Quarterly*, 22(1), vii-xvi.
- Chin, W. W. (1998b). The Partial Least Squares Approach of Structural Equation Modeling. In G. A. Marcoulides (Ed.), *Modern Methods for Business Research* (pp. 295-336). Mahwah, NJ: Lawrence Erlbaum.

- Chin, W. W. (2010). How to Write Up and Report PLS Analyses. In V. E. Vinzi, W. W. Chin, J. Henseler & H. Wang (Eds.), *Handbook of Partial Least Squares*. Berlin: Springer-Verlag.
- Chin, W. W. & Newsted, P. R. (1999). Structural Equation Modeling Analysis with Small Samples Using Partial Least Squares. In R. Hoyle (Ed.), *Statistical Strategies for Small Sample Research* (pp. 307-341). Thousand Oaks, CA: Sage Publications, Inc.
- Chinen, K. & Endo, H. (2012). Effects of Attitude and Background on Students' Personal Financial Ability: A United States Survey. *International Journal of Management*, 29(2), 778-791.
- Choi, S., Kariv, S., Muller, W. & Silverman, D. (2014). Who Is (More) Rational? *American Economic Review*, 104(6), 1518-1550. doi: 10.1257/aer.104.6.1518
- Choi, Y. & Kim, S. (2014). *Hyperbolic Discounting and Consumer Patience During Financial Crises in Korea* (pp. 1-20). Seoul: National Research Foundation of Korea.
- Chou, K. L., Yu, K. M., Chan, W. S., Wu, A. M., Zhu, A. Y. F. & Lou, V. W. Q. (2015). Perceived retirement savings adequacy in Hong Kong: an interdisciplinary financial planning model. *Ageing & Society*, 35(8), 1565-1586. doi: 10.1017/s0144686x14000373
- Christelis, D., Jappelli, T. & Padula, M. (2010). Cognitive Abilities and Portfolio Choice. *European Economic Review*, 54(1), 18-38.
- Cialdini, R. B. (2007). Descriptive social norms as underappreciated sources of social control. *Psychometrika*, 72(2), 263-268.
- Cialdini, R. B., Demaine, L. J., Sagarin, B. J., Barrett, D. W., Rhoads, K. & Winter, P. L. (2006). Managing social norms for persuasive impact. *Journal of Social Influence*, *1*(1), 3-15.
- Cialdini, R. B., Kallgren, C. A. & Reno, R. R. (1991). A focus theory of normative conduct: A theoretical refinement and reevaluation of the role of norms in human behavior. New York: Academic Press.
- Cialdini, R. B., Reno, R. R. & Kallgren, C. A. (1990). A focus theory of normative conduct: Recycling the concept of norms to reduce littering in public places. *Journal of Personality and Social Psychology, 58*, 1015-1026.
- Cohen, J. (1988). *Statistical power analysis for the behavioral sciences* (2nd ed.). Hillsdale, NJ: Lawrence Erlbaum Associates.
- Cohen, M. B. (2014). Superannuation retirement balances: A surprising outcome. *Australasian Journal on Ageing*, 33(4), E41-E45. doi: 10.1111/ajag.12094
- Collins, J. M. & O'Rourke, C. (2009). *Still Holding Out Promise: A Review of Financial Literacy Education and Financial Counseling Studies*. Indianapolis: Networks Financial Institute.

- Comrey, A. L. & Lee, H. B. (1992). *A first course in factor analysis* (2nd ed.). Hillsdale, NJ: Erlbaum.
- Conner, M. & Armitage, C. J. (1998). Extending the Theory of Planned Behavior: A Review and Avenues for Further Research. *Journal of Applied Social Psychology*, 28(15), 1429-1464.
- Conner, M. & Sparks, P. (2005). *Predicting Health Behaviour: Research and Practice with Social Cognition Models*. Philadelphia: Open University Press.
- Cooper, D. R. & Schindler, P. S. (2011). *Business research methods* (11th ed.). New York: McGraw Hill.
- Corley, K. G. & Gioia, D. A. (2011). Building Theory About Theory Building: What Constitutes A Theoretical Contribution? *Academy of Management Review*, 36(1), 12-32.
- Courchane, M., Gailey, A. & Zorn, P. (2008). Consumer credit literacy: What price perception. *Journal of Economics and Business*, 60(1-2), 125-138.
- Courchane, M. & Zorn, P. (2005). Consumer Literacy and Creditworthiness. Retrieved 18 August, 2018, from <u>https://www.researchgate.net/publication/5042561/download</u>
- Crano, W. D. & Prislin, R. (2006). Attitudes and persuasion. *Annual Review of Psychology*, 57, 345 374.
- Creswell, J. W. (2014). *Research Design: Qualitative, Quantitative and Mixed Methods Approaches* (4th ed.). Thousand Oaks: Sage Publication Inc.
- Cronbach, L. J. (1971). Test validation. *Educational measurement, issues and practice, 2*, 443-507.
- Croy, G., Gerrans, P. & Speelman, C. (2010a). Injunctive Social Norms Primacy Over Descriptive Social Norms In Retirement Savings Decisions. *International Journal* of Aging & Human Development, 71(4), 259-282. doi: 10.2190/AG.71.4.a
- Croy, G., Gerrans, P. & Speelman, C. (2010b). The role and relevance of domain knowledge, perceptions of planning importance, and risk tolerance in predicting savings intentions. *Journal of Economic Psychology*, 31(6), 860-871. doi: 10.1016/j.joep.2010.06.002
- Croy, G., Gerrans, P. & Speelman, C. (2012). Normative influence on retirement savings decisions: Do people care what employers and the government want? *Australian Journal of Psychology*, 64(2), 83-91. doi: 10.1111/j.1742-9536.2011.00029.x
- Croy, G., Gerrans, P. & Speelman, C. P. (2015). A mediating role for anticipated regret in predicting retirement savings intention between groups with (without) past behaviour. *Australian Journal of Psychology*, 67(2), 87-96. doi: 10.1111/ajpy.12070

- Davis, K. & Hustvedt, G. (2012). It's a Matter of Control: Saving for Retirement. International Review of Social Sciences and Humanities, 3(2), 248-261.
- Davis, L. L. (1992). Instrument Review: Getting the Most From a Panel of Experts. *Applied Nursing Research*, *5*, 104-107.
- Deci, E. L. & Ryan, R. M. (2000). The "What" And "Why" Of Goal Pursuits: Human Needs And The Self-Determination Of Behavior. *Psychological inquiry*, 11(4), 227-268.
- Delpachitra, S. & Beal, D. (2002). Factors influencing planning for retirement. *Economic Papers: A journal of applied economics and policy, 21*(3), 1-13.
- Denton, F. T. & Spencer, B. G. (2009). What Is Retirement? A Review and Assessment of Alternative Concepts and Measures. *Canadian Journal on Aging*, 28(1), 63-76.
- Denton, M. A., French, S., Gafni, A., Joshi, A., Rosenthal, C. J. & Webb, S. (2001). Reflexive Planning For Later Life: A Conceptual Model and Evidences from Canada (Social And Economic Dimensions of an Aging Population Research Paper No.44). Hamilton: McMaster University.
- Denton, M. A., Kemp, C. L., French, S., Gafni, A., Joshi, A., Rosenthal, C. J. & Davies, S. (2004). Reflexive planning for later life. *Canadian Journal on Aging*, 23(Supplement 1), S71-S82.
- Department of Statistics Malaysia. (2015). Population Distribution and Basic Demographic Characteristic Report 2010. Retrieved 3 August, 2020, from https://www.dosm.gov.my/
- Department of Statistics Malaysia. (2016a). Principal Statistics of Education Services, 1971-2010. Retrieved 13 March, 2019, from http://www.data.gov.my/data/en_US/dataset/services-statistics-1022
- Department of Statistics Malaysia. (2016b). Services Statistics Education (Private Sector) 2015. Retrieved 11 June, 2020, from <u>www.dosm.gov.my</u>
- DeVaney, S. & Chiremba, S. (2005). Comparing the retirement savings of the Baby Boomers and other cohorts. Retrieved 8 October, 2016, from <u>http://www.bls.gov/opub/mlr/cwc/comparing-the-retirement-savings-of-the-baby-boomers-and-other-cohorts.pdf</u>
- Donald, I. & Cooper, S. R. (2001). A Facet Approach to Extending the Normative Component of the Theory of Reasoned Action. *British Journal of Social Psychology*, 40, 599-621.
- Dulebohn, J. H. (2002). An Investigation of the Determinants of Investment Risk Behavior in Employer-Sponsored Retirement Plans. *Journal of Management, 28*(1), 3-26.
- Dulebohn, J. H. & Murray, B. (2007). Retirement Savings Behavior of Higher Education Employees. *Research in Higher Education*, 48(5), 545-582. doi: 10.1007/s11162-006-9038-z

- Dzhogleva, H. & Lamberton, C. P. (2014). Should Birds of a Feather Flock Together? Understanding Self-Control Decisions in Dyads. *Journal of Consumer Research*, 41. doi: 10.1086/676599
- Eagly, A. H. & Chaiken, S. (1993). *The psychology of attitudes*. Orlando: Harcourt Brace Jovanovich College Publishers.
- Eagly, A. H. & Chaiken, S. (1998). Attitude, Structure and Function. In D. T. Gilbert, S. T. Fisk & G. Lindsey (Eds.), *Handbook of Social Psychology* (pp. 269-322). New York: McGraw-Hill.
- Ekerdt, D. J. (2010). Frontiers of Research on Work and Retirement. *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences*, 65B(1), 69-80. doi: 10.1093/geronb/gbp109
- Ekerdt, D. J. & Baker, L. A. (2014). The Material Convoy After Age 50. The Journals of Gerontology Series B: Psychological Sciences and Social Sciences, 69(3), 442-450. doi: 10.1093/geronb/gbu003
- Ekerdt, D. J., De Viney, S. & Kosloski, K. (1996). Profiling Plans for Retirement. The Journals of Gerontology Series B: Psychological Sciences and Social Sciences, 51B(3), S140-S149. doi: 10.1093/geronb/51B.3.S140
- Ekerdt, D. J., Hackney, J., Kosloski, K. & DeViney, S. (2001). Eddies in the Stream: The Prevalence of Uncertain Plans for Retirement. *The Journals of Gerontology Series* B: Psychological Sciences and Social Sciences, 56(3), S162-S170. doi: 10.1093/geronb/56.3.S162
- Ellen, P. S., Wiener, J. L. & Fitzgerald, M. P. (2012). Encouraging People to Save for Their Future: Augmenting Current Efforts with Positive Visions of the Future. *Journal of Public Policy & Marketing*, 31(1), 58-72. doi: 10.1509/jppm.09.089
- Employee Benefit Research Institute. (2018). 2018 Retirement Confidence Survey. Retrieved 14 July, 2018, from <u>https://www.ebri.org/surveys/rcs/2018/</u>
- Employees Provident Fund. (2015). EPF Savings and Your Retirement. Retrieved 2 June, 2016, from <u>http://www.kwsp.gov.my/portal/en/web/kwsp/member/epf-savings-and-your-retirement</u>
- Falk, R. F. & Miller, N. B. (1992). *A Primer for Soft Modeling*. Akron, Ohio: University of Akron Press.
- Farmer, J. D. & Geanakoplos, J. (2009). *Hyperbolic discounting is rational: Valuing the far future with uncertain discount rates* (Cowles Foundation Discussion Paper No. 1719): Cowles Foundation For Research In Economics, Yale University.
- Fava, J. L. & Velicer, W. F. (1992). The effects of overextraction on factor and component analysis. *Multivariate Behavioral Research*, 27, 387-415.
- Federation of Malaysian Consumers Associations (FOMCA). (2016). A Survey of Financial Behaviours and Financial Habits of Young Working Malaysians. Retrieved 14 July, 2018, from https://mfpc.org.my/

- Fernandes, D., Lynch, J. G. & Netemeyer, R. G. (2014). Financial Literacy, Financial Education, and Downstream Financial Behaviors. *Management Science*, 60(8), 1861-1883. doi: 10.2307/42919641
- Fernández-López, S., Otero, L., Vivel, M. & Rodeiro, D. (2010). What Are the Driving Forces of Individuals' Retirement Savings? *Finance a Uver: Czech Journal of Economics & Finance*, 60(3), 226-251.
- Ferraro, K. F., Shippee, T. P. & Schafer, M. H. (2009). Cumulative Inequality Theory for Research on Aging and the Life Course. In V. L. Bengtson, M. Silverstein, N. M. Putney & D. Gans (Eds.), *Handbook Of Theories Of Aging*. New York: Springer Publishing Company.
- Financial Planning Association of Australia. (2014). The Future of the Financial Planning Profession "The Future Profession". *White Paper of the Financial Planning Association of Australia (May 2014)*. Retrieved 3 April, 2019, from <u>https://fpa.com.au/advocacy/fpa-10-point-plan/</u>
- Financial Planning Association Of Malaysia. (2014). Financial Planning in General. Retrieved 22 November, 2014, from <u>http://www.fpam.org.my</u>
- Financial Planning Standards Board. (2015). Financial Planning Standards Board. Retrieved 23 May, 2015, from <u>https://www.fpsb.org/</u>
- Fishbein, M. (1963). An Investigation of the Relationships Between Beliefs about an Object and the Attitude Toward That Object. *Human Relations*, 16, 233-240.
- Fishbein, M. & Ajzen, I. (1975). *Belief, attitude, intention, and behavior: An introduction to theory and research*. Reading, Massahusetts: Addison-Wesley.
- Fishbein, M. & Ajzen, I. (2010). Predicting and changing behavior: The reasoned action approach. New York: Psychology Press.
- Folk, J. Y., Beh, L. S. & Baranovich, D.-L. (2012). Financial education: Determinant of retirement planning in Malaysia. *Journal of Business Management and Economics*, 3(2), 69-78.
- Fontes, A. (2011). Differences in the Likelihood of Ownership of Retirement Saving Assets by the Foreign and Native-Born. *Journal of Family and Economic Issues, 32*(4), 612-624.
- Fornell, C. & Larcker, D. F. (1981). Evaluating Structural Equation Models with Unobservable Variables and Measurement Error. *Journal of Marketing Research*, *18*(1), 39-50.
- Francis, J., Eccles, M. P., Johnston, M., Walker, A. E., Grimshaw, J. M., Foy, R., Kaner, E., Smith, L. & Bonetti, D. (2004). *Constructing questionnaires based on the theory* of planned behaviour: A manual for health services researchers. Newcastle upon Tyne, UK: Centre for Health Services Research, University of Newcastle upon Tyne.

- Friedman, S. L. & Scholnick, E. K. (1997a). *The Developmental Psychology of Planning: Why, How, and when Do We Plan?* New York: Psychology Press.
- Friedman, S. L. & Scholnick, E. K. (1997b). An Evolving "Blueprint" for Planning: Psychological Requirement, Task Characteristics, and Social-Cultural Influences. In S. L. Friedman & E. K. Scholnick (Eds.), *The Developmental Psychology of Planning: Why, How, and when Do We Plan?* (pp. 3-22). New York: Psychology Press.
- Frieze, I. H., Olson, J. E. & Murrell, A. J. (2011). Working Beyond 65: Predictors of Late Retirement for Women and Men MBAs. *Journal of Women & Aging*, 23(1), 40-57. doi: 10.1080/08952841.2011.540485
- Furnham, A. (1984). Many sides of the coin: The psychology of money usage. *Personality* and Individual Differences, 5(5), 501-509. doi: <u>http://dx.doi.org/10.1016/0191-8869(84)90025-4</u>
- Furnham, A. & Goletto-Tankel, M.-P. (2002). Understanding savings, pensions and life assurance in 16-21-year-olds. *Human Relations*, 55(5), 603-628.
- Garcia, K. & Mann, T. (2003). From 'I Wish' to 'I Will': Social-Cognitive Predictors of Behavioural Intentions. *Journal of Health Psychology*, 8(3), 347-360.
- Gefen, D., Straub, D. W. & Boudreau, M.-C. (2000). Structural equation modelling and regression: Guidelines for research practice. *Communication of the Association for Information Systems*, 4(7), 2-77.
- Geisser, S. (1975). The Predictive Sample Reuse Method with Applications. *Journal of the American Statistical Association*, 70(350), 320-328.
- Gerrans, P. & Hershey, D. A. (2013). *The role of financial literacy and financial adviser anxiety in older Australians' advice seeking*. Canberra: National Seniors Productive Ageing Centre.
- Gerrans, P. & Hershey, D. A. (2016). Financial Adviser Anxiety, Financial Literacy and Financial Advice Seeking. *Journal of Consumer Affairs, Online advance access*. doi: 10.1111/joca.12120
- Giele, J. Z. & Elder, G. H., Jr. (1998). *Methods of life course research: Qualitative and quantitative approaches*. London: Sage.
- Giles, M., McClenahan, C., Cairns, E. & Mallet, J. (2004). An Application of the Theory of Planned Behaviour to Blood Donation: The Importance of Self-Efficacy. *Health Education Research*, 19(4), 380-391.
- Glass, J. C. & Kilpatrick, B. (1998). Gender comparisons of baby boomers and financial preparation for retirement. *Educational Gerontology*, *24*, 719-745.
- Godin, G. & Kok, G. (1996). The theory of planned behavior: A review of its applications to health-related behaviors. *American Journal of Health Promotion*, 11(2), 87-98.

- Goldberg, H. & Lewis, R. T. (1978). Money Madness: The Psychology of Saving, Spending, Loving, and Hating Money. London: Springwood Books.
- Gollwitzer, P. M. & Oettingen, G. (2011). Planning Promotes Goal Striving. In K. D. Vohs
 & R. F. Baumeister (Eds.), *Handbook Of Self-Regulation: Research, Theory, And Applications* (pp. 162-185). New York: Guilford Pres.
- Gotz, O., Liehr-Gobbers, K. & Krafft, M. (2010). Evaluation of Structural Equation Models using the Partial Least Squares (PLS) Approach. In V. E. Vinzi, W. W. Chin, J. Henseler & H. Wang (Eds.), *Handbook of Partial Least Squares: Concepts, Methods and Applications* (pp. 691-711). Heidelberg: Springer.
- Grable, J. E., Archuleta, K. L., Roy, N. & Roudi, R. (2011). *Financial Planning and Counseling Scales*. New York: Springer.
- Grant, J. S. & Davis, L. L. (1997). Selection and Use of Content Experts for Instrument Development. *Research in Nursing & Health, 20*, 269-274.
- Greenwell, L. & Bengtson, V. L. (1997). Geographic Distance and Contact Between Middle-aged Children and Their Parents: The Effects of Social Class Over 20 Years. *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences, 52B*(1), S13-S26. doi: 10.1093/geronb/52B.1.S13
- Griffin, B., Loh, V. & Hesketh, B. (2013). A mental model of factors associated with subjective life expectancy. *Social Science and Medicine*, *82*, 79-86. doi: 10.1016/j.socscimed.2013.01.026
- Gupta, R. & Hershey, D. A. (2016). Cross-National Differences in Goals for Retirement: the Case of India and the United States. *Journal of Cross-cultural Gerontology*, 2016(31), 221–236.
- Gutierrez, H. C. & Hershey, D. A. (2014). Age Differences In Expected Satisfaction With Life In Retirement. International Journal of Aging & Human Development, 78(2), 93-114.
- Hadar, L., Sood, S. & Fox, C. R. (2013). Subjective Knowledge in Consumer Financial Decisions. *Journal of Marketing Research*, 50(3), 303-316. doi: 10.1509/jmr.10.0518
- Hagger, M. S., Chatzisarantis, N. L. D. & Biddle, S. J. H. (2002). A meta-analytic review of the theories of reasoned action and planned behavior in physical activity: An examination of predictive validity and the contribution of additional variables. *Journal of Sport and Exercise Psychology*, 24(1), 3-32.
- Hair, J. F., Black, W. C., Babin, B. J. & Anderson, R. E. (2010). Multivariate Data Analysis: A Global Perspective (7th ed.). Upper Saddle River: Pearson Prentice Hall.
- Hair, J. F., Hult, G. T. M., Ringle, C. M. & Sarstedt, M. (2017). A Primer on Partial Least Squares Structural Equation Modeling (PLS-SEM) (2nd ed.). Los Angeles: Sage Publications, Inc.

- Hair, J. F., Ringle, C. M. & Sarstedt, M. (2011). PLS-SEM: Indeed a silver bullet. *Journal* of Marketing Theory and Practice, 19(2), 139-115.
- Hair, J. F., Risher, J. J., Sarstedt, M. & Ringle, C. M. (2019). When to use and how to report the results of PLS-SEM. *European Business Review*, 31(1), 2-24. doi: 10.1108
- Hair, J. F., Sarstedt, M., Ringle, C. M. & Mena, J. A. (2011). An assessment of the use of partial least squares structural equation modeling in marketing research. *Academy* of Marketing Science, 40(3), 414-433.
- Hardré, P. L. & Reeve, J. (2009). Training corporate managers to adopt a more autonomysupportive motivating style toward employees: an intervention study. *International Journal of Training & Development*, 13(3), 165-184. doi: 10.1111/j.1468-2419.2009.00325.x
- Harris, M. N., Loundes, J. & Webster, E. (2002). Determinants of household saving in Australia. *Economic Record*, 78(241), 207-223.
- Harrison, D. (2005). *Personal Financial Planning: Theory and Practice*. Essex: Pearson Education Limited.
- Harrison, T., Waite, K. & White, P. (2006). Analysis by paralysis: the pension purchase decision process. *International Journal of Bank Marketing*, 24(1), 5-23. doi: 10.1108/02652320610642317
- Hassan, Z. A., Schattner, P. & Mazza, D. (2006). Doing A Pilot Study: Why Is It Essential? *Malaysian Family Physician*, 1(2-3), 70-73.
- Hastings, J. S., Madrian, B. C. & Skimmyhorn, W. L. (2013). Financial literacy, financial education, and economic outcomes. *Annual review of economics*(5), 347-373.
- Hayes-Roth, B. & Hayes-Roth, F. (1979). A cognitive model of planning. *Cognitive* science, 3(4), 275-310.
- Hellevik, T. & Settersten, R. A. (2013). Life Planning among Young Adults in 23 European Countries: The Effects of Individual and Country Security. *European Sociological Review*, 29(5), 923-938. doi: 10.1093/esr/jcs069
- Henseler, J., Hubona, G. & Ray, P. A. (2016). Using PLS path modeling in new technology research: updated guidelines. *Industrial Management & Data Systems, 116*(1), 2-20.
- Henseler, J., Ringle, C. M. & Sarstedt, M. (2015). A new criterion for assessing discriminant validity in variance-based structural equation modeling. *Journal of the Academy of Marketing Science*, 43, 115-135.
- Henseler, J., Ringle, C. M. & Sinkovics, R. (2009). The use of Partial Least Squares path modeling in International Marketing. *International Marketing*, 20, 277-319.
- Heraty, N. & McCarthy, J. (2015). Unearthing Psychological Predictors of Financial Planning for Retirement Among Late Career Older Workers: Do Self-perceptions

of Aging Matter? *Work, Aging and Retirement, 1*(3), 274-283. doi: 10.1093/workar/wav008

- Hershey, D., Jacobs-Lawson, J., McArdle, J. & Hamagami, F. (2007). Psychological Foundations of Financial Planning for Retirement. *Journal of Adult Development*, 14, 26-36. doi: 10.1007/s10804-007-9028-1
- Hershey, D. & Walsh, D. (2000). Knowledge versus Experience in Financial Problem Solving Performance. *Current Psychology*, 19(4), 261.
- Hershey, D. A. (2004). Psychological influences on the retirement investor. CSA: Certified Senior Advisor, 22, 31-39.
- Hershey, D. A., Austin, J. T. & Gutierrez, H. C. (2015). Financial Decision Making across the Adult Life Span: Dynamic Cognitive Capacities and Real-World Competence. In T. M. Hess, J. Strough & C. E. Lockenhoff (Eds.), *Aging and Decision Making: Empirical and Applied Perspectives* (pp. 329-349). New York: Academic Press.
- Hershey, D. A. & Henkens, K. (2014). Impact of Different Types of Retirement Transitions on Perceived Satisfaction with Life. *The Gerontologist*, *54*(2), 232-244. doi: 10.1093/geront/gnt006
- Hershey, D. A., Henkens, K. & Van Dalen, H. P. (2007). Mapping the Minds of Retirement Planners: A Cross-Cultural Perspective. *Journal of Cross-Cultural Psychology*, 38(3), 361-382. doi: 10.1177/0022022107300280
- Hershey, D. A. & Jacobs-Lawson, J. M. (2009). Goals for retirement: Content, structure and process. In R. R. Brougham (Ed.), *New Directions in Aging Research* (pp. 167-189). Hauppage, NY: Nova Science Publishers.
- Hershey, D. A. & Jacobs-Lawson, J. M. (2012). Bridging the Gap: Anticipated Shortfalls in Future Retirement Income. *Journal of Family and Economic Issues*, *33*(3), 306-314. doi: 10.1007/s10834-012-9281-8
- Hershey, D. A., Jacobs-Lawson, J. M. & Austin, J. T. (2013). Effective financial planning for retirement. In M. Wang (Ed.), *Oxford Handbook of Retirement* (pp. 402-430). New York: Oxford University Press.
- Hershey, D. A., Jacobs-Lawson, J. M. & Neukam, K. A. (2002). Influences of age and gender on workers goals for retirement. *International Journal of Aging & Human Development*, 55, 163-179.
- Hershey, D. A., Jacobs-Lawson, J. M. & Walsh, D. A. (2003). Influences of Age and Training on Script Development. Aging, Neuropsychology, and Cognition: A Journal on Normal and Dysfunctional Development, 10(1), 1-19.
- Hershey, D. A., Mowen, J. C. & Jacobs-Lawson, J. M. (2003). An Experimental Comparison Of Retirement Planning Intervention Seminars. *Educational Gerontology*, 29(4), 339-359.

- Hershey, D. A., Walsh, D. A., Brougham, R., Carter, S. & Farrell, A. H. (1998). Challenges of training pre-retirees to make sound financial planning decisions. *Educational Gerontology*, 24(5), 447-470.
- Hilgert, M. A., Hogarth, J. M. & Beverly, S. G. (2003). Household financial management: the connection between knowledge and behavior. *Federal Reserve Bulletin*, 89(7), 309-322.
- Hira, T. K., Rock, W. L. & Loibl, C. (2009). Determinants of retirement planning behaviour and differences by age. *International Journal of Consumer Studies*, 33(3), 293-301. doi: 10.1111/j.1470-6431.2009.00742.x
- Holzmann, R. (2015). Participation in Mandated and Voluntary Social Risk Management Arrangements: The Role and Limits of Financial Education and Other Interventions. In M. Frölich, D. Kaplan, C. Pagés, J. Rigolini & D. Robalino (Eds.), Social Insurance, Informality, and Labor Markets: How to Protect Workers While Creating Good Jobs. Oxford: Oxford University Press.
- Holzmann, R. & Hinz, R. (2005). Old Age Income Support In The 21st Century: An International Perspective On Pension Systems And Reform. Washington, DC: World Bank.
- Holzmann, R., Mac Arthur, I. W. & Sin, Y. (2000). Pension Systems in East Asia and the Pacific: Challenges and Opportunities (Social Protection Discussion Paper Series No. 0014). Washington: The World Bank.
- Horn, J. L. (1965). A rationale and test for the number of factors in factor analysis. *Psychometrika*, 30, 179-185.
- Hung, A. A., Parker, A. M. & Yoong, J. K. (2009). *Defining and Measuring Financial Literacy* (Working paper for RAND Labor and Population). California: RAND Corporation.
- Huston, S. J. (2010). Measuring Financial Literacy. *Journal of Consumer Affairs*, 44(2), 296-316. doi: 10.1111/j.1745-6606.2010.01170.x
- Igbaria, M., Iivari, J. & Maragahh, H. (1995). Why do individuals use computer technology? A Finnish case study. *Information and Management*, *5*, 227-238.
- In, J. (2017). Introduction of a pilot study. *Korean journal of anesthesiology*, 70(6), 601-605. doi: 10.4097/kjae.2017.70.6.601
- International Organization for Standardization. (2005). Personal financial planning-Requirements for personal financial planners (ISO22222:2005). Retrieved 2 June, 2016, from <u>http://www.iso.org/</u>
- Jacobs-Lawson, J. M. & Hershey, D. A. (2003). Perceptions of financial stability in retirement: Do Americans really know what to expect? In S. P. Shohov (Ed.), *Advances in Psychology Research* (pp. 123-136). New York: NOVA Publishing Company.

- Jarvis, C. B., Mackenzie, S. B. & Podsakoff, P. M. (2003). A Critical Review of Construct Indicators and Measurement Model Misspecification in Marketing and Consumer Research. *Journal Of Consumer Research*, 30, 199-218.
- Johnson, M. K., Crosnoe, R. & Elder, G. H. (2011). Insights on Adolescence From a Life Course Perspective. Journal of Research on Adolescence, 21(1), 273-280. doi: 10.1111/j.1532-7795.2010.00728.x
- Joo, S. & Grable, J. E. (2005). Employee education and the likelihood of having a retirement savings program. *Financial Counseling and Planning*, 16(1), 37-50.
- Jump\$tart Coalition for Personal Financial Literacy. (2018). Frequently Asked Questions. Retrieved 9 August, 2018, from <u>https://www.jumpstart.org/who-we-are/jumpstart-faqs/</u>
- Kaiser, H. (1974). An index of factorial simplicity. Psychometrika, 39, 31-36.
- Karpinska, K., Henkens, K. & Schippers, J. (2011). The recruitment of early retirees: a vignette study of the factors that affect managers' decisions. *Ageing & Society*, 31, 570-589. doi: 10.1017/s0144686x10001078
- Kemp, C. L., Rosenthal, C. J. & Denton, M. (2005). Financial planning for later life: Subjective understandings of catalysts and constraints. *Journal of Aging Studies*, 19(3), 273-290. doi: 10.1016/j.jaging.2004.08.004
- Kennedy, J. & Matwijiw, P. (2010). Retirement Savings: A Consumer Perspective. *Australian Economic Review*, 43(3), 321-325. doi: 10.1111/j.1467-8462.2010.00603.x
- Khazanah Research Institute. (2016). Budget 2020, The state of household 2018-different realities. Report of household income and basic amenities survey. Malaysia: Ministry of Finance.
- Kim, J. E. & Moen, P. (2001). Moving into Retirement: Preparation and Transitions in Late Midlife. In M. E. Lachman (Ed.), *Handbook of Midlife Development* (pp. 487-527). New York: John Wiley & Sons, Inc.
- Kimiyaghalam, F., Mansori, S., Safari, M. & Yap, S. (2017). Parents' Influence on Retirement Planning in Malaysia. *Family and Consumer Sciences Research Journal*, 45(3), 315-325.
- Kiso, H. & Hershey, D. A. (2017). Working Adults' Metacognitions Regarding Financial Planning for Retirement. *Work, Aging and Retirement, 3*(1), 77-88. doi: 10.1093/workar/waw021
- Klobas, J. & Ajzen, I. (2015). Making the Decision to Have a Child. In D. Philipov, A. C. Liefbroer & J. Klobas (Eds.), *Reproductive Decision-Making in a Macro-Micro Perspective*. New York: Springer.
- Koch, A. K. & Nafziger, J. (2011). Self-regulation through Goal Setting. *Scandinavian Journal of Economics*, 113(1), 212-227. doi: 10.1111/j.1467-9442.2010.01641.x

Koh, S. K. & Fong, W. M. (2000). Personal Financial Planning. Singapore: Prentice Hall.

- Koposko, J. L., Bojórquez, M. I., Pérez, A. E. & Hershey, D. A. (2016). College Student Attitudes toward Retirement Planning: The Case of Mexico and the United States. *Journal of Personal Finance*, 15(2), 52-67.
- Koposko, J. L. & Hershey, D. A. (2014). Parental and Early Influences on Expectations of Financial Planning for Retirement. *Journal of Personal Finance*, 13(2), 17-27.
- Koposko, J. L. & Hershey, D. A. (2016). When I First Learned about Retirement: Financial and Retirement Concept Recognition among College Students. *Current Psychology*, 35(4), 540-548. doi: 10.1007/s12144-015-9319-9
- Koposko, J. L., Kiso, H., Hershey, D. A. & Gerrans, P. (2016). Perceptions of Retirement Savings Relative to Peers. *Work, Aging and Retirement, 2*(1), 65-72.
- Kramer, M. M. (2012). Financial Advice and Individual Investor Portfolio Performance. *Financial Management*, *41*(2), 395-428. doi: 10.1111/j.1755-053X.2012.01185.x
- KRC Research. (2016). 2016 Consumer Finance Segmentation Survey. Retrieved 14 July, 2018, from <u>https://www.cfp.net/news-events/research-facts-figures/cfp-board-research/consumer-surveys</u>
- Krejcie, R. V. & Morgan, D. W. (1970). Determining Sample Size For Research Activities. Educational And Psychological Measurement, 30, 607-610.
- Lai, M.-M. & Tan, W.-K. (2009). An Empirical Analysis of Personal Financial Planning in an Emerging Economy. *European Journal of Economics, Finance and Administrative Sciences, 16*(1), 99-111.
- Lapinski, M. K., Kerr, J. M., Zhao, J. H. & Shupp, R. S. (2017). Social Norms, Behavioral Payment Programs, and Cooperative Behaviors: Toward a Theory of Financial Incentives in Normative Systems. *Human Communication Research*, 43(1), 148-171. doi: 10.1111/hcre.12099
- Lazarus, R. S. & Folkman, S. (1984). Stress, Appraisal, and Coping. New York: Springer.
- Lown, J. (2008). Retirement savings adequacy for the baby boom generation. *Journal of Personal Finance*, 7(1), 109-134.
- Lusardi, A. & Mitchell, O. S. (2007). Financial Literacy and Retirement Preparedness: Evidence and Implications for Financial Education. *Business Economics*, 42(1), 35-44.
- Lusardi, A. & Mitchell, O. S. (2010). How ordinary consumers make complex economic decisions: Financial literacy and retirement readiness (CFS Working Paper No. 2010/11). Frankfurt: Center for Financial Studies.
- Lusardi, A. & Mitchell, O. S. (2011a). Financial Literacy and Planning: Implications for Retirement Well-Being. In O. S. Mitchell & A. Lusardi (Eds.), *Financial Literacy: Implications for Retirement Security and the Financial Marketplace* (pp. 17-39). Oxford: Oxford University Press.

- Lusardi, A. & Mitchell, O. S. (2011b). Financial Literacy Around The World: An Overview. *Journal of Pension Economics & Finance, 10*(4), 497-508.
- Lusardi, A. & Mitchell, O. S. (2011c). The Outlook for Financial Literacy. In O. S. Mitchell & A. Lusardi (Eds.), *In Financial Literacy: Implications for Retirement Security and the Financial Marketplace* (pp. 1–15). Oxford: Oxford University Press.
- Lusardi, A. & Mitchell, O. S. (2014). The Economic Importance of Financial Literacy: Theory and Evidence. *Journal of Economic Literature*, 52(1), 5-44. doi: 10.1257/jel.52.1.5
- Lusardi, A. & Mitchell, O. S. (2017). How Ordinary Consumers Make Complex Economic Decisions: Financial Literacy and Retirement Readiness. *Quarterly Journal of Finance*, 7(3), 1-31. doi: 10.1142/s2010139217500082
- Lusardi, A., Mitchell, O. S. & Curto, V. (2010). Financial Literacy among the Young. Journal of Consumer Affairs, 44(2), 358-380. doi: 10.1111/j.1745-6606.2010.01173.x
- Lusardi, A., Schneider, D. & Tufano, P. (2011). Financially Fragile Households: Evidence and Implications. *Brookings Papers on Economic Activity, Spring*, 83-150.
- Lynn, M. (1986). Determination and quantification of content validity. *Nursing Research*, *35*, 382-385.
- MacDonald, B.-J., Jones, B., Morrison, R. J., Brown, R. L. & Hardy, M. (2011). Research and Reality – A Literature Review on Drawing Down Retirement Savings. Retrieved 11 August, 2016, from <u>http://www.soa.org/research/researchprojects/pension/default.aspx</u>.
- Madden, T. J., Ellen, P. S. & Ajzen, I. (1992). A Comparison of the Theory of Planned Behavior and the Theory of Reasoned Action. *Personality and Social Psychology Bulletin, 18*(1), 3-9.
- Malaysian Qualifying Agency. (2009). Malaysian Qualifications Register. Retrieved 12 June, 2016, from <u>http://www.mqa.gov.my/mqr/</u>
- Mandell, L. (2007). Financial literacy of high school students. In J. J. Xiao (Ed.), *Handbook* of Consumer Finance Research (pp. 163-183). New York: Springer.
- Marsden, M., Zick, C. D. & Mayer, R. N. (2011). The Value of Seeking Financial Advice. Journal of Family and Economic Issues, 32(4), 625-643.
- Martin, M. (2007). A Literature Review on the Effectiveness of Financial Education (Working Paper No. 07-03). Richmond: Federal Reserve Bank of Richmond.
- McDonald, L. & Donahue, P. (2011). Retirement Lost? *Canadian Journal on Aging*, *30*(3), 401-422. doi: 10.1177/0149206309352246
- Mermin, G. B. T., Johnson, R. W. & Murphy, D. P. (2007). Why do boomers plan to work longer? *Journals of Gerontology: Series B: Psychological Sciences and Social Sciences, 62B*(5), S286-S294.

- Meyers, L. S., Gamst, G. & Guarino, A. J. (2006). *Applied Multivariate Research: Design and Interpretation*. Thousand Oaks, California: Sage Publications.
- Millar, M. & Devonish, D. (2009). Attitudes, savings choices, level of knowledge and investment preferences of employees toward pensions and retirement planning: Survey evidence from Barbados. *Pensions: An International Journal*, 14(4), 299-317. doi: 10.1057/pm.2009.17
- Ministry of Education Malaysia. (2014a). *National Education Statistic: Higher Education Sector 2013*. Putrajaya: Ministry of Education Malaysia.
- Ministry of Education Malaysia. (2014b). Quick Facts 2014 Malaysia Educational Statistics. Retrieved 10 June, 2016, from <u>http://emisportal.moe.gov.my</u>
- Ministry of Finance. (2020). Annual Budget. Retrieved 3 February, 2020, from <u>http://www.treasury.gov.my/</u>
- Mintzberg, H. (1967). The science of strategy-making. *Industrial Management Review*, 8(2), 71.
- Mintzberg, H. (1990). The design school: Reconsidering the basic premises of strategic management. *Strategic Management Journal*, 11(3), 171-195.
- Mintzberg, H. & Waters, J. A. (1985). Of Strategies, Deliberate and Emergent. *Strategic Management Journal*, 6(3), 257-272. doi: 10.2307/2486186
- Mintzberg, H. & Westley, F. (1992). Cycles of Organizational Change. Strategic Management Journal, 13, 39-59. doi: 10.2307/2486365
- Mitchell, O. & Utkus, S. (2004). Lessons from Behavioral Finance for Retirement Plan Design. In O. S. Mitchell & S. P. Utkus (Eds.), *Pension design and structure: New lessons from behavioral finance* (pp. 3–42). Oxford: Oxford University Press.
- Modigliani, F. & Brumberg, R. (1954). Utility analysis and the consumption function: An interpretation of cross section data. In K. Kurihara (Ed.), *Post keynesian economics*. New Brunswick, NJ: Rutgers University Press.
- Moen, P., Huang, Q. L., Plassmann, V. & Dentinger, E. (2006). Deciding the future Do dual-earner couples plan together for retirement? *American Behavioral Scientist*, 49(10), 1422-1443. doi: 10.1177/0002764206286563
- Morgan, D. (1998). Practical Strategies for Combining Qualitative and Quantitative Methods: Application to Health Research. *Qualitative Health Research*, 8(3), 362-376.
- Morgan, L. A. & Eckert, J. K. (2004). Retirement Financial Preparation. Journal of Aging & Social Policy, 16(2), 19-34. doi: 10.1300/J031v16n02_02
- Natalio, M. A., Faria, C. D. C. M., Teixeira-Salmela, L. F. & Michaelsen, S. M. (2014). Content Validation Of A Clinical Assessment Instrument For Stair Ascent And Descent In Individuals With Hemiparesis. *Brazilian Journal Physical Therapy*, 18(4), 353-363. doi: 10.1590/bjpt-rbf.2014.0052

- Neukam, K. A. & Hershey, D. A. (2003). Financial Inhibition, Financial Activation, and Saving for Retirement. *Financial Services Review*, 12(1), 19-37.
- Neuman, W. (2006). *Social Research Methods: Qualitative and Quantitative Approaches*. Boston: Pearson Education.
- Noone, J., Alpass, F. & Stephens, C. (2010). Do Men and Women Differ in Their Retirement Planning? Testing a Theoretical Model of Gendered Pathways to Retirement Preparation. *Research on Aging*, 32(6), 715-738. doi: 10.1177/0164027510383531
- Noone, J., O'Loughlin, K. & Kendig, H. (2012). Socioeconomic, psychological and demographic determinants of Australian baby boomers' financial planning for retirement. *Australasian Journal on Ageing*, *31*(3), 194-197.
- Noone, J. H., Stephens, C. & Alpass, F. (2010). The Process of Retirement Planning Scale (PRePS): Development and Validation. *Psychological Assessment*, 22(3), 520-531.
- Noone, J. H., Stephens, C. & Alpass, F. M. (2009). Preretirement Planning and Well-Being in Later Life: A Prospective Study. *Research on Aging*, *31*(3), 295-317. doi: 10.1177/0164027508330718
- Nosi, C., D'Agostino, A., Pagliuca, M. M. & Pratesi, C. A. (2014). Saving for old age: Longevity annuity buying intention of Italian young adults. *Journal of Behavioral* and Experimental Economics, 51, 85-98. doi: 10.1016/j.socec.2014.05.001
- Nosi, C., D'Agostino, A., Pagliuca, M. & Pratesi, C. A. (2017). Securing Retirement at a Young Age. Exploring the Intention to Buy Longevity Annuities through an Extended Version of the Theory of Planned Behavior. *Sustainability*, 9(6), 1-20.
- Notani, A. S. (1998). Psychology Moderators of Perceived Behavioral Control's Predictiveness in the Theory of Planned Behavior: A Meta-Analysis. *Journal of Consumer Psychology*, 7(3), 247-271.
- Nunnally, J. C. (1978). Psychometric Theory. New York: McGraw-Hill.
- O'Donoghue, T. & Rabin, M. (1999). Procrastination in Preparing for Retirement. In H. Aaron (Ed.), *Behavioral dimensions of retirement economics* (pp. 125-156). Washington D.C.: Brookings Institution Press
- O'Donoghue, T. & Rabin, M. (2001). Choice And Procrastination. Quarterly Journal of Economics, 116(1), 121-160. doi: 10.1162/003355301556365
- Ogden, J. (2015). Time to retire the theory of planned behaviour?: one of us will have to go! A commentary on Sniehotta, Presseau and Araújo-Soares. *Health Psychology Review*, 9(2), 165-167. doi: 10.1080/17437199.2014.898679
- Organisation for Economic Co-operation and Development. (2016). Household savings. Retrieved 2 June, 2016, from <u>https://data.oecd.org/hha/household-savings.htm</u>

- Organization for Economic Co-Operation and Development. (2005). Improving Financial Literacy: Analysis of Issues and Policies. Retrieved 15 August, 2018, from http://www.oecd.org/finance/financial-education/
- Osgood, C. E., Suci, G. J. & Tannenbaum, P. H. (1957). *The Measurement of Meaning*. Urbana: University Illinois Press.
- Overton, R. H. (2008). Theories Of The Financial Planning Profession. *Journal of Personal Finance*, *7*(1), 13-41.
- Owusu-Frimpong, N., Omar, O. E. & Mmieh, F. (2011). An understanding of baby boomers' attitudes toward financial services products in Ghana. *Thunderbird International Business Review*, 53(1), 69-78. doi: 10.1002/tie.20390
- Padawer, E., Jacobs-Lawson, J., Hershey, D. & Thomas, D. (2007). Demographic Indicators as Predictors of Future Time Perspective. *Current Psychology*, 26(2), 102-108. doi: 10.1007/s12144-007-9008-4
- Pallant, J. (2016). SPSS Survival Manual (6th ed.). Berkshire: McGraw-Hill Education.
- Pearce, J. A. & Robinson, R. B. (2011). *Strategic management: formulation, implementation, and control* (12th ed.). New York: McGraw-Hill/Irwin.
- Petkoska, J. & Earl, J. K. (2009). Understanding the Influence of Demographic and Psychological Variables on Retirement Planning. *Psychology & Aging*, 24(1), 245-251.
- Petter, S., Straub, D. & Rai, A. (2007). Specifying Formative Constructs in Information Systems Research. *Management Information Systems Quarterly*, 31(4), 623-656.
- Pfau, W. D. & Kariastanto, B. D. (2012). An International Perspective on "Safe" Savings Rates for Retirement. *Journal of Financial Service Professionals*, 66(5), 62-71.
- Pham, M. T., Lee, L. & Stephen, A. T. (2012). Feeling the Future: The Emotional Oracle Effect. *Journal of Consumer Research*, 39(3), 461-477. doi: 10.1086/663823
- Phua, V. C. & McNally, J. W. (2008). Men Planning for Retirement: Changing Meanings of Preretirement Planning. *Journal of Applied Gerontology*, 27(5), 588-608.
- Podsakoff, P. M., MacKenzie, S. B., Lee, J.-Y. & Podsakoff, N. P. (2003). Common Method Biases in Behavioral Research: A Critical Review of the Literature and Recommended Remedies. *Journal of Applied Psychology*, 88(5), 879-903.
- Podsakoff, P. M., MacKenzie, S. B. & Podsakoff, N. P. (2012). Sources of Method Bias in Social Science Research and Recommendations on How to Control It. *Annual Review of Psychology*, 63, 539-569.
- Podsakoff, P. M. & Organ, D. W. (1986). Self-reports in Organizational Research: Problems and Prospects. *Journal of Management*, 12(4), 531-544.

- Polit, D. F. & Beck, C. T. (2006). The Content Validity Index: Are You Sure You Know What's Being Reported? Critique and Recommendations. *Research in Nursing & Health*, 29, 489-497.
- President's Advisory Council on Financial Literacy. (2008). 2008 Annual Report to the President. Retrieved 9 August, 2018, from <u>https://www.treasury.gov/resource-center/financial-education/Pages/commission-index.aspx</u>
- Princeton Survey Research Associates International. (2013). Financial Planning Profiles of American Households: The 2013 Household Financial Planning Survey and Index. Retrieved 14 July, 2018, from <u>https://www.cfp.net/news-events/research-facts-figures/cfp-board-research/consumer-surveys</u>
- Reeve, J., Ryan, R., Deci, E. L. & Jang, H. (2007). Understanding and promoting autonomous self-regulation: A self-determination theory perspective. In D. Schunk & B. Zimmerman (Eds.), *Motivation and self-regulated learning: Theory, research,* and application (pp. 223-244). Mahwah, New Jersey: Lawrence Erlbaum Associates.
- Remund, D. L. (2010). Financial Literacy Explicated: The Case for a Clearer Definition in an Increasingly Complex Economy. *Journal of Consumer Affairs*, 44(2), 276-295. doi: 10.1111/j.1745-6606.2010.01169.x
- Reuveni, Y. & Werner, P. (2015). Factors Associated with Teenagers' Willingness to Volunteer with Elderly Persons: Application of the Theory of Planned Behavior (TPB). *Educational Gerontology*, 1-12. doi: 10.1080/03601277.2015.1029768
- Rigdon, E. E. (2016). Choosing PLS path modeling as analytical method in European management research: A realist perspective. *European Management Journal*, 1-8.
- Rimal, R. N. & Lapinski, M. K. (2015). A Re-Explication of Social Norms, Ten Years Later. *Communication Theory*, 25(4), 393-409.
- Rivis, A., Sheeran, P. & Armitage, C. J. (2009). Expanding the affective and normative components of the theory of planned behaviour: A meta-analysis of anticipated affect and moral norms. *Journal of Applied Social Psychology*, 39(12), 2985-3019.
- Rivis, A. J. & Sheeran, P. (2004). Descriptive norms as an additional predictor in the theory of planned behaviour: A meta-analysis. *Current Psychology*, *22*, 264-280.
- Roscoe, J. T. (1975). *Fundamental research statistics for the behavioral sciences* (2nd ed.). New York: Holt, Rinehart and Winston.
- Rosenberg, M. J. & Hovland, C. I. (1960). Cognitive, Affective, and Behavioral Components in Attitudes. In C. I. Hovland & M. J. Rosenberg (Eds.), Attitude Organization and Change: An analysis of Consistency Among Attitude Components (pp. 1-14). New Haven: Yale University Press.
- Rowley, M. E., Lown, J. M. & Piercy, K. W. (2012). Motivating Women to Adopt Positive Financial Behaviors. *Journal of Financial Counseling and Planning*, 23(1), 47-62.

- Ryack, K. (2012). Evidence that time perspective factors depend on the group: Factor analyses of the CFC and ZTPI scales with professional financial advisors. *Personality and Individual Differences*, 52(6), 723-727.
- Santosa, P., Wei, K. & Chan, H. (2005). User Involvement and User Satisfaction with Information-Seeking Activity. *European Journal of Information Systems*, 14(4), 361-370.
- Saunders, M., Lewis, P. & Thornhill, A. (2009). *Research methods for business students* (5th ed.). Harlow: Pearson Education Limited.
- Schafer, J. L. (1997). *Analysis of Incomplete Multivariate Data*. New York: Chapman and Hall.
- Schulze, R. & Wittmann, W. W. (2003). A meta-analysis of the theory of reasoned action and the theory of planned behavior: The principle of compatibility and multidimensionality of beliefs as moderators. In R. Schulze, H. Holling & D. Böhning (Eds.), *Meta-analysis: New developments and applications in medical and social sciences* (pp. 219-250). Cambridge: Hogrefe and Huber Publishers.
- Schwitzgebel, E. (2006). Belief. *The Stanford Encyclopedia of Philosophy*. Retrieved 24 October, 2017, from <u>https://plato.stanford.edu/entries/belief/</u>
- Segel-Karpas, D. & Werner, P. (2014). Perceived Financial Retirement Preparedness and Its Correlates: A National Study in Israel. *International Journal of Aging & Human Development*, 79(4), 279-301. doi: 10.1177/0091415015574177
- Sekaran, U. (2003). *Research Methods for Business: A Skill Building Approach* (4th ed.). New York: John Wiley & Sons, Inc.
- Sekaran, U. & Bougie, R. (2010). Research Methods for Business: A Skill Building Approach (5th ed.). West Sussex: John Wiley & Sons, Inc.
- Sellin, N. & Keeves, J. P. (1994). Path Analysis with Latent Variables. In T. Husen & T. N. Postlethwaite (Eds.), *The International Encyclopedia of Education* (pp. 4352-4359). Oxford: Pergamon Press.
- Selnow, G. W. (2004). Motivating Retirement Planning: Problems and Solutions. In O. S. Mitchell & S. P. Utkus (Eds.), *Pension design and structure: New lessons from behavioral finance* (pp. 43–52). Oxford: Oxford University Press.
- Sheeran, P. (2002). Intention-Behavior Relations: A Conceptual and Empirical Review. In
 W. Stroebe & M. Hewstone (Eds.), *European Review of Social Psychology* (Vol. 12, pp. 1-36). West Sussex: John Wiley & Sons.
- Sheeran, P. & Orbell, S. (1999). Augmenting the theory of planned behavior: Roles for anticipated regret and descriptive norms. *Journal of Applied Social Psychology, 29*, 2107-2142.
- Sheppard, B. H., Hartwick, J. & Warshaw, P. R. (1988). The Theory of Reasoned Action: A Meta-Analysis of Past Research with Recommendations for Modifications and Future Research. *Journal of Consumer Research*, 15(3), 325-343.

- Sim, O. F. & Hamid, T. A. (2010). Social Protection in Malaysia Current State and Challenges Towards Practical and Sustainable Social Protection n East Asia: A Compassionate Community. In M. G. Asher, S. Oum and F. Parulian (Ed.), Social Protection in East Asia - Current State and Challenges (Vol. 2009-9, pp. 182-219). Jakarta: ERIA.
- Sivaramakrishnan, S., Srivastava, M. & Rastogi, A. (2017). Attitudinal factors, financial literacy, and stock market participation. *International Journal of Bank Marketing*, *35*(5), 818-841. doi: 10.1108/ijbm-01-2016-0012
- Smeeding, T. M. (2007). Retirement and Financial Security: An Economist Thinks Out Loud. In J. M. Wilmoth & K. F. Ferraro (Eds.), *Gerontology: Perspectives and Issues* (3rd ed., pp. 267-288). New York: Springer Publishing Company.
- Soley-Bori, M. (2013). *Dealing with missing data: Key assumptions and methods for applied analysis* (Technical Report No. 4) (pp. 1-19). Boston: Department of Health Policy & Management, Boston University.
- Sparks, P., Shepherd, R., Wieringa, N. & Zimmermans, N. (1995). Perceived behavioural control, unrealistic optimism and dietary change: An exploratory study. *Appetite*, 24, 243-255.
- Stawski, R. S., Hershey, D. A. & Jacobs-Lawson, J. M. (2007). Goal clarity and financial planning activities determinants of retirement savings contributions. *International Journal of Aging and Human Development*, 64, 13-32.
- Straub, D., Boudreau, M. C. & Gefen, D. (2004). Validation Guidelines for IS Positivist Research. Communications of the Associations for Information Systems, 13(24), 380-427.
- Street, D. & Desai, S. (2011). Planning for old age. In R. A. Settersten & J. L. Angel (Eds.), Handbook of Sociology of Aging (pp. 379–397). New York: Springer.
- StudyMalaysia.com. (2015). The Malaysian Higher Education System An Overview. Retrieved 12 June, 2016, from <u>https://www.studymalaysia.com/education/higher-education-in-malaysia/the-malaysian-higher-education-system-an-overview</u>
- Tabachnick, B. G. & Fidell, L. S. (2013). *Using Multivariate Statistics* (6th ed.). Boston: Pearson Education, Inc.
- Tan, H. B., Hoe, S. Y. & Hung, W. T. (2011). Financial Literacy and Personal Financial Planning in Klang Valley, Malaysia. *International Journal of Economics and Management*, 5(1), 149 – 168.
- Tan, H. K. & Folk, J. Y. (2011). Knowing when to Retire: The First Step Towards Financial Planning in Malaysia. *Educational Gerontology*, 37(10), 854-884.
- Tang, N., Baker, A. & Peter, P. C. (2015). Investigating the Disconnect between Financial Knowledge and Behavior: The Role of Parental Influence and Psychological Characteristics in Responsible Financial Behaviors among Young Adults. *Journal* of Consumer Affairs, 49(2), 376-406. doi: 10.1111/joca.12069

- Tang, T. L.-P. (1992). The meaning of money revisited. *Journal of Organizational Behavior*, 13(2), 197-202.
- Tashakkori, A. & Teddlie, C. (1998). *Mixed Methodology: Combining Qualitative and Quantitative Approaches*. Thousand Oaks, CA: Sage.
- Taylor-Carter, M. A., Cook, K. & Weinberg, C. (1997). Planning and expectations of the retirement experience. *Educational Gerontology*, 23(3), 273-288.
- Taylor, S. & Todd, P. A. (1995). Understanding Information Technology Usage: A test of competing models. *Information Systems Research*, 6(2), 144-176.
- Teh, K. L., Ng, S. T., Tey, N. P. & Siti-Norlasiah Ismail. (2013). Ethnicity and Support for Parents in Malaysia. *Asian Women, 29*(4), 51-72.
- Tengku Aizan, H. & Chai, S. T. (2013). Meeting the Needs of Older Malaysians: Expansion, Diversification and Multi-sector Collaboration. *Malaysian Journal of Economic Studies*, 50(2), 157-174.
- Teuscher, U. & Mitchell, S. H. (2011). Relation Between Time Perspective And Delay Discounting: A Literature Review. *Psychological Record*, *61*(4), 613-632.
- Thaler, Richard H. & Benartzi, S. (2004). Save More Tomorrow[™]: Using Behavioral Economics to Increase Employee Saving. *Journal of Political Economy, 112*(S1), S164-S187. doi: 10.1086/380085
- Thurstone, L. L. (1947). Multiple factor analysis. Chicago: University of Chicago Press.
- Topa, G., Lunceford, G. & Boyatzis, R. E. (2018). Financial Planning for Retirement: A Psychosocial Perspective. *Frontiers in psychology*, 8, 2338-2338. doi: 10.3389/fpsyg.2017.02338
- Townsend, C. & Liu, W. (2012). Is Planning Good for You? The Differential Impact of Planning on Self-Regulation. *Journal of Consumer Research*, 39(4), 688-703. doi: 10.1086/665053
- Trafimow, D. & Findlay, K. (1996). The Importance of Subjective Norms for a Minority of People: Between Subjects and Within Subjects Analyses. *Personality and Social Psychology Bulletin*, 22, 820-828.

Triandis, H. C. (1977). Interpersonal behavior. Monterey: Brooks/Cole.

- Trivedi, V. U., Shehata, M. & Mestelman, S. (2005). Attitudes, Incentives, and Tax Compliance. *Canadian Tax Journal*, 53(1), 29-61.
- Turner, J. C. (1991). Social Influences. Milton: Open University Press.
- United Nations. (2002). World Population Aging: 1950-2050. United Nations Publications. Retrieved 1 November, 2012, from http://www.un.org/esa/population/publications/worldageing19502050/

- Urbach, N. & Ahlemann, F. (2010). Structural equation modeling in information systems research using Partial Least Square. *Journal of Information Technology Theory and Application*, 11(2), 5-40.
- Vagias, W. M. (2006). Likert-type scale response anchors. Retrieved 3 April, 2019, from https://www.clemson.edu/cbshs/centers-institutes/institute-for-parks/research/
- Van Campenhout, G. (2015). Revaluing the Role of Parents as Financial Socialization Agents in Youth Financial Literacy Programs. *Journal of Consumer Affairs*, 49(1), 186-222. doi: 10.1111/joca.12064
- Van Dalen, H. P., Henkens, K. & Hershey, D. A. (2010). Perceptions and expectations of pension savings adequacy: a comparative study of Dutch and American workers. *Ageing & Society*, 30(5), 731-754. doi: 10.1017/S0144686X09990651
- Van Deventer, M., De Klerk, N. & Bevan-Dye, A. L. (2014). African Generation Y students' attitudes towards personal financial planning. *Mediterranean Journal of Social Sciences*, 5(21), 111-120. doi: 10.5901/mjss.2014.v5n21p111
- Van Rooij, M. & Teppa, F. (2014). Personal traits and individual choices: Taking action in economic and non-economic decisions. *Journal of Economic Behavior & Organization*, 100, 33-43. doi: 10.1016/j.jebo.2013.12.019
- Van Rooij, M. C. J., Lusardi, A. & Alessie, R. J. M. (2011). Financial literacy and retirement planning in the Netherlands. *Journal of Economic Psychology*, 32(4), 593-608. doi: 10.1016/j.joep.2011.02.004
- Van Rooij, M. C. J., Lusardi, A. & Alessie, R. J. M. (2012). Financial Literacy, Retirement Planning and Household Wealth. *Economic Journal*, 122(560), 449-478. doi: 10.1111/j.1468-0297.2012.02501.x
- Van Schie, R. J. G., Dellaert, B. G. C. & Donkers, B. (2015). Promoting later planned retirement: Construal level intervention impact reverses with age. *Journal of Economic Psychology*, 50, 124-131. doi: 10.1016/j.joep.2015.06.010
- Van Solinge, H. (2014). Who opts for self-employment after retirement? A longitudinal study in the Netherlands. *European Journal of Ageing*, 11(3), 261-272. doi: 10.1007/s10433-013-0303-7
- Velicer, W. F. & Jackson, D. N. (1990). Component analysis versus common factor analysis: Some issues in selecting an appropriate procedure. *Multivariate Behavioral Research*, 25, 1-28.
- Voorhees, C. M., Brady, M. K., Calantone, R. & Ramirez, E. (2016). Discriminant validity testing in marketing: An analysis, causes for concern, and proposed remedies. *Journal of the Academy of Marketing Science*, 44, 119-134.
- Vyvyan, V., Blue, L. & Brimble, M. (2014). Factors that Influence Financial Capability and Effectiveness: Exploring Financial Counsellors' Perspectives. *Australasian Accounting, Business and Finance Journal,* 8(4), 3-22.

- Walker, A. (2002). A strategy for active ageing. *International Social Security Review*, 55(1), 121-139.
- Waltz, C. F., Strickland, O. & Lenz, E. (1991). *Measurement in nursing research* (2nd ed.). Philadelphia: F.A. Davis.
- Wang, M. (2007). Profiling Retirees in the Retirement Transition and Adjustment Process: Examining the Longitudinal Change Patterns of Retirees' Psychological Well-Being. *Journal of Applied Psychology*, 92(2), 455-474.
- Wang, M., Henkens, K. & Van Solinge, H. (2011). Retirement Adjustment: A Review of Theoretical and Empirical Advancements. *American Psychologist*, 66(3), 204-213.
- Wang, M. & Shi, J. Q. (2014). Psychological Research on Retirement. Annual Review of Psychology, 65(1), 209-233. doi: 10.1146/annurev-psych-010213-115131
- Wang, M. & Shultz, K. S. (2010). Employee Retirement: A Review and Recommendations for Future Investigation. *Journal of Management*, 36(1), 172-206. doi: 10.1177/0149206309347957
- Wang, M. & Zhan, Y. (2012). The Critical Role Of Retirement Decision In Linking Pre-Retirement Attitudes To Post-Retirement Life Quality. *Gerontologist*, 52, 616-626.
- Wang, W. C., Worsley, A., Cunningham, E. & Hunter, W. (2014). The Heterogeneity of Middle-Age Australians' Retirement Plans. Social Work Research, 38(1), 36-46. doi: 10.1093/swr/svu005
- Webb, T. L. & Sheeran, P. (2006). Does changing behavioral intentions engender behavior change? A meta-analysis of the experimental evidence. *Psychological Bulletin*, 132(2), 249-268. doi: 10.1037/0033-2909.132.2.249
- Weinstein, N., Przybylski, A. K. & Ryan, R. M. (2012). The Index Of Autonomous Functioning: Development Of A Scale Of Human Autonomy. *Journal of Research* in Personality, 46 (4), 397-413.
- Wernimont, P. F. & Fitzpatrick, S. (1972). The meaning of money. *Journal of Applied Psychology*, *56*(3), 218-226.
- Werts, C. E., Linn, R. L. & Joreskog, K. G. (1974). Intraclass reliability estimates: Testing structural assumptions. *Educational and Psychological Measurement*, *34*, 25-33.
- Whetten, D. A. (1989). What Constitutes a Theoretical Contribution? Academy of Management Review, 14(4), 490-495.
- Wigfield, A. & Eccles, J. S. (2000). Expectancy–Value Theory of Achievement Motivation. *Contemporary Educational Psychology*, 25(1), 68-81. doi: 10.1006/ceps.1999.1015
- Willis, L. E. (2008). Against Consumer Financial Literacy Education. *Iowa Law Review*, 94(1), 197-285.

- Willis, L. E. (2011). The Financial Education Fallacy. *American Economic Review*, 101(3), 429-434. doi: 10.1257/aer.101.3.429
- Wilson, B. (2010). Using PLS to Investigate Interaction Effects Between Higher Order Branding Constructs. In V. E. Vinzi, W. W. Chin, J. Henseler & H. Wang (Eds.), *Handbook of Partial Least Squares: Concepts, Methods and Applications* (pp. 621-652). Heidelberg: Springer.
- Wold, H. (1985). Systems Analysis by Partial Least Squares. In P. Nijkamp, H. Leitner & W. Wrigley (Eds.), *Measuring the Immeasurable* (pp. 221-251). Dordrecht: Martinus Nijhoff.
- World Bank. (1994). Averting the old age crisis: policies to protect the old and promote growth. Washington, D.C.: World Bank.
- World Bank. (2008). *The World Bank Pension Conceptual Framework*. (World Bank Pension Reform Primer Series). Washington: World Bank.
- Xiao, J. J., Sorhaindo, B. & Garman, E. T. (2006). Financial behavior of consumers in credit counseling. *International Journal of Consumer Studies*, 30(2), 108-121.
- Xiao, J. J., Tang, C., Serido, J. & Shim, S. (2011). Antecedents and Consequences of Risky Credit Behavior Among College Students: Application and Extension of the Theory of Planned Behavior. *Journal of Public Policy & Marketing*, 30(2), 239-245. doi: 10.1509/jppm.30.2.239
- Xiao, J. J. & Wu, J. (2006). *Applying the Theory of Planned Behavior to Retain Credit Counseling Clients* (Working Paper, Take Charge America Institute for Consumer Financial Education and Research). Tucson: University of Arizona.
- Xiao, J. J. & Wu, J. (2008). Completing Debt Management Plans in Credit Counseling: An Application of the Theory of Planned Behavior. *Journal of Financial Counseling and Planning*, 19(2), 29-45.
- Yamauchi, K. T. & Templer, D. I. (1982). The Development of a Money Attitude Scale. Journal of Personality Assessment, 46(5), 522.
- Yang, T. Y. & Devaney, S. A. (2012). Determinants of retirement assets and the amount in stock in retirement assets. *Family and Consumer Sciences Research Journal*, 41(1), 36-55. doi: 10.1111/j.1552-3934.2012.02127.x
- Yeske, D. (2010). Finding the planning in financial planning. *Journal of Financial Planning*, 23(9), 40-51.
- Yilmazer, T. & Lich, S. (2015). Portfolio choice and risk attitudes: a household bargaining approach. *Review of Economics of the Household, 13*(2), 219-241. doi: 10.1007/s11150-013-9207-8
- Yin, R. K. (1994). *Case study research: Design and methods* (2nd ed.). Beverly Hills: Sage Publications.

- Yoo, G. & Kim, B. (2010). Remembering Sacrifices: Attitude and Beliefs Among Secondgeneration Korean Americans Regarding Family Support. *Journal of Cross-Cultural Gerontology*, 25(2), 165-181. doi: 10.1007/s10823-010-9116-8
- Zhan, Y., Wang, M. & Yao, X. (2013). Domain specific effects of commitment on bridge employment decisions: The moderating role of economic stress. *European Journal* of Work and Organizational Psychology, 22(3), 362-375. doi: 10.1080/1359432X.2012.762763
- Zheng, J., Wang, H. & Gao, L. Y. (2012). Pursuit of Meticulous Management to Construct University Financial Information System. Paper presented at the International Conference on Management Innovation and Public Policy (ICMIPP), Chongqing, China.
- Zikmund, W. G. (2003). Business research methods (7th ed.). Ohio: South-Western.
- Zimmerman, L., Canale, A., Britt, S. L. & Seay, M. (2015). The Theory of Planned Behavior and the Earned Income Tax Credit. *Journal of Financial Therapy*, 6 (1). doi: 10.4148/1944-9771.1066

LIST OF PUBLICATIONS AND PAPERS PRESENTED

Articles under review

Choo, S. M., Mansor, N. & Baranovich, D. (2018). The psychology of planning: Financial planning for retirement. *Educational Gerontology*.

Educational Gerontology is a journal indexed by ISI Web of Knowledge.

Papers presented

Choo, S. M., Mansor, N. & Baranovich, D. (2017). *An Explorative Study of Planning for Retirement Activities.* Paper presented at the Third International Conference on Social Protection, University of Malaya, Kuala Lumpur.